# **COLLECTIVE AGREEMENT**

between

# WESTSHORE TERMINALS LIMITED PARTNERSHIP

and

# INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Effective

February 1, 2020 to January 31, 2022

03253 (13)

| Article |                                       | Page # |
|---------|---------------------------------------|--------|
| ARTICL  | E 1 PURPOSE OF AGREEMENT              | 8      |
| 1.01    | Intent and Purpose                    | 8      |
| 1.02    | Notice                                | 8      |
| ARTICLE | 2 SCOPE OF AGREEMENT                  | 9      |
| 2.01    | Scope                                 | 9      |
| 2.02    | Introduction of Technology            | 9      |
| 2.03    | Performance of Duties                 |        |
| 2.04    | Contracting Out                       | 10     |
| ARTICL  | E 3 MANAGEMENT                        | 14     |
| 3.01    | Management Functions                  | 14     |
| 3.02    | Rules and Regulations                 |        |
| 3.03    | Employment of Personnel               | 15     |
| ARTICL  | E 4 ADJUSTMENTS OF GRIEVANCES         | 16     |
| 4.01    | Purpose                               |        |
| 4.02    | Procedure                             |        |
| 4.03    | Arbitrator's Remuneration             |        |
| 4.04    | Employee Records - Limitation         |        |
| ARTICL  | E 5 CESSATION OF WORK                 | 19     |
| 5.01    | Strikes and Lockouts                  | 19     |
| 5.02    | Continuation of Work                  |        |
| 5.03    | Pollution Control                     | 19     |
| ARTICL  | E 6 HOURS OF WORK AND SALARY SCHEDULE | 20     |
| 6.01    | Definitions                           |        |
| 6.02    | Rest and Lunch Breaks                 |        |
| 6.03    | Normal Work Pattern                   |        |
| 6.04    | Starting Times                        |        |
| 6.05    | Hours of Work                         |        |
| 6.06    | Industrial Accidents                  |        |

Article

| Page | # |
|------|---|
|------|---|

| ARTICLE 7 OVERTIME |   |    |
|--------------------|---|----|
| 7.01               | Purpose   | 25 |
| 7.02               | Definition of Terms                                 | 25 |
| 7.03               | Conditions Under Which Overtime Rates Shall Be Paid |    |
| 7.04               | Extended Shifts                                     | 27 |
| 7.05               | Call-outs   |    |
| 7.06               | Banked/Deferred Time Off                            |    |
| 7.07               | Book-offs   |    |
| 7.08               | Overtime Callouts                                   |    |
| 7.09               | Personal Days                                       |    |
| ARTICLE            | 8 HOLIDAYS  | 35 |
| 8.01               | Days  |    |
| 8.02               | Eligibility   |    |
| ARTICLE            | 9 VACATIONS   |    |
| 9.01               | Vacation Entitlement                                |    |
| 9.02               | Vacation Pay Rates                                  | 41 |
| 9.03               | Conditions  | 41 |
| 9.04               | Vacation Pay  | 42 |
| 9.05               | Holidays During Vacation Period                     | 44 |
| 9.06               | Time of Vacation                                    | 44 |
| 9.07               | Vacation Lists/Schedules                            |    |
| 9.08               | Vacation Scheduling & Disability                    |    |
| 9.09               | Vacation Payout on Retirement                       | 47 |
| ARTICLE            | 10 INSURANCE BENEFITS                               | 49 |
| 10.01              | Benefit Plans                                       | 49 |
| 10.02              | Widow's Benefits                                    | 51 |
| 10.03              | Explanatory Brochures                               |    |
| 10.04              | Insurance Plan Policies                             |    |
| 10.05              | Eligibility   |    |
| 10.06              | Duplication of Coverage                             | 53 |

Article

| Page | # |
|------|---|
|------|---|

| ARTICLE | 11 SALARY CONTINUANCE                                   | 54 |
|---------|---|----|
| 11.01   | Scope   | 54 |
| 11.02   | Sick Leave Allowance                                    | 54 |
| 11.03   | Eligibility   |    |
| 11.04   | Amounts Payable   |    |
| 11.05   | Long Term Disability                                    | 57 |
| 11.06   | Sequence  |    |
| 11.07   | Claims Procedure  |    |
| 11.08   | Medical Certificates                                    |    |
| 11.09   | Health Committee  | 61 |
| 11.10   | False Claims  | 63 |
| 11.11   | Modified Work   |    |
| 11.11   | Labour Relations Health and Attendance Review Committee |    |
| ARTICLE | 12 ABSENCES   | 66 |
| 12.01   | Salary Payment and Notification                         |    |
| 12.02   | Union Business  |    |
| 12.03   | Bereavement Leave                                       |    |
| 12.04   | Allowance for Jury Duty                                 | 70 |
| 12.05   | Leaves of Absence                                       | 70 |
| ARTICLE | 13 PHYSICAL EXAMINATIONS                                | 72 |
| 13.01   | Medical and Physical Examinations                       | 72 |
| 13.02   | Designated Medical Clinic                               | 72 |
| 13.03   | Salary Continuance Provisions                           | 72 |
| ARTICLE | 14 SAVING CLAUSE  | 73 |
| 14.01   | Invalid Provisions                                      | 73 |
| ARTICLE | 15 SENIORITY  | 74 |
| 15.01   | Determination of Seniority                              | 74 |
| 15.02   | Loss of Seniority                                       | 74 |
| 15.03   | Reductions in Workforce                                 | 74 |

19.08

19.09

| Article |  | Page # |
|---------|--|--------|
| 15.04   | Temporary Employees - Seniority              |        |
| 15.05   | Temporary Layoffs                            | 77     |
| 15.06   | Permanent Layoffs - Early Retirement Program |        |
| 15.07   | Permanent Layoffs - Severance Allowance      |        |
| 15.08   | Continuation of Benefits - Regular Employees |        |
| ARTICLE | 16 SAFETY                                    | 80     |
| 16.01   | Safety Equipment, Appliances and Conditions  |        |
| 16.02   | Safety Committee                             |        |
| 16.03   | Review of Safety Disputes                    |        |
| ARTICLE | 17 TRUST FUND                                | 82     |
| 17.01   | Trust Fund                                   |        |
| 17.02   | Eyeglass Benefits                            |        |
| ARTICLE | 18 TRAINING                                  | 84     |
| 18.01   | Training Programs                            |        |
| 18.02   | Wages, Registrations Fees and Expenses       |        |
| 18.03   | New or Changed Equipment                     |        |
| ARTICLE | 19 PENSIONS                                  | 85     |
| 19.01   | Pension Plan                                 |        |
| (a)     | Amount of Pension                            |        |
| (b)     | Bridging Supplement                          |        |
| (c) Cr  | edited Service                               |        |
| 19.02   | Severance Benefit                            |        |
| 19.03   | Pensioners Medical Benefits                  |        |
| 19.04   | Funeral Allowance                            |        |
| 19.05   | Credited Pension Service                     |        |
| 19.06   | Vesting Provision                            |        |
| 19.07   | Pension Plan Conditions                      |        |

| Article |                                    | Page # |
|---------|------------------------------------|--------|
| 19.10   | Defined Contribution Plan          |        |
| 19.11   | Pension Service Lists              |        |
| ARTICLE | 20 VACANCIES                       | 94     |
| 20.01   | Temporary Vacancies                |        |
| 20.02   | Shift Replacement - Short Shifting |        |
| 20.03   | Replacements - Spareboard List     |        |
| 20.04   | Permanent Vacancies                | 97     |
| ARTICLE | 21 FIRST AID                       | 98     |
| 21.01   | First Aid Man                      |        |
| 21.02   | First Aid Courses and Certificates |        |
| ARTICLE | 22 EMPLOYEE DEFINITION             | 99     |
| 22.01   | Employee Categories                |        |
| ARTICLE | 23 PROTECTIVE CLOTHING             | 100    |
| 23.01   | Clothing and Equipment             |        |
| 23.02   | Safety Work Boots                  |        |
| 23.03   | Raingear                           |        |
| ARTICLE | 24 MANNING                         |        |
| 24.01   | (a) Operations                     |        |
| 24.01   | (b) Maintenance                    |        |
| 24.01   | (c) Day Shift Operations           |        |
| 24.02   | Replacements                       |        |
| 24.03   | Vacation Relief Dispatch           |        |
| 24.04   | General                            |        |
| 24.05   | Absence Replacements               |        |
| 24.06   | Manpower Shortages                 |        |
| ARTICLE | 25 SALARY                          | 110    |
| 25.01   | Rates of Pay                       |        |
| 25.02   | (a) Temporary Employees            |        |

| Article  | Page # |
|--|--------|
| 25.02 (b) New Hire Pensions                                |        |
| ARTICLE 26 MISCELLANEOUS                                   | 112    |
| 26.01 Elevations   |        |
| 26.02 Discipline   |        |
| ARTICLE 27 TERM  | 113    |
| 27.01 Term of Agreement                                    |        |
| APPENDIX "A" ASSIGNED WORK SCHEDULE                        | 115    |
| Assigned Work Schedules                                    | 115    |
| Schedule I – Eight (8) Hour and Thirty-Five Minute Shift   |        |
| Schedule II – Eight (8) Hour and Thirty-Five Minute Shift  |        |
| Schedule III – Eight (8) Hour and Thirty-Five Minute Shift |        |
| Schedule IV – Eight (8) Hour and Thirty-Five Minute Shift  |        |
| Schedule V – Eight (8) Hour and Thirty-Five Minute Shift   |        |
| APPENDIX "B" REPLACEMENTS                                  | 121    |
| APPENDIX "C" TRANSFER SEQUENCE                             |        |
| CALENDARS 2021 TO 2025                                     |        |

# **ARTICLE 1 PURPOSE OF AGREEMENT**

# 1.01 Intent and Purpose

It is the intent and purpose of this Agreement to set forth the rates of pay, hours of work, and other conditions of employment in order that orderly, harmonious and peaceful relations are achieved between the Parties.

# 1.02 Notice

The representatives of the Company and the Union shall provide each other with advance notice, in writing, on all matters affecting the administration of the terms of this Agreement.

# **ARTICLE 2 SCOPE OF AGREEMENT**

# 2.01 Scope

- (a) This Agreement shall cover all laborers, production and maintenance Employees engaged in handling bulk commodities or other cargoes for the Company at Westshore Terminals Limited Partnership ("Westshore"). This Agreement excludes all Employees who are members of Locals 514 and 517, professional Employees, Security personnel and Management personnel.
- (b) The Company recognizes the Union as the exclusive representative of all Employees covered by this Agreement for the purpose of collective bargaining with respect to wages, hours of work and other working conditions.
- All regular production, maintenance, and pollution control work will be performed by Employees covered by this Agreement.

### 2.02 Introduction of Technology

- (a) The Company agrees that the Union will be advised in writing a minimum of six (6) months in advance where the introduction of new technology will affect the work normally performed by such Employees. Such change shall not be implemented without prior discussion with the Union. The discussion shall be meaningful and will include a description of the new technology, how it will affect the performance of the work, the employee(s), and whether training or retraining is required with respect to such technology. If training as referenced is required, it will be provided to the affected employees.
- (b) There shall be no reduction in the workforce through layoff as a direct result of the introduction of technological change. Where a reduction in the workforce is necessary, as a result of the introduction of technological change, such reduction shall be achieved through attrition.

# 2.03 Performance of Duties

- (a) Nothing in this Article shall interfere with the normal performance of duties such as the carrying out of research programs, the necessary instruction of Employees, the inspection of equipment, product or process by persons not covered by this Agreement.
- (b) Employees will not be assigned to work beyond their personal skills unless they are being directed by an Employee having the technical skills for the work assigned.
- (c) It is understood that the loading of ships to the point where the hatch covers may be safely closed; including the handling of spillage, and overloads is part of the Company's regular production work. If a spill or overload occurs, which cannot be cleaned up by the designated men from the crew in two (2) hours of shoveling, extra help will be hired from the Union Hall, (consisting of a four (4) man crew).
- (d) The Company shall have qualified regular Electricians who are members of Local502 update and maintain electrical blueprints.
- (e) It is understood and agreed that the switching, re-coupling of trains and setting of hand brakes, as required, shall be done by Employees covered by this Agreement. This provision shall not apply to work presently performed on site by railway personnel.

# 2.04 Contracting Out

### (a) **General**

The Company may contract out work involving new construction, new facilities (new or replacement buildings), new installations (new or replacement major equipment) or warranty work, including removal and reinstallation, as required to validate the warranty. Warranty work is defined as the original warranty as specified at the date of purchase or installation.

### (b) **Definition of Regular Maintenance**

"Regular Maintenance" for the purposes of this Article shall be defined as that work normally performed to maintain the facilities and equipment on an ongoing basis. The normal course of such work, subject to subsection 2.04 (a) above, includes the removal, repair, replacement, and reinstallation of all parts, motors, electrical systems and fabrication of replacement components.

#### (c) **Contracting Out Committee**

#### (i) Structure

The Company and the Union shall establish a Contracting Out Committee composed of two (2) representatives of the Company and two (2) representatives of the Union. The Company and the Union representatives may each have an adviser in attendance at Committee meetings.

#### (ii) *Meetings*

The Contracting Out Committee shall meet on a regular monthly basis, and/or as needed, to deal with matters related to planned shutdowns or other contracting out issues referred to in this Article, including warranty work and service agreements. <u>In addition, on an</u> <u>annual basis, the Company will provide an overview of potential</u> <u>major projects planned for the year. This is provided as a forecast</u> <u>only and is subject to additions or deletions.</u>

#### (iii) **Responsibility**

It shall be the responsibility of the Contracting Out Committee to review the requirements of the Company related to planned shutdowns of equipment or facilities and to determine the following.

- requirement to maintain ongoing regular maintenance of the facility, and
- the equipment and/or facilities necessary to perform the work required, and
- the skills required to perform the work required, and
- work that would normally be done in-house and that work which would be done by outside personnel, and
- the availability of the required tradesmen, and
- to ensure coordination of all work to be performed by Employees and outside personnel, and
- requirement to complete work within reasonably established time constraints.

### (iv) **Agenda/Minutes**

The Company shall provide a written agenda and minutes pertaining to meetings of the Committee. In addition, the Company shall provide the Union with a work program related to planned shutdowns prior to the commencement of such work.

### (d) **Notices**

The Maintenance Planning Department shall be responsible for all matters to be contracted out and shall advise the Union in writing, and post notices to all maintenance and production bulletin boards, as follows:

 (i) within a minimum of seventy-two (72) hours when the Company intends to have equipment and/or parts not normally sent off site to be repaired, and  (ii) within a minimum of forty-eight (48) hours when contractors are to be brought on site to perform work, excluding those covered by ongoing master agreements.

It is understood and agreed that the above notice provisions shall not apply to contracting out issues related to emergency breakdowns. However, the Company will advise the Union prior to the commencement of any work to be performed by contractors.

# **ARTICLE 3 MANAGEMENT**

# 3.01 Management Functions

The Management of the Company, and the direction of the work force, including the right to hire, suspend or discharge for proper cause, or transfer, the right to relieve Employees from duty, and the right to make and alter "written and posted" rules and regulations to be observed by the Employees, is vested exclusively in the Company, provided that this will not be used for purposes of discrimination against any Employee; provided, further, that the provisions hereof are subject to the other provisions of this Agreement.

# 3.02 Rules and Regulations

Where the Company establishes rules and regulations, as referred to in section 3.01 of this Article, such rules and regulations shall be in accordance with the following:

- (a) Consistent with the terms and conditions of this Agreement;
- (b) Clear and reasonable;
- (c) Applied consistently taking into account individual circumstances; and
- (d) Employees shall be advised in writing prior to implementation.
- (e) Prior to implementing or changing a rule or regulation, the Company will provide reasonable advance notice and a copy of the rule or regulation to the Union, and meet with representatives of the Union to review the circumstances giving rise to the rule or regulation and engage in meaningful discussions about its application and effect. In circumstances where the delay in implementing the rule or regulation would create demonstrable prejudice, the Company will provide the Union with a copy of the rule or regulation and implement it prior to the aforementioned meeting, which will be held within seven days unless mutually agreed otherwise.

# 3.03 Employment of Personnel

The number of persons to be employed shall be at the discretion of the Employer with the following rules applying:

- (a) Maintenance of safety;
- (b) Avoidance of undue individual work burden:
- (c) Prohibition of individual speed-up:
- (d) In accordance with the provisions of Article 24; and
- (e) In accordance with past practice it is understood that the Company will discuss transfers from crew to crew with the Union.

# **ARTICLE 4 ADJUSTMENTS OF GRIEVANCES**

# 4.01 Purpose

It is the intent and purpose of this Article, which shall be available to both the Union and the Employer, to provide for the presentation and equitable adjustment of grievances. Should any differences arise between the Employer and the Union as to the meaning or application of the provisions of this Agreement or as to any question relating to the wages, hours of work or other conditions of employment of any Employee, the same shall be disposed of in accordance with the provisions of this Article.

Pending the investigation and settlement of any grievance, work shall continue to be performed as provided in Article 5, Cessation of Work.

# 4.02 Procedure

Each grievance shall be presented to the appropriate Party hereinafter indicated within seven (7) calendar days after the occurrence of the event causing the grievance.

#### **STEP 1**:

Any Employee who believes that he has a justifiable request or complaint may discuss the matter with his Supervisor, with or without his Shop Steward (or his replacement), Plant Chairman, Plant Secretary or Business Agent, being present to settle same. Where the request or complaint has been submitted to the Supervisor, he shall give the aggrieved Employee an answer as quickly as possible, but in no event later than three (3) days (exclusive of days off or recognized holidays), the matter may be referred in writing within seven (7) days to Step No. 2 if the answer given is not acceptable.

#### STEP 2:

Grievances which are appealed to Step 2 in a timely fashion shall be dated and signed by the aggrieved and his Business Agent, and shall be discussed by the Department Superintendent, or his designate, and the Business Agent at a mutually convenient time. Either Party may call

witnesses who are Employees of the Company and attendance shall be limited to time required for their testimony. At this time the issues and facts will be thoroughly reviewed for the purpose of securing a prompt resolution. The Company representative will give a written disposition signed and dated, of the decision reached within seven (7) calendar days from the date of presentation in this Step 2.

### STEP 3:

In the event a grievance is not settled in Step 2 hereof, written notice of appeal to Step 3 shall be served by either party prior to the expiration of ten (10) calendar days following disposition of the grievance in Step 2 hereof. Such notice shall state:

- (a) The subject matter of the grievance; and,
- (b) The objections of the appealing party of the previous disposition.

The Step 3 discussions shall be between two (2) persons, certified in writing, by the Union as the representatives for this purpose, and not more than two (2) representatives of the Employer certified to the Union in writing as the representative of the Employer for the purpose of attempting to settle the grievance.

Either party may request a further statement of facts to be made available not later than three (3) calendar days preceding the date set for the Step 3 meeting. Either party may produce at the Step 3 meeting any persons familiar with the facts involved to aid in a solution to the problem.

The number of meetings required in Step 3 to reach a satisfactory conclusion shall be determined jointly between the representative(s) of the Employer and the representative(s) of the Union.

Typewritten minutes shall be kept of Step 3 meetings and shall conform essentially to the following outline:

- (a) Date and place of meeting.
- (b) Names and positions of those present.
- (c) Identifying number and description of the grievance.
- (d) Brief statement of Union position.

- (e) Brief statement of Employer's position.
- (f) Abstract of important aspects of the discussion.
- (g) Decision reached.

These minutes are to be presented within ten (10) calendar days following the Final Step 3 meeting.

#### STEP 4:

In the event a grievance is not settled in Step 3, written notice of appeal to arbitration shall be served by either party within the expiration of 30 calendar days following receipt of the minutes of the final Step 3 meeting. The arbitrator shall be mutually selected from a panel of three (3) names submitted by the Company and three (3) names submitted by the Union. The selected arbitrator shall be appointed, in writing, and shall have the powers, duties and jurisdiction to resolve the grievance within the confines of this Agreement.

The agreed upon arbitrator shall not be authorized to make any decision inconsistent with the provisions of this Agreement, nor, to alter, modify or amend any part of this Agreement. The decision of the Arbitrator shall be final and binding on the parties hereto.

# 4.03 Arbitrator's Remuneration

The Arbitrator shall devote such time as is necessary to discharge his duties and responsibilities and he shall be paid at a rate and upon a basis to be agreed upon between the Arbitrator and the parties. His remuneration shall be paid one-half by the Union and one-half by the Company.

# 4.04 Employee Records - Limitation

For the purpose of this Article the Company cannot consider the grieving Employee's past record for more than two (2) years prior to the date of the grievance in question.

# **ARTICLE 5 CESSATION OF WORK**

### 5.01 Strikes and Lockouts

- (a) The Union agrees that during the term of this Agreement there will be no slowdown nor strike, stoppage of work, interruption or cessation of work, or refusal to work or to continue to work.
- (b) The Company agrees that during the term of this Agreement there will be no lockout.

# 5.02 Continuation of Work

The Union agrees that in the event of strikes or walkouts, the Union will not take similar action on the ground of sympathy, but will continue to work.

# 5.03 Pollution Control

- (a) If the Picket Line is manned by members of the Union, such measures as are necessary to fulfill obligations regarding pollution control will be performed with supervision from Local 514, I.L.W.U. by the Union.
- (b) In the event picket lines are established by parties other than members of the Union, the Union will make every effort to obtain picket passes for Local 502 members to perform pollution control functions.
- (c) Except for the above, the Company does not expect members of the Union to cross a picket line.

# **ARTICLE 6 HOURS OF WORK AND SALARY SCHEDULE**

### 6.01 Definitions

- Normal Work Week means that period of time from midnight on Saturday night to midnight on the succeeding Saturday night.
- (b) Normal Work Day shall be hours of work set out in the Employee's assigned work schedules, which form part of this Agreement. The hours of work in a normal work day shall be consecutive.

#### 6.02 Rest and Lunch Breaks

- (a) When production and/or maintenance is on a continuous operation, the shifts shall be those set out in the assigned work schedules with the Employee allowed onehalf (1/2) hour to eat on the Company's time. No Employee will be required to work in excess of five (5) consecutive hours without a meal break. In any case, an Employee's meal break will be observed between the third and fifth hours of any shift worked.
- (b) All Employees shall receive two (2) fifteen (15) minute rest breaks per shift. The Company agrees that this provision shall be administered on a reasonable basis given individual circumstances and working conditions.
- (c) It is also understood and agreed that the duration of lunch and rest breaks shall be time spent in the Dry facilities.

### 6.03 Normal Work Pattern

- (a) This Agreement provides for the operation of Westshore, on continuous operations seven (7) days per week in accordance with the shift schedules attached to and forming part of this Agreement.
- (b) No hours worked in accordance with any schedule contained in this Agreement shall be construed as overtime.

- (c) Persons employed to fill a position temporarily will be paid the equivalent hourly rate of the position being filled on the schedule for which the position is designated.
- (d) It is agreed the Company retains the option to maintain a steady day shift crew, if required.
- (e) It is further understood and agreed that the Company may institute a maintenance night shift (MA-ME - 12 midnight to 08:00) at such time as deemed necessary by the Company. The Company shall give a minimum of sixty (60) calendar days notice prior to instituting such night shift.
- (f) Any changes to existing schedules or implementation of a new schedule shall be subject to mutual agreement of the Company and the Union.

# 6.04 Starting Times

(a) The starting times for shifts will under normal conditions be:

Continuous Operation

| Day Shift       | 8:00 a.m.      |
|-----------------|----------------|
| Afternoon Shift | 4:00 p.m.      |
| Night Shift     | 12:00 Midnight |

(b) The above starting times, as established by the Company, may be changed by mutual agreement of the Company and the Union to meet the requirements and level of operations.

# 6.05 Hours of Work

- (a) **Operations** 
  - (i) Day Shift (M1 to M5 schedules)
  - 8:00 a.m. to 4:00 p.m.

The above hours of work include a twenty (20) minute washup period at the end of the shift which commences at the time the Employee leaves the work area.

- (ii) Continuous (A to E schedules) Hot Change
- <u>8:00 a.m. to 4:35 p.m.</u>
- <u>4:00 p.m. to 12:35 a.m.</u>
- <u>12 midnight to 8:35 a.m.</u>
   <u>The dispatch of Employees working a Hot Change schedule shall be as per past practice and an Employee who is relieved late (after fifteen (15) minutes past the hour) will be paid one (1) hour at overtime rates.</u>
- (iii) Continuous (A to E schedules) No Hot Change
- 8:00 a.m. to 4:20 p.m.
- <u>4:00 p.m. to 12:20 a.m.</u>
- <u>12 midnight to 8:20 a.m.</u>

The above hours of work include a fifteen (15) minute period at the end of the shift to cleanup tools and equipment and a twenty (20) minute period for washup. The cleanup and washup time commences at the time the Employee leaves the work area.

#### (b) Maintenance

- (i) Day Shift (M1 to M5 schedules)
- 8:00 a.m. to 4:00 p.m.
   The above hours of work include a twenty (20) minute washup period at the end of the shift which commences at the time the Employee leaves the work area.
- (ii) Continuous Schedule

- 8:00 a.m. to 4:20 p.m.
- <u>4:00 p.m. to 12:20 a.m.</u>
- <u>12 midnight to 8:20 a.m. (if a Night Shift is instituted)</u>

The above hours of work include a fifteen (15) minute period at the end of the shift to exchange information and a twenty (20) minute period for washup. The information transfer and washup time commences at the time the Employee leaves the work area.

### (c) Interpretation

- The "work area" for the purpose of subsections (a) (i) and (b) (i) shall be the specific area of the site where the employee is working immediately prior to the end of the shift.
- 2. <u>The time for leaving the "work area" and proceeding to the Dry facility, as</u> <u>it applies to subsection (a) (i) and (b) (i), shall be as follows:</u>
  - <u>an employee will leave the work area at the end of the shift to provide</u> <u>twenty (20) minutes for washup in the dry facility.</u>
- 3. <u>The time for leaving the "work area" at the end of the shift, as it applies to</u> <u>subsection (b) (ii). shall be as follows:</u>
  - an employee will leave the work area at the end of the shift at fifteen

     (15) minutes to the top of the hour. (All work related to cleanup of the
     work area, loading the vehicle with tools, etc., and removing personal
     locks from equipment, shall be completed prior to the commencement
     of the fifteen (15) minute period).

The foregoing hours of work. and related conditions, shall remain in effect unless changed by mutual agreement of the Parties.

# 6.06 Industrial Accidents

An Employee involved in an industrial accident while employed by the Company shall be paid for the entire shift for the day of the accident. This provision shall only apply where the Employee requires the attention and services of a medical practitioner or other registered medical professional.

# **ARTICLE 7 OVERTIME**

# 7.01 Purpose

- (a) This Article contains the provisions concerning overtime and specifies the conditions and rates applicable.
- (b) The intent of Article 7 is not to be used to circumvent the addition or reduction of a second or third shift.

### 7.02 Definition of Terms

- (a) The payroll week shall consist of seven (7) consecutive days beginning at 00:01 hours Sunday, or at the turn changing hour nearest that time.
- (b) The workday for purposes of this Article is the twenty-four (24) hour period beginning with the time the Employee begins work, except that a tardy Employee's workday shall begin at the time it would have begun had he not been tardy.
- (c) The regular rate of pay, as the term is used in Section 7.03 of this Article, shall mean the hourly equivalent salary rate which the Employee would have received for his scheduled work had it been performed during non-overtime hours.

### 7.03 Conditions Under Which Overtime Rates Shall Be Paid

- (a) Double time the regular rate of pay shall be paid for hours worked in excess of the Employee's normal work week, or work day, as set out in the shift schedules which are appended hereto and form part of this Agreement.
- (b) Three (3) times the regular rate of pay shall be paid for consecutive hours worked in excess of the normal work day where he works at double time for a normal work day.
- (c) Employees who perform work on a Statutory Holiday, as defined in Article 8, shall be entitled to either:

- Double the Employee's equivalent hourly rate for hours worked on the Statutory Holiday, (except that if New Year's Day is worked the rate shall be two and one-half (2.5) times the equivalent hourly rate), or
- (ii) Two (2) days off in lieu, provided he qualifies for pay for the full shift, or
- (iii) Elect to defer such days in accordance with the provisions of Article 7, section 7.07 of this Agreement.

Employees who are eligible under Article 8 shall receive either (i) or (ii) above in addition to their regular salary, should they perform work on a holiday.

Employees shall elect in writing either (i) or (ii) above at the time they report for their shift on the Statutory Holiday.

Employees who have elected (ii) above shall notify the Company in writing when they wish to take their days off. Such notice shall be at least seven (7) days prior to the requested days off. The days requested must be mutually agreeable to both the Company and the Employee before the Employee shall be granted those days off.

- (d) If the Employee's normal work day falls on a Statutory Holiday and the Employee is paid double time, in accordance with Article 7, subsection 7.03 (c), the Employee shall receive triple time for consecutive hours worked in excess of the hours in the Employee's normal work day.
- (e) Payment of overtime rates shall not be duplicated for the same hours worked. To the extent that hours are compensated for at overtime rates under one provision, they shall not be counted as hours worked in determining overtime under the same or any other provision.
- (f) An Employee will not be required to work in excess of eight (8) hours overtime within his normal work day.

- (g) If an Employee works in excess of eight (8) hours (or one shift) that Employee shall have ten (10) hours off with no loss of pay from his next regular scheduled shift.
- (h) An Employee working a full shift of O/T (8.58 hours) has the option of banking, as per limitations of the agreement, two (2) shifts of nine (9.0) hours or being paid nine (9.0) hours at the double time rate of pay. The third option would be banking one day and being paid for one day (9.0 hours and banking 9.0 hours). Subsequent days which are taken off will be taken on either 8.0 hour or 8.58 hour schedule and the appropriate day charged from the banked time. O/T worked on a day schedule 8.0 hours can also be taken off as noted above, either on 8.58 hour or 8.0 hour shift, and the deduction made from the banked time.

### 7.04 Extended Shifts

- (a) Extended shifts is defined as the consecutive hours worked in excess of the Employee's normal work day where prior to the end of the shift the Employee is notified accordingly. An extended shift is not a call-out.
- (b) When an extended shift is needed for an individual or crew, the Company will be obligated to furnish a hot meal after the first hour of the extended period. Exceptions to this may be agreed to on a local basis, but in no case would the cost exceed the above stipulated benefit.
- (c) The parties to this Agreement recognize that from time to time extension of shifts will be required because of the nature of the work involved. However, the Company pledges that there will not be a flagrant abuse of shift extensions and further, when they are required, such extensions should be divided equitably among the Employees if possible. The work schedules lend themselves to this application.
- (d) Shift extensions as they apply to such maintenance Employees and electricians working the day shift shall not exceed one (1) hour in accordance with present practice. It is understood that overtime work required in excess of the one (1)

hour shall be assigned in accordance with the provisions of Article 20, section 20.01, subsections (c) and (d).

#### 7.05 Call-outs

- (a) Employees may be required by the Company to work non-scheduled hours. An Employee called back to work, or otherwise called to work off schedule, will receive a minimum of eight (8) hours at double time for non-scheduled work. An Employee called out to work overtime during any shift, but prior to the midpoint of the shift, shall only be required to work to the end of that shift to perform the work for which he was called out. Where an Employee is called out after the midpoint of any shift, such Employee will only be required to work to the midpoint of the following shift.
- (b) In the event that the call-out overtime work continues into the Employee's regular scheduled work shift, and:
  - (i) The Employee worked on call-out in excess of four (4) hours immediately prior to the commencement of his scheduled shift, that Employee will be provided with the option to continue to work his scheduled shift or book off work during his scheduled shift; however, if the Employee books off on his scheduled shift, that Employee will have deducted from his regular pay an amount equal to his salary equivalent for the hours he was absent from his scheduled shift, except 12:00 a.m. to 8:00 a.m. shift, or
  - (ii) The Employee worked on call-out less than four (4) hours immediately prior to his scheduled shift, such Employee will be required to work his scheduled shift.
  - (iii) An Employee called out to work four (4) consecutive hours early beginning at 4:00 a.m. on the 12 to 8 a.m. shift, in accordance with Article 20, section 20.01 subsections (a) (iii),

or (b) (iii), shall be paid (4) hours at double time plus one (1) hour at straight time in lieu of a meal break. The Employee shall not be required to work past 8:00 a.m., and this shall not result in any reduction of regular salary.

(c) When the Company is scheduling planned work on a graveyard shift and requires an additional Foreman to its normal complement, and has received confirmation of an available Foreman, they will order two Employees to work under the direction of the Foreman once they are reasonably certain the planned work is required to be performed. Substitutions for classifications ordered will be at the discretion of the Company, and should the maintenance priority change for the shift in question due to an emergency breakdown that the Company has determined requires immediate attention, the Company will exhaust all avenues as per past practice for procuring manning prior to these Employees being assigned to the new tasks.

### 7.06 Banked/Deferred Time Off

An Employee shall be allowed to bank all overtime worked in a calendar year.

- (a) An Employee may elect to withdraw or defer overtime banked in a calendar year prior to March 31st of the year subsequent to the year in which the banked overtime was accumulated in accordance with the following:
  - (i) the Employee shall notify the Company, in writing, a minimum of seven (7) calendar days in advance stating the requested day or day days to be taken off. The days requested must be mutually agreed to by the Company and the Employee before the days are taken off, or
  - (ii) the Employee shall be paid out unused bank days by April15th of the calendar year following the year in which the

banked overtime was accumulated at the rate at which such days were earned.

- (b) It is understood and agreed that where Employees elect to take banked overtime off, the Company may where required, replace such Employees from the Union Hall without regard to the provisions of Article 20, section 20.01.
- (c) An Employee may also elect by March 31st to defer banked time to be used as Early Retirement Credits, Paid Extended Absences or paid in cash in accordance with the provisions of this section.
- (d) An Employee who has elected to defer banked overtime, days in lieu of Statutory Holidays which occur on scheduled days off, or the premium portion of time worked on Statutory Holidays, in accordance with the provisions of this Agreement, shall be compensated for such banked time at the rate of pay in effect at the time of payment or at the time of utilizing such deferred time.
- (e) An Employee who elects the deferral option referred to in subsection (c) above, may receive payment for all or a portion of such deferred time by providing the Company with a minimum notice of seven (7) calendar days. The Employee may also elect to take such time off as Paid Extended Absences of one (1) continuous work week block or more. In the event an Employee uses such time as a paid extended absence, and the Employee makes application for Salary Continuance during this period, the balance of the absence will be cancelled, and at the conclusion of Salary Continuance entitlement the Employee will return to regular employment.
- (f) It is further agreed that where an Employee elects to take deferred time off, in accordance with the above provisions, the Company will replace such Employee from the Union Hall.
- (g) An employee who has received approval to use his deferred bank as Early Retirement Credits must have submitted a signed notice of resignation with a confirmed date of termination of service. Once this letter is accepted and the use of the Early Retirement Credits has commenced, the employee will not be entitled

to work for the Company. In the event of extenuating circumstances the use of Early Retirement Credits may be cancelled with mutual agreement of the Company and the Union.

# 7.07 Book-offs

- (a) It is understood and agreed that an Employee may book-off from an assigned shift by providing the Company with notice of the intent to not work such shift or shifts. The notice shall be appropriate to allow for a replacement to be obtained at the commencement of the Employee's assigned shift(s). The maximum number of consecutive working days an Employee shall be able to book-off shall not exceed six (6) at any one (1) time.
- (b) All book-offs, in accordance with the above provision, and the provisions of Article
   20, section 20.03 and Article 22, section 22.01, shall be replaced from the Union
   Hall.

# 7.08 Overtime Callouts

Overtime callouts may be required to provide additional personnel on either scheduled or nonscheduled shifts as per, but not limited to the following:

The categories where overtime callouts may occur include the following:

<u>Welders</u> <u>Electricians</u> <u>Lubemen</u> <u>Printman</u> <u>Technicians</u> <u>Gearman</u>

- (a) Employees called out on overtime to replace absent Employees or to perform extra work shall be assigned work as follows:
  - i. <u>Replacement (Absent Employee)</u>

An employee called out to replace an absent employee shall perform all functions of the job being replaced.

#### ii. Scheduled Shift

An employee may be called out to a specific job category and the work to be performed shall be stated in the callout. Where the work called out to perform has not been started due to High Winds, the Employee called out may be assigned Shop Work for the balance of the shift. This provision shall not apply to Employees called out to perform Belt Splicing.

iii. Non Scheduled Shift

An employee may be called out to perform work, as stated in the overtime request, on a specific machine (for a Warehouseman the work shall be job specific). When the specified work is finished the callout shall be completed. A Cleanup Employee may also be called out to perform work on chutes/transfers or belt lines or roads and when the work is finished the callout shall be completed. However, where cleanup requirements are necessary due to inclement weather or product quality, involving chutes and/or transfers, the employee shall be called out for the entire shift.

- iv. "Extra" Regular Employees
  - (a) <u>Where "Extra" Regular Employees are utilized for three (3) days or less, in</u> <u>accordance with R2-94, the work will not be offered to other Regular Employees</u> <u>and the Regular Employees called out will perform the work in accordance with the</u> <u>provisions contained herein.</u>
  - (b) In addition to the above, where "Extra" Regular Employees are utilized, the down rules, contained in Article 24 of the Collective Agreement, in the Operations or Maintenance Department where the "Extras" are assigned, shall not apply and the regular scheduled crews in the Department will be manned at the full normal complement.

- v. <u>Miscellaneous</u>
  - (a) <u>When cleanup is being called on overtime for work on a stacker, the particular</u> <u>stacker does not have to be identified until the commencement of the shift.</u>
  - (b) When lube is being called for overtime work, the equipment will be identified at the commencement of the shift.
- (b) Employees called out to perform work as a result of a breakdown shall be assigned work as <u>follows:</u>
  - (i) Maintenance Personnel

The callout shall be defined as the repairs required, as a result of the initial breakdown, to allow the specific piece of equipment to be put back into operation. The callout shall be finished when the repairs are completed and the equipment is operational.

(ii) Shift Operator

The callout shall be defined as work to be performed, as a result of the initial breakdown, of a specific piece of equipment, or specific jobs for the Maintenance Department. When the specific work is finished the callout shall be completed.

(iii) Cleanup Operators

The callout shall define the areas on one (1) piece of equipment where cleanup work is to be performed. When the specific work is finished the callout shall be completed.

# 7.09 Personal Days

The Company will provide employees with six personal guaranteed book-off days per calendar year that may be taken at the employee's discretion. Notwithstanding any article in the Collective Agreement, there will be no requirement to replace the employee on overtime. The use of a guaranteed book-off must be the initial notification or request for time off for the day(s) in question and must be requested a minimum two hours prior to the scheduled shift start time. An employee may use their deferred bank or DIL bank to request payment for such day(s) used, and may go into arrears of up to six days in his DIL bank.

# **ARTICLE 8 HOLIDAYS**

# 8.01 Days

(a) The following days shall be considered holidays:

| 1. New Year's Day | 7. B.C. Day         |
|-------------------|---------------------|
| 2. Family Day     | 8. Labour Day       |
| 3. Good Friday    | 9. Thanksgiving Day |
| 4. Easter Monday  | 10. Remembrance Day |
| 5. Victoria Day   | 11. Christmas Day   |
| 6. Canada Day     | 12. Boxing Day      |

(b) In addition to the above holidays, an Employee shall be entitled to his birthday as a paid holiday. For regular Employees and Apprentices, this holiday can be taken at anytime that is mutually agreeable to the Employee and the Company during the calendar year in which the birthday falls.

Should this holiday occur during a dispatch for other qualifying Employees, they will be paid as per the calculation in Article 8.02 (b) at their normal rate of pay.

- (c) Any holiday declared by the Province or Federal Governments.
- (d) Unless agreed otherwise, the terminal will not operate the scheduled shifts, (as outlined in <u>Clause 6.05</u>), that commence on Christmas Day. However, the provisions of Letter of Agreement R1 85, and in particular sections 7 and 8, continue to apply.

# 8.02 Eligibility

(a) Except as herein otherwise provided, to be entitled to any salary for the abovenamed holidays not worked, an Employee must meet the following qualifications:

- (i) He has qualified for pay purposes to receive holiday pay under the terms set forth in the Canada Labour (Standards)
   Code as General Holidays; and
- (ii) The Employee has worked a minimum of ten (10) days or eighty (80) hours in the thirty (30) calendar days immediately preceding the general holidays.
- (iii) It is agreed that for regular Employees, hours paid shall be considered as hours worked for the purpose of meeting the eligibility requirements of (ii) above, provided the Employee returns to work on or immediately prior to the holiday. This provision shall also apply to Temporary Employees during their dispatch, in accordance with point 10 of Appendix "B".
- (b) Employees entitled, under (a) above, to receive salary payments for the above named holidays not worked, shall be paid as follows:

| <u>Time worked in 30 calendar days</u><br>immediately preceding general holiday | Salary Payment             |
|---|----------------------------|
| 15 days/120 hrs   | Full day's salary          |
| 14 days/112 hours   | 14/15 of full day's salary |
| 13 days/104 hours   | 13/15 of full day's salary |
| 12 days/96 hours  | 4/5 of full day's salary   |
| 11 days/88 hours  | 11/15 of full day's salary |
| 10 days/80 hours  | 2/3 of full day's salary   |

- (c) An Employee scheduled for work on any of the above holidays but who is unable to, or who fails to arrange for a replacement satisfactory to the Company, will have the hourly equivalent salary rate for the day deducted from his regular salary.
- (d) When a holiday falls on an Employee's scheduled day off, the Employee at the option of the Company, will be entitled to and shall be granted either the straight

time salary equivalent for that day or a day off in lieu thereof at a mutually agreeable time, however, not later than the Employee's next annual vacation. In addition, the Employee may elect to defer such days in lieu in accordance with the provisions of Article 7, section 7.07 of this Agreement. In the final year of the Collective Agreement, employees will be permitted to receive the pay for the general holiday or be entitled to a day off in lieu. At the end of the final year, the option reverts to that of the Company.

- (e) No salary payment shall be paid to any Employee for any holiday which occurs subsequent to the thirty (30) calendar day period following a layoff, leave of absence or retirement. In the event of termination, except for cause, an Employee shall be paid for any Holiday which occurs in the fifteen (15) calendar day period immediately following the date of termination.
- (f) If an Employee is entitled to holiday pay benefit from any other source, Westshore will not duplicate that benefit.
- (g) Holidays worked will be paid as set forth under the Overtime Article 7, subsection7.03 (c).
- (h) (i) The Company agrees that when the site operates on a holiday, all scheduled Employees shall be required to work in accordance with their assigned schedule.
  - (ii) Where a crew or crews of maintenance Employees are required to work on a shift and/or shifts on a holiday, the number of maintenance Employees required to work on each crew shall be in accordance with the manning provisions of Article 24, subsection 24.01 (b). A crew for the purpose of this section is defined as either the Electrical Day crew, the Lube crew, the crew assigned to the MA-ME or M-5 schedule on the day shift, the crew on the MA-ME schedule assigned to the night shift but scheduled to work the day

shift in the absence of such night shift, or the crew assigned to the MA-ME schedule on the afternoon shift.

- (iii) Where work is required to be performed on a holiday, as set out in (ii) above, other related Employees may be required to work by the Company as deemed necessary.
- (i) The Company shall advise the Employees a minimum of seven (7) calendar days in advance of any paid holiday when such Employees are not required to work their scheduled shift on any of the paid holidays referred to in subsection 8.01 (a) of this Article.
- (j) In the event the Company requires an Employee to work New Year's Day and the Employee books off such day, the book off will not affect the Employee's eligibility for Holiday Pay for such day, provided the Employee meets all other eligibility requirements for Holiday Pay as referred to in this section.

# **ARTICLE 9 VACATIONS**

### 9.01 Vacation Entitlement

#### (a) Vacation Entitlement

| Years of Continuous Service | Length of Vacation |
|-----------------------------|--------------------|
| 1-4                         | 3 weeks            |
| 5-9                         | 4 weeks            |
| 10-14                       | 5 weeks            |
| 15-19                       | 6 weeks            |
| 20+                         | 7 weeks            |

#### (b) Vacation Entitlement - Leaves of Absence

- An Employee granted a leave of absence in accordance with the provisions of Article 12, subsection 12.05 (a) of this Agreement (twenty-one (21) consecutive working days or less) shall receive no reduction in vacation entitlement as a result of such absence.
- (ii) An Employee granted a leave of absence in accordance with the provisions of Article 12, subsection 12.05 (b) of this Agreement (in excess of twenty-one (21) consecutive working days or more) shall have vacation entitlement adjusted, for that calendar year only, based on the following formula:

<u>Gross Regular Earnings for the Year X Vacation taken Percent</u> Regular Weekly Pay - Vacation weeks taken

= Amended Vacation Entitlement

(iii) In the event application of the above formula results in a negative balance, the Employee's vacation entitlement in the subsequent year shall be reduced by such negative balance to a maximum of three (3) weeks only. Any negative balance will be rounded to the lowest whole number of vacation weeks.

### (c) Vacation Entitlement - Salary Continuance, L.T.D. and W.C.B.

- (i) An Employee returning to work following an absence on Salary Continuance, or a combination of Salary Continuance and L.T.D., or Worker's Compensation Benefits (W.C.B.) shall receive no reduction in vacation entitlement in that calendar year. Where an Employee's vacation entitlement exceeds the balance of time left to December 31st in that calendar year, the Employee will be granted vacation entitlement to December 31st and the balance of such Employee's vacation entitlement shall be paid out as a Vacation Pay Adjustment in accordance with the provisions of section 9.02 of this Article.
- An Employee absent an entire calendar year on L.T.D. or
   W.C.B. shall receive no vacation entitlement for that calendar year.

- (iii) An Employee returning to work following an absence on L.T.D., which commenced on or before January 1 of that calendar year, shall receive the lesser of full vacation entitlement or the balance of the time available from the date of return to work to December 31st of that calendar year.
- (iv) It is understood and agreed that an absence on Salary Continuance shall be considered as regular earnings for the purposes of vacation pay adjustments for the vacation year in which the absence occurred.

### 9.02 Vacation Pay Rates

Pay for vacations will be made at an Employee's rate of pay at the time vacation is taken, or, depending upon his vacation entitlement, at the rate of 6% (3 weeks), 8% (4 weeks), 10% (5 weeks), 12% (6 weeks) or 14% (7 weeks) of his earnings during such calendar year, whichever is the greater. Percentage calculations and payments will be made by February 1st of the following year.

# 9.03 Conditions

- (a) An Employee is eligible for a one (1) week vacation after six (6) months of unbroken service. This one (1) week vacation shall be taken in the same calendar year in which it is earned. If less than one (1) full week remains between the end of the six (6) months period of service and the end of the calendar year, the Employee is only entitled to the remaining portion of the week as vacation.
- (b) Subject to the other provisions of this Article, an Employee who is hired during the preceding calendar year is eligible for the vacation, as set out in section 9.01, after completion of one (1) year of unbroken service provided that the vacation can be completed within the calendar year. If the time between the end of the Employee's first year of service and the end of the calendar year is less than the Employee's

vacation entitlement, the Employee is entitled to only the remaining portion of the year as vacation.

- (c) In determining eligibility for vacation benefits, under section 9.01 of this Article, an Employee's service is computed in full calendar years subsequent to the first calendar year. For the first year of service an Employee who works a combination of eight hundred (800) or more hours in that calendar year shall receive a full year's credit for that year upon completion of the second calendar year of employment. An Employee who works less than the required eight hundred (800) hours in the first calendar year shall receive no credit for the time worked in that calendar year.
- (d) After one year of continuous service with the Company and for the sole purpose of calculating vacation entitlement, regular Employees shall be granted full credit for continuous I.L.W.U. service accumulated immediately prior to employment with the Company, provided satisfactory proof of such service is submitted to the Company. Thereafter vacation entitlement shall be at the rates set out in section 9.01 of this Article. Such service shall not be considered nor computed for any other purpose whatever unless specifically provided for elsewhere in this Agreement.
- (e) The I.L.W.U. service credit, referred to in subsection (d) above, shall be interpreted as those years in which the Employees worked five hundred hours (500) or more.
- (f) A week of vacation is seven consecutive days.
- (g) A "calendar year" is January 1 through December 31.

### 9.04 Vacation Pay

- (a) An Employee shall receive six percent (6%) (or where applicable the B.C.M.E.A. rate if greater) of his gross earnings from his date of hire until December 31st of the first calendar year, less any vacation pay received for that period. Thereafter vacation entitlements shall be in accordance with section 9.01 of this Article.
- (b) Commencing January 1st following the year of hire, and each calendar year thereafter, regular Employees are entitled to vacation pay which shall be paid an

amount equivalent to the Employee's normal salary which he would have received if he had worked on a straight time basis during his vacation period. Such vacation pay will be paid immediately prior to the Employee taking his vacation if the Employee so requests.

- (c) Temporary Relief Employees and Daily Dispatch Employees shall be entitled to a percentage of their gross earnings in lieu of vacation entitlement with pay. The percentage of gross earnings shall be based on their B.C.M.E.A. years of service and paid at the time of their termination or at December 31st of the calendar year in which they worked for the Company, whichever first occurs. In the event a Temporary Employee fails to earn a B.C.M.E.A. year of service, as a result of being employed by the Company, and works for the Company for eight hundred (800) hours or more in that year, then such Employee shall be granted a year of service for the purposes of determining the appropriate percentage rate and calculating vacation pay.
- (d) Vacation pay for regular Employees, whose employment is terminated for whatever reason, shall be computed as follows:
  - (i) Those who have not completed their first calendar year of continuous service and have not received any vacation, will upon termination be paid six percent (6%) (or where applicable the B.C.M.E.A. rate if greater) of gross earnings less any vacation pay already received for that period.
  - (ii) Those Employees who have completed their first calendar year of service and have not reached their anniversary date, shall upon termination, receive six percent (6%) (or where applicable the B.C.M.E.A. rate if greater) of gross earnings from January 1st following the year of hire until termination, less vacation pay already received for the period from January 1st following the year of hire until termination.

- (iii) Those who have passed their first anniversary date in the calendar year of their termination will receive the appropriate percentage(s) of gross earnings for the period worked from January 1st following the year of hire to the date of termination less any vacation pay already received for that period.
- (iv) Those who have received their vacation pay and who subsequently terminate prior to December 31st shall have deducted from their final pay cheque the difference between vacation pay received and the appropriate percentage of gross earnings for the period from January 1st to the date of termination.

### 9.05 Holidays During Vacation Period

If a statutory holiday falls within an Employee's authorized vacation, the Employee will be permitted to take one (1) additional work day off with pay in conjunction with his regular vacation provided such arrangements are made in writing immediately prior to the Employee taking his vacation.

### 9.06 Time of Vacation

- (a) Each eligible Employee shall be required to specify the vacation period(s) desired a minimum of thirty-seven (37) calendar days prior to the commencement of such vacation.
- (b) Vacation requests submitted, in accordance with the above, that meet all other requirements of this Article shall be approved by the Company on or before the thirtieth (30th) calendar day prior to the commencement of such vacations. Where vacation requests are submitted with a lesser notice period than that referred to in the above provisions, such requests will be considered on a first come first considered basis.

- (c) An Employee requesting an exception to the provisions contained in this section must complete a "Vacation Exception Request" and submit same to the Manager of Human Resources or his designate. All such requests shall be subject to mutual agreement of the Company and the Union, and shall be approved or denied within the ten (10) calendar days of receipt of such request by the Company and the Union.
- (d) Vacations will, so far as possible, be granted at times most desired by Employees (longer service Employees will be given preference as to choice) but the final right to allot vacation periods, and the right to change such allotments, are exclusively reserved to the Company in order to ensure the orderly operation of the facilities. It is, however, understood and agreed that the Company's right to change an Employee's approved vacation shall only apply where trained and qualified replacements are not available after exhausting all other provisions contained in this Agreement, and existing Letters of Agreement, related to such replacements.
- (e) Notwithstanding the foregoing, it is understood and agreed that the approved vacation schedule shall not reduce the number of regular Employees scheduled to work at any one (1) time by less than three (3) regular Employees on each Operations crew and two (2) regular Employees on each Maintenance crew.
- (f) Subject to the provisions of this Article, Employees must complete their vacation in the calendar year.
- (g) All Employees shall be allowed to take their vacations in one (1) week splits; provided management approval is obtained and the scheduling of such vacation will not interfere with the operation of the facilities.
- (h) At least thirty (30) calendar days must elapse between a vacation for one (1) calendar year and a vacation for the succeeding calendar year.
- (i) In the second calendar year of employment, no Employee shall be entitled to take his vacation before April 1st or his anniversary date, whichever is the later.

# 9.07 Vacation Lists/Schedules

The Company shall post a schedule each month indicating vacations approved for the following month. In addition, the Company will post a list each month indicating each Employee's vacation entitlement, vacation weeks taken to date, and vacation weeks outstanding. The Company shall provide the Union with a copy of such lists/schedules.

### 9.08 Vacation Scheduling & Disability

- (a) <u>At such time an Employee commences his approved vacation the Employee will</u> <u>continue on such vacation without change.</u>
- (b) In the event that an Employee has scheduled his vacation and is subsequently placed on a waiting list for admittance to a hospital, the Employee may have his vacation re-scheduled subject to the following: If the Employee is admitted to a hospital, prior to the commencement of his approved vacation entitlement, then the Employee may request and may be granted permission to re-schedule his vacation. If the request is not made within two (2) calendar days of admittance to the hospital, or prior to the start of the approved vacation entitlement, then the Employee will continue on vacation as previously scheduled.
- (c) <u>Scheduled days off that occur between consecutive vacation weeks are considered</u> <u>as part of the Employee's vacation schedule.</u>
- (d) If an Employee is disabled as a result of an accident or illness immediately prior to the commencement of scheduled vacation entitlement, then such Employee may request and be granted permission to re-schedule his vacation entitlement. This provision shall only apply if the Employee notifies the Company, prior to the start of his vacation entitlement, of his request to re-schedule and further provided that the Employee gives to the Company, without undue delay, satisfactory proof of disability.
- (e) <u>If, pursuant to the above conditions, an Employee becomes disabled or sick and</u> <u>re-schedules his vacation entitlement, then the Employee will receive salary</u>

continuance in accordance with the provisions of Article 11 of the Collective Agreement.

- (f) If an Employee is hospitalized, and has an ongoing disability as a result of an illness or injury sustained during their scheduled vacation time, then such employee may request permission to re-schedule any remaining full vacation week(s) entitlement that were scheduled and approved to run concurrently with the vacation week during which the Employee was hospitalized. Any rescheduling of vacation will be subject to the mutual agreement of the Union President and Director of Human Resources, or their designates.
- (g) <u>The employee may be required to provide documentation to support the claim of</u> <u>hospitalization and ongoing disability on a form supplied by the Company.</u>
- (h) Any re-scheduling of vacation entitlement shall be subject to mutual agreement of the Company and the Employee and shall not result in any change in approved vacation periods for other Employees regardless of seniority.

# 9.09 Vacation Payout on Retirement

- (a) Employees who retire between the dates January 1 and June 29 inclusive of a calendar year will receive a vacation payout calculated on the following basis:
   A percentage (i.e. at the percentage rate of his vacation entitlement) of his year to date earnings (including severance payment) less any vacation pay received to date.
- (b) Employees who retire between the dates June 30 and December 31 inclusive of a calendar year will receive a vacation payout calculated on one of the following two basis:
  - (i) the number of outstanding weeks of vacation times (x) thirty-six (36) hours, times (x) the hourly rate in effect at the time, or

(ii) a percentage (i.e. at the percentage rate of his vacation entitlement) of his year to date earnings (including severance payment) less any vacation pay received to date.

The Employee will be required to notify the Company of the method he wishes to have his vacation payout calculated at least two (2) weeks prior to his last day of work.

# **ARTICLE 10 INSURANCE BENEFITS**

### **10.01 Benefit Plans**

The Company shall provide all Employees, except daily dispatch Employees, with the following benefits, without contribution or deduction from those Employees; and in accordance with the provisions of section 10.06 of this Article.

### (a) *Medical Insurance and Extended Health Benefits*

Medical Insurance Plan (MSP) and Extended Health Plan shall provide coverage for the Employee, his spouse and their dependent children under the age of 21 years. The Extended Health Plan shall provide for one hundred percent (100%) reimbursement of prescription drug expenses, however the existing deductible and integration of benefits with the Pharmacare plan will continue to apply. The overall extended health care maximum is \$400,000 lifetime. If an employee reaches \$300,000, the parties will meet to discuss the individual circumstances and if necessary create special trust fund payments over and above the existing contributions to address any funding shortfalls.

### (b) **Dental Care Plan**

The Plan shall include orthodontia, and cover the Employee, his spouse and their dependent children under the age of 21 years. Orthodontia coverage to be three thousand dollars (\$3,000.00) lifetime maximum. The basic Dental Plan (Plan "A") shall be extended to provide coverage to a regular Employee's dependent children to age twenty-five (25), provided that the dependent children reside at home, and further provided that the dependent children are in full-time attendance at a recognized University, College or Technical Institute in the Province of British Columbia. The Dental Plan B shall provide coverage to a maximum of seventy-five percent (75%).

### (c) **Accidental Death and Dismemberment Insurance**

The Plan shall provide death benefits equal to twice the Employee's basic rate of annual salary rounded to the next highest \$1,000.

#### (d) Long-Term Disability Insurance Plan

The Plan shall provide coverage for eligible Employees who are totally disabled for a period of time in excess of six (6) months. During which time Employees continue to be covered for their Medical and extended health insurance at Company expense.

### (e) Life Insurance

Life Insurance will be for a sum of three (3) times the Employee's basic rate of annual salary rounded to the next highest \$1,000 of coverage.

### (f) **Plan B Dental Mechanic Services**

- (i) Coverage for partial and complete dentures but not more often than once every five (5) years.
- (ii) The Dental Care Plan, upon acceptance of a claim under Plan
   B (Dental Mechanics) shall pay one hundred percent (100%)
   of the appropriate fee in the schedule of fees approved by
   the Dental Care Plan Carrier. In addition, if there is a
   surcharge, the Plan will assume this amount to a maximum
   of twenty percent (20%) of the appropriate fee upon receipt
   of an invoice from the Dental Mechanic.
- (iii) The preceding arrangement will only apply to the services of a Dental Mechanic and all other provisions or terms of the Dental Care Plan shall remain in effect.

### (g) Eyeglass Coverage Program

The present Optical Program shall be amended, effective February 1, 1997, to provide a maximum of two hundred and seventy-five dollars (\$275.00) every year

to an Employee required to purchase corrective lenses or frames. Effective January 1, 2008, this amount will increase to three hundred and seventy five dollars (\$375). Effective February 1, 2012, the annual amount will increase to four hundred and fifty (\$450) dollars. Effective February 1, 2016, vison care will increase to \$1000 every 24 months for employees and dependents. Retirees will receive \$900 every 24 months. This program shall only cover regular Employees of the Company.

(h) The Company will provide a direct pay drug card for all regular employees and retirees.

The drug card will be subject to the terms and conditions of the policyholder and card issuer.

Employees who quit their employment or are terminated will be required to return their card, and will be responsible for any charges incurred after their resignation/termination. Any such charges may be deducted from their final cheque.

The card will require the employee to receive a generic drug where such drug is available.

# 10.02 Widow's <u>Benefits</u>

The Company will also provide Medical Insurance and Extended Health and Dental Care Benefits currently in effect to a surviving widow and her dependent children under the age of twenty-one (21) years until the spouse obtains full time employment or remarries or obtains coverage elsewhere whichever occurs first. All such benefits will cease if and when the widow or children become ineligible for B.C. Medical and Extended Health Benefits or Dental Benefits. Children born within 9 months following the Employee's death will be considered dependent children.

# **10.03 Explanatory Brochures**

The Company will provide explanatory brochures for each area of insurance coverage but it is understood and agreed that all insurance plans are subject to the terms and conditions of the policy in each case and the Company's obligation ceases with the payment of premiums. In the case of any claim, it is the duty of the Employee claiming to ensure that the terms and conditions of the policy are met. The Company will assist the Employee in every reasonable way, but no such assistance will be deemed or considered as increasing the obligations of the Company under this Article.

### **10.04 Insurance Plan Policies**

The Company shall provide the Union with copies of the Insurance Plans and Policies referred to in this Article, and each will make copies available for inspection at their respective offices. Neither the Company nor the Union will be deemed to be the agent of any plan or insurer. The Company will advise and discuss with the Union any changes prior to initiating any such changes in insurance carriers with respect to the insurance plans referred to in this Collective Agreement.

# **10.05 Eligibility**

- (a) Daily dispatch Employees are not entitled to any benefits under this Article. Regular Employees and temporary relief Employees are entitled, from the date of their hire, to the benefits set out in Article 10.01 (a), (b), (c), (e), and (f). Regular Employees and temporary relief Employees shall become entitled to the benefits under Article 10 subsection 10.01 (d) upon the completion of twenty-two (22) shifts worked with the company provided that:
  - (i) The twenty-two (22) shifts are within the ninety (90) day period immediately preceding the date of claim;
  - (ii) The Employee has completed one (1) year's continuous I.L.W.U. service; and

- (iii) The Employee has passed the physical examination under Article 13.
- (iv) Shifts worked by regular Employees and temporary relief
   Employees as daily dispatch Employees in the ninety (90)
   day period immediately preceding the day of claim shall be
   included in the calculation of the twenty-two (22) shifts
   worked with the Company.
- (b) The requirements of Article 10, subsections 10.05 (i) and (iii) will not apply to regular Employees who have completed twelve (12) months of continuous service with the Company.

# **10.06 Duplication of Coverage**

To reduce duplicating of coverage for benefits under Article 10, subsections 10.01 (a) and (b), the Company and the Union may make alternate arrangements for any temporary Employee for the purposes of granting coverage for those benefits. The parties must mutually agree before such arrangements become effective. The parties agree to establish a system to provide for the exchange of the data in respect to dental coverage for temporary relief Employees, in order to minimize the problem of double payment.

# **ARTICLE 11 SALARY CONTINUANCE**

### 11.01 Scope

Subject to section 11.03, the provisions of this Article shall cover regular and temporary relief Employees absent from work as a result of personal disability caused by accident or illness and includes policies pertaining to Salary Continuance during such absences.

### 11.02 Sick Leave Allowance

- (a) The Company shall have the right to employ a doctor to determine the nature and extent of disability or sickness and the duration of such disability or sickness at any time an Employee requests benefit under the provisions of this Article. Under normal conditions and where practical, the Company will select a doctor from the designated medical clinic referred to in Article 13 of this Agreement. The Company may request proof of disability or sickness by way of a certificate from the Employee's own doctor.
- (b) The following table shows the entitlement of an eligible Employee under this Article to have salary continued at the Employee's basic salary rate in the event of personal disability or sickness.

#### Length of Continuous Service and Maximum Salary Continuance

| At least one (1) month but less than one (1) year           | - | balance of month plus one (1) month    |
|---|---|--|
| At least one (1) year but less than five (5) years          | - | balance of month plus two (2) months   |
| At least five (5) years but less than ten (10) years        | - | balance of month plus three (3) months |
| At least ten (10) years but less than fifteen (15) years    | - | balance of month plus four (4) months  |
| At least fifteen (15) years but less than twenty (20) years | - | balance of month plus five (5) months  |
| At least twenty (20) years or more                          | - | balance of month plus six (6) months   |

- (c) An Employee who returns to work following an absence on Salary Continuance, and is subsequently absent as a result of a different illness or injury, shall be entitled to the full benefit entitlement under section 11.02 of this Article.
- (d) An Employee who has utilized all of the benefits available under this Plan, returns to work, and has a reoccurrence of the same illness or disability arising out of an injury, and is not entitled to L.T.D. benefits, shall be entitled to one hundred percent (100%) of the Employee's basic salary during any subsequent absences for the same reason for a maximum of six (6) months. It is, however, understood that an Employee's entitlement shall not exceed the provisions of subsection 11.02 (b).
- (e) If an Employee returns to work in accordance with subsection (d) above, was entitled to L.T.D. at the date of return to work, and works two (2) continuous weeks or more, and has a reoccurrence of the same illness or disability, such Employee shall be entitled to continuance of the above benefit for a maximum period of six (6) months.
- (f) An Employee who returns to work from an absence on Salary Continuance or L.T.D. and, following a period of not less than two (2) years from the date of return to work, is subsequently absent as a result of the same illness or injury shall be entitled to the full benefit entitlement under section 11.02 of this Article.
- (g) In addition to the above, an Employee in receipt of Long Term Disability Benefits who returns to work shall be entitled to the benefits referred to in subsection (d) above for any incidental absences as a result of a reoccurrence of the same illness or disability arising out of an injury, unless such period of absence is covered by, or the Employee is again eligible for, Long Term Disability Benefits.
- (h) The Parties agree that successive periods of absence of one (1) month or less for the same illness or disability shall not be accumulated to determine the maximum entitlement under the provisions of subsection 11.02 (b) of this Article.

# **11.03 Eligibility**

- (a) For the purposes of determining the amount of benefits under this Article, Employees who have I.L.W.U. continuous service at time of hire will be given an interim credit for previous continuous service with the I.L.W.U., provided such service was continuous and immediately prior to employment by the Company. The absolute maximum credit given for such I.L.W.U. service for the purpose of this Article at the time of hire will be two (2) years. Combination of credit for previous I.L.W.U. service, if any, and continuous service with the Company, will constitute "length of continuous service" for the purpose of computing applicable benefits under this Article for eligible Employees who are absent from work as a result of personal disability caused by accident or sickness for the first year of employment.
  (b) Upon completion of twelve (12) months employment with the Company, regular
- (b) Upon completion of twelve (12) months employment with the Company, regular and temporary relief Employees shall receive credit for all previous I.L.W.U. service (less any credit for I.L.W.U. service in subsection (a) above), provided such service was continuous and immediately prior to employment by the Company. Such service shall not be considered, nor computed, for any other purpose whatever unless specifically provided for elsewhere in this Agreement, without reference to this Article. Continuous service for the purposes of this provision shall be as defined in Article 15, subsection 15.04 (c).
- (c) To be eligible for benefits set out under this Article, regular Employees and temporary relief Employees must comply with the following conditions:
  - They must complete twenty-two (22) shifts worked with the Company in the ninety (90) day period immediately preceding the date of claim; and
  - (ii) They shall have passed the physical examination set out in Article 13.

- (d) For the purposes of calculating shifts worked in subsection (c) above, shifts worked by regular Employees and temporary relief Employees as daily dispatch Employees shall be included.
- (e) The requirements of subsection (c) above will not apply to regular Employees who have completed twelve (12) months of continuous service with the Company.

# 11.04 Amounts Payable

- (a) It is the intention of this Article to ensure that eligible Employees suffer no reduction in pay for the period of their entitlement, such period not to exceed six
  (6) months at full salary. Worker's Compensation Benefits accruing to the Employee for work incurred injury or disability will be deducted from the Employee's entitlement under this Article, as will be benefits accruing from any other insurance.
- (b) No deduction will be made as a result of any demotion or reduction in force which occurs during the period of entitlement so long as the Employee remains disabled and performs no work elsewhere.
- (c) There will be no duplication or pyramiding of salary continuance benefits and vacation benefits.
- (d) Employees shall be entitled to benefits under this Article for Medical and Dental specialists appointments only, where such appointments cannot be arranged on an Employee's regular days off.

# **11.05 Long Term Disability**

(a) If an Employee is eligible for Long Term Disability Benefits in accordance with the provisions of Article 10, and remains disabled and eligible for a period exceeding six (6) months, calculated from the first day of absence for disability, then such Employee shall be entitled to receive the benefits provided by such Long Term Disability Insurance Plan, subject to the terms and conditions of that Plan. The

Company's direct obligations under Article 11 shall cease unless and until the Employee works again under this Agreement or a renewal of it.

- (b) During the term of this Agreement the Company agrees that Employees in receipt of Long Term Disability Benefits shall receive the appropriate percentage of any scheduled salary increases in accordance with provisions of Article 25 of this Agreement.
- (c) Notwithstanding the above, once Long Term Disability Benefits commence for an Employee, any subsequent increase in C.P.P. disability payments will not offset any payments under this Plan.
- (d) Should an Employee be entitled to benefits under a self-insured plan, such benefits will not offset any payments under this Plan.
- (e) (i) The Long Term Disability Policy will provide a maximum benefit of 80% of pre-disability salary from all sources.
  - (ii) An Employee entitled to receive Long Term Disability benefits shall receive sixty percent (60%) of the Employee's regular salary immediately prior to disability, less income from any other sources that exceeds eighty percent (80%) of the Employee's pre-disability salary, subject to all other applicable conditions contained in the Agreement.
- (f) The Company will advise and discuss with the Union any rehabilitation program to be instituted by the L.T.D. Insurance Carrier to assist the Employee in returning to active employment at Westshore.

# 11.06 Sequence

- (a) An Employee who is eligible for benefits under this Article shall claim benefits and shall be entitled to receive benefits in the following order and priority:
  - (i) The Employee will first claim under section 11.02 of this Article;

- (ii) If the Employee has exhausted benefits for full salary continuance as set out in section 11.02 of this Article, and has exhausted accrued vacation benefits, and if at the inception of the disability the Employee was eligible for Long Term Disability Benefits under Article 10 of this Agreement, then such Employee shall receive salary continuance at half salary until the Employee returns to work or upon the expiration of six (6) months from the first day of absence for disability, whichever occurs first. Under special circumstances the mandatory use of vacation benefits may be waived or altered by mutual agreement.
- (iii) The above provision regarding use of vacation benefits shall not apply to an Employee who is entitled to full salary for six(6) months.
- (b) If the Employee has exhausted benefits under (i), (ii) and (iii) above, and if at the inception of disability the Employee was eligible for Long Term Disability Benefits under Article 10, then such Employee will receive the Long Term Disability Benefits subject to the terms and conditions of this Article and the L.T.D. Plan.

# **11.07 Claims Procedure**

An Employee requesting benefits under the Salary Continuance Program shall be required to comply with the following Claims Procedure:

- (a) A request for benefits under the program is to be made directly to the Foreman or Management person in the Employee's department at the time of notification of absence.
- (b) The following information shall be required and provided by the Employee to initiate a claim under the Program:
  - (i) name of the Employee and Badge number;

- (ii) telephone number, and
- (iii) reason for and expected duration of the absence.

# **11.08 Medical Certificates**

Employees requesting benefits under the Program shall provide the Company with a Medical Certificate from a qualified medical practitioner, on a form supplied by the Company, (the cost of obtaining such medical certificate shall be paid by the Company) in accordance with the following:

- (a) an absence by an Employee for a period of four (4) days or more;
- (b) any absence by a Temporary Employee, and
- (c) repetitive absences by any Employee where such absences indicate a pattern of absence has been established by such Employee.
- (d) an absence that may be considered a false claim.
- (e) if a medical note is required as per (a), (b), (c), or (d) foregoing, in order to qualify for salary continuance, the Employee must have visited the medical practitioner during the period of the absence from work for medical reasons. If the Employee is unable to do so, they must contact the Company to advise them of the reason for the failure to attend at a qualifying medical practitioner's office and obtain a medical certificate during their illness. Subject to having advised the Company of the late certificate, and providing a certificate, the Company will review the reason for the late certificate, and unless the Employee has had previous similar occurrences during the previous twelve (12) months, and the certificate in question is not unduly late, the Company will accept the certificate. Otherwise the Company will review the circumstances, and the reason for the late attendance at a medical practitioner's office, and determine whether or not they will accept the certificate.

Employees who do not agree with the Company's decision may request to present their position to the Health Committee. The Health Committee will have the authority, by consensus to amend the decision of the Company. It is understood and agreed that the Health Committee may alter subsection (a) above at their discretion based on the Employee's claims record.

# **11.09 Health Committee**

### (a) **Committee**

- (i) The Company and the Union have mutually agreed to establish a Health Committee comprised of two (2) representatives of the Company, two (2) representatives of the Union, one (1) alternate member for both the Company and the Union, and a Health Professional mutually agreed to by the Company and the Union. In addition, the Union shall designate one (1) Committee member to deal with Employees with abnormal absenteeism and such designated member of the Committee shall be paid for regular time off, where required.
- (ii) It is understood and agreed that the Health Professional shall be utilized by the Committee for the purposes of reviewing claims, under the provisions of Article 11 of this Agreement, and recommending reasonable means of rehabilitating such Employees or providing such other medical advice as may be appropriate under the circumstances. The Employee's own doctor shall have primary responsibility for treatment of the Employee's medical condition.
- (iii) The Company and Union representatives of the Committee must be present at all meetings of the Committee. The Committee shall meet at least once per month or at such

other times as deemed necessary by a majority of the members.

 (iv) All decisions and/or recommendations of the Committee will be by a majority of the members, however, the Health Professional shall not vote on such matters.

#### (b) **Purpose of Committee**

- (i) It is the intent and commitment of both the Company and the Union to ensure that application of the provisions of this Article are carried out in a fair and consistent manner.
- (ii) The Committee will carry out their functions in full recognition that the Salary Continuance Program is intended for the use of Employees unable to report for scheduled work as a result of a legitimate illness, disability or non-occupational injury.

### (c) **Committee Functions**

- (i) The Committee shall review all claims for benefits under this Article on a monthly basis to ensure such claims are properly administered in accordance with the provisions of this Article. The Committee shall also review the year to date record of absenteeism for sickness of all Employees at such meetings.
- (ii) It is understood and agreed that the Committee has been established to ensure that the benefits of Article 11 are not abused. In order to ensure the Committee's success, the Committee will have the authority to modify, alter, discontinue, or disallow benefits, or amend the eligibility

requirements for Salary Continuance Benefits with respect to any claim or claims it deems such action is appropriate.

- (iii) The Committee, in conjunction with the Health Professional and the Employee's own Doctor, shall recommend and approve any necessary action to ensure the early rehabilitation of an affected Employee.
- (iv) The Committee shall, in individual cases, establish specific criteria for Employees to meet when requesting benefits under this Article.
- (v) It shall also be a function of the Committee to advise the Parties of their "Duty to Accommodate".

It is understood and agreed that the Company and the Union shall meet to discuss and mutually agree on the conditions applying to any Employee referred to the Parties by the Committee in accordance with the above.

# 11.10 False Claims

- (a) It is just cause for dismissal for an Employee to submit a false claim under the provisions of this Article.
- (b) An example of a false claim would be claiming benefits under this Article while working or being employed at another job. Another example would be severe abuse of the Salary Continuance Benefits.

# 11.11 Modified Work

Prior to an employee reporting for work on modified duty, be it pursuant to a work related or non-work related injury or illness, the Union office will be supplied with the details of the modified work plan. A meeting between the parties will be held to review the return to work plan. The modified work plan should include the hours and days of work, any restrictions in duties, the duration of the program and the compensation. Employees are responsible to advise the Company as soon as possible as to their status as it relates to modified work.

Should an employee be absent for work such that it is anticipated he may exhaust salary continuance benefits, the Company will forward to the employee an application for Long Term Disability, (LTD), benefits approximately six weeks prior to the expiry of the salary continuance benefit. In addition to the application form, the Company will include an application form for Canada Pension Plan benefits and will advise the employee of the option to use any remaining vacation from his allotment prior to commencing LTD. A copy of the notification will be forwarded to the Union Hall. Approximately two weeks prior to the expiry of the salary continuance benefit, and only if the employee has neither returned the completed forms nor advised the Company of his intention to use all or a portion of his remaining vacation entitlement, the employer will attempt to contact the employee to follow-up on the LTD application. The Company will advise the Union Hall of the results of the follow-up attempt. The employee is responsible for the timely completion and return of the forms.

Employees who have been absent from work for medical reasons will not be accepted back to work prior to their anticipated return to work date without medical clearance.

# **11.12 Labour Relations Health and Attendance Review Committee**

A committee comprised of a representative from each of the parties to this agreement, normally the President of Local 502 and the Director of Human Resources, will meet on an as required basis to review matters of concern referred to it by the health committee or matters related to attendance of employees.

<u>All matters that the Health Committee cannot resolve, will, in a final effort to resolve the</u> <u>outstanding issue, be referred to this Committee, prior to either party availing themselves of other</u> <u>options, including the grievance procedure, which are available to them.</u> In addition to all remedies outlined under Article 11.09, the Committee may, when an employee has shown a pattern of missing work due to illness and working on the days off either before or after such an illness, impose restrictions on the employee's ability to work overtime on his days off.

# **ARTICLE 12 ABSENCES**

### **12.01 Salary Payment and Notification**

#### (a) **Salary Payment**

It is understood and agreed that an Employee who is absent from work will have deducted from his regular pay an amount equal to his salary equivalent for the period of such absence, unless the Employee is eligible for salary continuance pursuant to Article 11 hereof, or unless the Employee is given leave with pay, the granting of such leave to be entirely at the discretion of the Company.

### (b) **Notification of Absence**

If an Employee will be absent for any reason, he must notify the Company prior to the time he is scheduled to commence work. The Company may grant leave without pay at its sole discretion. In the event an Employee is absent from work without leave, then he may be subject to disciplinary action.

### **12.02 Union Business**

- (a) The Company shall allow designated Union representatives, who are regular Employees, paid leave of absence from their normal shift to attend regular monthly union meetings of Local 502. For General Monthly Meetings, five (5) representatives, and for Executive Monthly meetings, four (4) representatives will be granted such paid leave. The names of the designated Union representatives will be forwarded to the Company in writing by the Union in November of each year. Should an individual so named resign from his position, the Union will submit an alternate name.
- (b) The Union shall advise the Company at least twenty-four (24) hours in advance of any Employee(s) required to attend the above stated meetings.
- (c) Unless otherwise required by the Company, it shall not be necessary to replace Employees who attend Union meetings and are paid by the Company. In any event

where the Company elects to replace Employees such replacements shall be obtained from the Union Hall.

- (d) Where a regular Employee is elected to a full-time paid union position (President, Secretary-Treasurer, Business Agent or Dispatcher) and is granted a leave of absence, the Employee may elect to continue the benefits contained in Article 10, Sections 10.01, (a), (b), (c), (e) and (f) of this Agreement, provided the cost of such benefits are paid by the Employee or the Union. On completion of the term of the elected position the Employee shall be returned to his former position with full seniority rights and privileges.
- (e) In addition to the above, the Company shall arrange with the insurance underwriter to extend the Long Term Disability insurance benefit, referred to in Article 11, to cover not more than one (1) regular Employee at any one (1) time, for a temporary period not to exceed twelve (12) calendar months, when the Employee is on leave to serve full-time in one of the paid Union positions outlined in the previous paragraph. The Union recognizes and understands that any request to extend the L.T.D. coverage for more than twelve (12) months is subject to the sole discretion and approval of the Insurance Underwriter.
- (f) If a regular Employee is appointed as a temporary replacement for a full-time union position, as noted above, and the appointment is for not more than thirty (30) calendar days in a calendar year, the benefits, as referred to above, will continue during the absence with out change. Should the temporary appointment exceed thirty (30) calendar days in a calendar year, the benefits referred to above shall continue provided the cost of such benefits are paid by the Employee or the Union. This provision shall be limited to not more than two (2) regular Employees per calendar year.
- (g) All requests for leave of absence to fill a full-time union position, as referred to above, shall be made in writing at least seven (7) consecutive days prior to the commencement of such leave. The seven (7) days notice may be waived under emergency situations. The Company shall, where required in accordance with the

provisions of Article 24, obtain qualified replacements from the Union Hall for regular Employees accepting union positions.

- (h) Regular Employees who have been granted a special temporary leave of absence, without pay, to work for the I.L.W.U. Local 502 as Dispatcher, Secretary, Business Agent or President, pursuant to the provisions of Article 12 - Union Business are entitled to long term disability coverage subject to the following:
  - (i) Not more than two (2) persons would be on leave of absence from Westshore Terminals Ltd. to work for the I.L.W.U. Local 502 at any one (1) time.
  - (ii) The Union would reimburse the Company for the premium cost of the insurance plus any subsequent premium increases.
  - (iii) No person who is on leave of absence from Westshore Terminals to work for the I.L.W.U. local 502 would be permitted to "wolf".
  - (iv) Any claim for benefits under this arrangement would be adjudicated using the earnings and job description at Westshore Terminals Ltd. from which the claimant is on temporary leave.
  - (v) The elimination period for entitlement to this benefit is six (6) months.
  - (vi) After the elimination period has been served the claimant would be required to exhaust any accrued vacation entitlement prior to receiving Long Term Disability benefits.
  - (vii) The provision of this L.T.D. benefit to a person who is granted leave of absence from Westshore Terminals Ltd. to work for the I.L.W.U. Local 502, as outlined above, is a special arrangement and does not provide the Medical and Extended Health Insurance referred to in Article 10 of the Collective Agreement.

- (viii) Any claim for benefits will be subject to the terms and conditions contained in the L.T.D. insurance plan.
- (ix) Employees holding Union office who become regular employees after being elected to a full-time paid position in one of the following positions, President, Secretary Treasurer, Business Agent, or Dispatcher may be eligible to both the benefits available under 12.02 (d) and 12.02 (h). The Company may, at its sole discretion, determine to waive the requirements of the probationary period and/or pre-employment medical for the individuals the Union has requested be covered, and will endeavor to do so within 30 days of the request. If the Company agrees to waive the requirements, the individual will have the same rights afforded to him as other Union officials who are on leave from the Company. The number of employees entitled to such leave will be limited to four persons.

### 12.03 Bereavement Leave

- (a) When <u>a</u> death occurs in an Employee's immediate family (spouse, mother, father, mother-in-law, father-in-law, son, daughter, brother, sister or Employee's grandparents or grandchildren, <u>or any relative of the employee who resides</u> permanently with the employee or with whom the employee permanently resides the Employee, upon request, will be excused <u>for one complete work cycle if the absence is taken in one block, and if taken in two allotments, or spans two shift cycles, up to a maximum of five days. The days must be taken within six weeks following the latest date on which the funeral, memorial services or burial of the <u>deceased occurs.</u></u>
- (b) In the event of the death of an Employee's sister-in-law or brother-in-law, the Employee will, on request be excused for one (1) day to attend the funeral.
- (c) The Company shall, where required in accordance with the provisions of Article 24, obtain qualified replacements for an Employee granted bereavement leave in accordance with the provisions of Article 20.

# 12.04 Allowance for Jury Duty

- (a) A regular Employee who is called for jury service or subpoenaed as a witness (but not in his own defense) shall be excused from work without deduction from salary for the days on which he serves (which includes required reporting for jury duty when summoned, whether or not he is used as a juror). The Employee will present proof of reporting for jury service, or the jury, when summoned.
- (b) The above provision shall apply to a temporary relief Employee provided such temporary relief Employee was notified of the requirement for jury service after the commencement of his dispatch and the jury service occurs during his dispatch.
- (c) The Company shall, where required in accordance with the provisions of Article 24, obtain qualified replacements for an Employee granted jury duty or witness leave in accordance with the provisions of Article 20.

# **12.05 Leaves of Absence**

- (a) The Company shall grant an Employee an unpaid leave of absence, not to exceed twenty-one (21) consecutive working days in any calendar year, provided the Employee submits the request for such leave in writing a minimum of fourteen (14) calendar days in advance of the date the requested leave is to commence.
- (b) The Company shall grant an Employee an unpaid extended leave of absence not more than once every five (5) years, not to exceed six (6) consecutive calendar months, provided the Employee submits the request for such leave in writing a minimum of fourteen (14) calendar days in advance of the date the requested leave is to commence.
- (c) It is understood and agreed that an unpaid leave granted in accordance with this section shall be based on the Employee's seniority (crew) and the operational requirements of the Company.
- (d) During an approved unpaid leave of absence, in accordance with the provisions of subsection (a) above, the Employee shall continue to be entitled to the benefits

and provisions of Articles 9, 10, 15 and 19, but excluding Salary Continuance and Long Term Disability.

- (e) An Employee granted an unpaid extended leave of absence, in accordance with the provisions of subsection (b) above, shall continue to be entitled to the benefits and provisions of Articles 10 and 15, excluding Salary Continuance and Long Term Disability, provided the Employee pays the balance of the monthly premium costs to provide such benefits. In addition, at retirement, the Employee shall be granted pensionable service, up to a maximum of one (1) year of service, for any time absent on unpaid extended Leaves of Absence.
- (f) Eligibility for the Salary Continuance and Long Term Disability benefits, referred to in subsection (d) and (e) above, shall be reinstated on the first day the Employee is scheduled to return to work following the leave of absence.

# **ARTICLE 13 PHYSICAL EXAMINATIONS**

### **13.01 Medical and Physical Examinations**

- (a) The Company shall have the right to require that any applicant for employment shall pass a physical examination by a physician selected by the Company within thirty (30) days.
- (b) The Company shall also have the right to require periodic physical examinations for any Employee whose activities, in the sole opinion of the Company, affect the safety of other persons or of the property of the Company. Prior to making such decision the Company shall first discuss the matter with the Union.

# **13.02 Designated Medical Clinic**

The Parties have agreed to designate one (1) Medical Clinic for performing such periodic medical examinations as required during the term of this Agreement. The Company shall have the right to discharge any Employee who shall fail to pass any of said physical examinations. The Company shall bear the cost of all physical examinations required by it.

# **13.03 Salary Continuance Provisions**

The Company assures the Union that this Article is in no way meant to circumvent the provisions of Article 11.

# **ARTICLE 14 SAVING CLAUSE**

## **14.01 Invalid Provisions**

If any article, section, paragraph, clause or phrase of this Agreement shall by Provincial, Federal or other law, or by decision of any court, be declared or held illegal, void or unenforceable, the remaining portions of this Agreement shall continue to be valid and in full force and effect; and the parties shall thereupon immediately start to renegotiate that part or parts declared or held invalid.

# **ARTICLE 15 SENIORITY**

## **15.01 Determination of Seniority**

Seniority shall be determined on the basis of length of continuous service with the Company and shall commence thirty (30) days after date of employment. but when commenced shall date from the date of employment.

## **15.02 Loss of Seniority**

Continuous service shall be broken by:

- (a) Discharge;
- (b) Resignation or other termination of service by voluntary act of the Employee;
- (c) Absence due to lay off extending more than twelve (12) calendar months.

## **15.03 Reductions in Workforce**

- (a) When a reduction in the regular work force is to occur the Company shall give the Union and the Employees concerned fourteen (14) calendar days notice of the action to be taken where the layoff is temporary (Section 15.05 of this Article) and forty-two (42) calendar days notice where the layoff is permanent (Section 15.06 and 15.07 of this article). In all cases of layoff the Company and Union will meet to discuss the administration of such layoff.
- (b) It is further agreed that an Employee shall give the Company seven (7) calendar days notice of his intention to quit.
- (c) In the event of a permanent layoff it is understood and agreed that Employees will be rehired through application of the Posting Procedures.
- (d) Layoffs shall be effected in the following manner. The last Employee hired into a category shall be the first Employee laid off in that particular category.
- (e) For the purposes of this Article categories shall be defined as follows:

| Operations  | - Operators         |
|-------------|---------------------|
|             | - First Aid         |
| Maintenance | - Electricians      |
|             | - Mechanics/Welders |

(f) All laid off Employees will automatically be added to the Spareboard List, however, through attrition the list shall be reduced to its previous maximum number.

#### **15.04 Temporary Employees - Seniority**

For the purpose of attaining seniority and seniority related benefits (pension benefits, vacation entitlement and pay, salary continuance, and layoff/severance benefits), or B.C.M.E.A. service credits, applicable to Temporary Relief or Daily Dispatch Employees, the following shall apply with respect to determining credited service:

#### (a) **Pension Benefits**

- (i) For the above purposes Westshore service, a Temporary Employee's service shall be deemed non-continuous where there is a break in such service in excess of six (6) calendar days or more, provided the Temporary Employee does not become a Regular Employee.
- (ii) Effective January 1, 1998, upon permanent hire, a Temporary Employee's service prior to January 1, 1997 where there was not a break in Westshore continuous service of twenty-one (21) days or more, will be deemed Westshore service for the purpose of Pension Benefit entitlement.
- (iii) Effective January 1, 1998, upon permanent hire, a Temporary Employee shall be given credit for Westshore service for each calendar year in which the Temporary Employee worked twelve hundred (1,200) hours or more. The twelve

hundred (1,200) hours shall include, subject to the approval of the Health Committee, time absent during a dispatch period as a result of a Workers' Compensation Claim or while on Salary Continuance.

- (iv) A Temporary Employee, upon permanent hire, shall be provided a credited year of B.C.M.E.A. service for each year in which the Temporary Employee completed a minimum of eight hundred (800) hours of B.C.M.E.A. service, or eight hundred (800) hours of Westshore service, or a combination of eight hundred (800) hours of BC.M.E.A. and Westshore service.
- (v) It is understood and agreed that application of the above provisions shall not be duplicated or compounded for the same year of service.

#### (b) **Annual Vacation Entitlement and Pay**

- A Temporary Employee, upon permanent hire, who has a minimum of one (1) year of Westshore service, as referred to in (a)(ii) or (iii) of this section, shall be eligible for the vacation entitlement referred to in Article 9, section 9.01 (a) of this Agreement.
- (ii) If at the time of permanent hire a Temporary Employee does not have a credited year of Westshore service, as referred to in (b) (i) above, such Employee shall be given credit for up to twelve (12) months of continuous Westshore service immediately prior to the date of permanent hire where there was not a break in such service of twenty-one (21) days or more.

#### (c) Salary Continuance

For the purposes of Article 11, subsection 11.03 (a) and (b), I.L.W.U. continuous service shall be considered as a year in which the Regular or Temporary Employee has a minimum of eight hundred (800) hours of B.C.M.E.A. or Westshore service, or a combination of service of eight hundred (800) hours or more.

#### (d) Severance Allowance

- A Temporary Employee, upon permanent hire, shall be given credit for all credited Westshore years of service, as referred to in (a) (ii) or (iii) of this section, for the purpose of determining entitlement under Article 15, section 15.07.
- (ii) If at the time of permanent hire a Temporary Employee does not have a credited year of Westshore service, as referred to in (i) above, such employee shall be given up to a maximum of six (6) months of credited Westshore service where there is not a break in such service of twenty-one (21) days or more.

#### (e) **Seniority**

At the time of permanent employment, a Union member shall be senior to a Temporary Employee whose employment commenced on the same date.

#### **15.05 Temporary Layoffs**

- (a) The Company agrees there shall be no temporary layoffs (twelve (12) weeks or less)during the term of this Agreement, unless any of the following conditions occur:
  - Major Production Equipment breakdown, destruction of equipment or an Act of God affecting port facilities that are vital to the continuation of the Port operations.

- (ii) Cessation or interruption of rail traffic, ocean navigation or extended mine closures that are vital to the continuation of the Port operations.
- (iii) Any act by a government body or agency which restricts or limits the operation of the Port facilities.
- (iv) Where the Port discontinues ship loading and train dumping operations for any of the reasons contained in this provision.
- (b) It is further understood and agreed that no Employee shall be laid off on a temporary basis more than once per calendar year, unless further layoffs are required as a result of an Act of God, as referred to in subsection (a) (i) above.
- (c) In the event a temporary layoff exceeds twelve (12) weeks, the provisions of subsection 15.07 (a) shall apply, and the Employee shall be paid four (4) additional weeks severance pay, in lieu of the notice referred to in section 15.03 of this Article. Such severance pay shall be paid within seven (7) calendar days of the layoff becoming permanent.

## 15.06 Permanent Layoffs - Early Retirement Program

In the event of a permanent layoff the Company and the Union shall meet, prior to implementation of the layoff, to discuss and attempt to mutually agree on other economic alternatives, and/or an early retirement program, which shall not be less than that contained in the current Pension Plan, except that such program shall not include any form of actuarial reduction with respect to Pension Benefits.

Should the parties agree on an early retirement program, and bridging is a component of the program, calendar years bridged from age 55 to age 61 will be considered time worked for the purpose of calculating severance under 19.02. Further should the parties agree on an early retirement program, the eligible senior employee, or employees as the case may be, shall be given the option to accept the program prior to the 42 day notice being given. However,

notwithstanding the foregoing, if the parties cannot agree on an early retirement package within 7 calendar days, the Company may issue notice of intent to layoff, and discussions on the early retirement package may continue.

#### **15.07 Permanent Layoffs - Severance Allowance**

- (a) In the event that any regular Employee with six (6) months or more continuous Westshore Terminals service is laid off for a period in excess of twelve (12) consecutive weeks, such Employee shall be entitled to a severance allowance of five (5) weeks pay. Should a laid off Employee have in excess of one year's continuous Westshore Terminals service, he shall be entitled to an additional allowance of two (2) weeks pay for each full year of continuous Westshore Terminals service.
- (b) Payment of this severance allowance shall constitute final separation of employment with the Company. If such laid off Employee should be rehired as a regular permanent Employee his Westshore Terminals service for the purposes of this provision shall commence as of the date of his permanent rehire.

# **15.08 Continuation of Benefits - Regular Employees**

Where a regular Employee is laid off in accordance with Section 15.05 or 15.07 of this Article, such Employee shall continue to be covered by and eligible for only those benefits contained in Article 10 of this agreement, (excluding the benefits referred to in subsection 10.01 (d)), for a maximum period of six months, or until such time as the Employee becomes eligible for benefit coverage with any other employer, whichever occurs first.

# **ARTICLE 16 SAFETY**

#### 16.01 Safety Equipment, Appliances and Conditions

The Company agrees to provide such reasonable safe equipment, sanitation, appliances and places of work as shall be consistent with the type of work covered by this Agreement and as are ordinarily furnished for similar work. The Union agrees that Employees shall cooperate with the Company in the use of safety equipment.

#### **16.02 Safety Committee**

- (a) The Union and the Company agrees to appoint a Safety Committee comprised of three (3) members appointed by the Company and three (3) members appointed by the Union. The Union Committee representatives will consist of at least one (1) representative from maintenance. Alternate Union Committee representatives will be appointed to act in the absence of regular Committee representatives.
- (b) The Safety Committee will promote the desire of all Employees to work safely and make suggestions and recommendations with regard to safety rules and regulations. The Safety Committee shall ensure that all such safety rules and regulations are observed by all Employees. Employees that do not comply or abide by safety rules and regulations may be subject to disciplinary action.
- (c) The Safety Committee representatives shall meet once each month based on a predetermined schedule as mutually agreed to by the Committee. Minutes of the meetings will be distributed to all members of the Committee with copies to the Union and Management. It is understood that a First Aid man will be the recording secretary to the Safety Committee meetings.
- (d) Safety Committee representatives will receive no additional remuneration for time spent as a result of membership on the committee.
- (e) The Company shall convene monthly Crew Safety meetings which shall be held on the basis of a predetermined schedule as mutually agreed to by the Company and the Union.

## **16.03 Review of Safety Disputes**

- (a) It is the obligation of the parties to objectively review safety dispute items for immediate disposition and resolution.
- (b) Unless an Employee in good faith believes that to perform work under particular circumstances would endanger health and/or safety, he shall not refuse to work. Where good faith is shown on such a refusal the Employee may be reassigned to other work until the issue is resolved by the Company and the Union through the Grievance Procedure or by correction of item.
- (c) Employees misusing this intent and right are subject to disciplinary action.

# **ARTICLE 17 TRUST FUND**

#### 17.01 Trust Fund

(a) Effective February 1, 2013, the contribution will increase to thirty (30) cents per hour worked. The Company contributions shall continue monthly provided the fund does not exceed thirty thousand dollars (\$30,000) (excluding interest) at any one time. Effective on the date of ratification of this contract the maximum amount, excluding interest, will increase to fifty thousand (50,000) dollars. Effective February 1, 2013 the maximum amount, excluding interest, will increase to sixty thousand (60,000) dollars.

The Company agrees to contribute monthly an amount equal to thirty (30) cents per hour for all hours worked for all Employees to establish a benefit trust fund. Effective February 1, 2020 the Company contributions shall continue monthly provided the fund does not exceed <u>\$80,000</u> (excluding interest) at any one time.

- (b) The fund shall be administered by two (2) Employees appointed by the Company and two (2) members of the Union bargaining committee appointed by the Union (one of whom may be the Local 502 Union President provided the other is a regular Employee).
- (c) The proceeds of the fund shall be used to reimburse permanent Employees for extraordinary expenses not covered by the provisions of Article 10. The administrators of the fund shall meet as required, but not more than once per month, to consider any request for payment from the fund. Any payment from the fund must have the majority approval of the administrators.

#### **17.02 Eyeglass Benefits**

(a) In addition to the above contributions, the Company will contribute annually, on January 1st of each calendar year, fifteen thousand dollars (\$15,000) in place of the present insured eyeglass coverage, which pays the cost of frames, corrective lenses and or contact lenses.

- (b) The Company will review with the Trustees of the Trust Fund, every six (6) months, payments made for the above benefit and where such payments, on an annualized basis, exceed the Company's annual contribution, the Company shall contribute to the Trust Fund, immediately subsequent to such review, the necessary amount to meet the projected total annual cost to provide such benefits, plus an additional two thousand dollars (\$2,000.00).
- (c) On January 1st of each calendar year, the payments made for the previous calendar year shall be reviewed by the Company and the Trustees, and any necessary adjustment to increase or decrease the annual contributions shall be made at that time.
- (d) It is understood and agreed that the above contributions may only be used by the Trustees to provide benefits as referred to in this section.

# **ARTICLE 18 TRAINING**

# **18.01 Training Programs**

The Company and the Union shall meet during the term of this Agreement to establish the principles and guidelines of training programs, including apprenticeship programs, to upgrade the skills of operations and maintenance personnel. The implementation of such programs shall be based on mutual agreement of the Company and the Union.

# 18.02 Wages, Registrations Fees and Expenses

The Company shall pay an Employee for regular lost time, registration fees and other reasonable expenses to attend courses required by the Company. The Company and the Union will mutually agree to such courses.

# **18.03 New or Changed Equipment**

Upon the introduction of a new piece of equipment, or a change to an existing piece of equipment that impacts the way it is operated, employees will be give suitable instruction to familiarize them with the safe operation prior to them operating the equipment.

# **ARTICLE 19 PENSIONS**

#### 19.01 Pension Plan

The following shall form a part of the Pension Plan arrangement applicable to Employees covered by this agreement. The complete details governing the Pension Plan arrangement are contained in a separate document.

#### (a) **Amount of Pension**

Article F, Section F-2 of the Pension Plan shall be amended as follows:

(i) The monthly amount of normal pension payable to a member who retires on his normal retirement date shall be equal to an amount multiplied by years of credited Westshore service, not to exceed thirty five (35) years of such credited service, in accordance with the following:

| Effective February 1, 2008 | \$95.00         |
|----------------------------|-----------------|
| Effective February 1, 2013 | \$101.50        |
| Effective February 1, 2015 | \$105.00        |
| Effective February 1, 2017 | \$106.25*       |
| Effective February 1, 2018 | \$107.75*       |
| Effective February 1, 2019 | \$111.50**      |
| Effective February 1, 2020 | <u>\$113.68</u> |
| Effective February 1, 2021 | <u>\$118.00</u> |

\* This rate applies to active employees for Westshore service.

\*\*This rate increase to \$ 111.50 applies to active and retired employees for Westshore service. The following increases apply to active and retired members, as at the date of each increase:

Effective February 1, 2020 – pension increase for Westshore service equal to the year over year CPI for Vancouver for the previous calendar year.

Effective February 1, 2022 – pension increase for Westshore service equal to the year over year CPI for Vancouver for the previous calendar year.

(i) The amount of pension benefit applicable to B.C.M.E.A. service shall be based on the B.C.M.E.A. pension rate at the member's retirement date in accordance with the credited service provision (Article C) of the plan. This clause continues as currently practiced for regular employees, including those that commuted their BCMEA pension, but will not apply to any employees hired after the ratification date of the Collective Agreement.

#### (b) Bridging Supplement

Article F, Section F - 7 of the Pension Plan shall be amended as follows:

- (i) Effective February 1, 1988 the Bridging Supplement for a member with twenty-five (25) years or more of Westshore service shall be equivalent to the Old Age Security and Canada Pension Plan Benefits. A member with less than twenty-five (25) years of Westshore service shall receive a prorated benefit based on the equivalent Old Age Security and Canada Pension Plan Benefits, divided by twenty-five, and multiplied by the number of years of Westshore service.
- (ii) The adjustments in the Old Age Security and Canada Pension Plan benefits, as such adjustments occur, shall apply to members who retire on or after their <u>60<sup>th</sup></u> birthday.
- (iii) The Bridging Supplement for such credited B.C.M.E.A. service shall be the B.C.M.E.A. rate in effect at the time of the Member's retirement date based on the years of such service to bring the total years of Credited Service to a

maximum of twenty-five (25) years, less any benefit received from the B.C.M.E.A. Pension Plan.

#### (c) **Credited Service**

Article C, of the Pension Plan shall be amended to include the following: Section C-5 Credit for British Columbia Maritime Employers Association Service.

- (i) All permanent Employees who retire at Westshore with less than thirty-five (35) years of continuous service with the Company shall be entitled to credit for as many years of B.C.M.E.A. service as necessary to bring the total years of credited service to a maximum of thirty-five (35) years.
- (ii) All permanent Employees shall have their Company pension benefits reduced by the pension payable from the BCMEA Pension Plan calculated as if they retire from the BCMEA Pension Plan on the same date that they retire from the Company Pension Plan.
- (iii) Notwithstanding section 19.01 (c)(ii), all permanent Employees of the Company at October 18, 2003 who have commenced payment of their benefit from the BCMEA Pension Plan prior to October 18, 2003, shall have their Company pension benefits reduced by the pension they are receiving from the BCMEA Pension Plan on the date they retire from Westshore.
- (iv) Notwithstanding section 19.01 (c)(ii), all permanent Employees of the Company at October 18, 2003 who have elected to receive the lump sum value of their benefit from the BCMEA Pension Plan prior to October 18, 2003 and are identified in the Memorandum of Agreement dated October

8, 2003, shall have their amount of pension and bridging supplement reduced by \$300 per month.

- (v) Effective February 1, 2003, all Employees who retire from the Company, may elect at the time of retirement, in writing, to opt out of the above provisions and not receive credit for BCMEA service when determining either the amount of pension in accordance with 19.01 (a)(ii) or the bridging supplement in accordance with 19.01 (b)(ii) or both. In no event shall the benefit payable from the Company Pension Plan in respect of BCMEA service be less than zero.
- (d) Effective February 1, 2003, in the event an Employee's service with the Company terminates, and the person is subsequently rehired by the Company, his credited service shall only include his service with the Company and BCMEA service earned after his most recent termination date with the Company.

Any pension payable from the Company Pension Plan in respect of previous periods of employment shall equal the pension determined at the Employee's previous termination date(s), based on the plan provisions and benefit rates in effect at that time.

#### **19.02 Severance Benefit**

(a) All permanent Employees who retire at Westshore at age (60) who have a minimum of five (5) years continuous service with the Company immediately prior to retirement, and credited service according to subsection 19.01 (c) above of fifteen (15) years with the Company, shall be entitled to a lump sum severance benefit equivalent to the current B.C.M.E.A. rate for each year of credited service to a maximum of twenty-five (25 years), except that in the event of early retirement, the amount of severance benefit attributable to B.C.M.E.A. service shall be reduced by an amount equal to one-half of one percent (.5%) for each month that the Member's retirement precedes the last day of the month in which his <u>60<sup>th</sup></u> birthday occurs.

- (b) All permanent Employees who retire at Westshore who have a minimum of five (5) years continuous service with the Company immediately prior to retirement shall be entitled to a lump sum severance benefit of two thousand and ninety dollars (\$2,090) effective February 1, 2003, for each year of credited Westshore service to a maximum of thirty-five (35) years.
- (c) The amount of the lump sum severance benefit will be increased during the term of this Agreement as follows:

| Effective February 1, 2010 | \$2,600.00        |
|----------------------------|-------------------|
| Effective February 1, 2012 | \$2,625.00        |
| Effective February 1, 2013 | \$2,650.00        |
| Effective February 1, 2014 | \$2,675.00        |
| Effective February 1, 2015 | \$2,700.00        |
| Effective February 1, 2017 | \$2,800.00        |
| Effective February 1, 2018 | \$2,900.00        |
| Effective February 1, 2019 | \$3,000.00        |
| Effective February 1, 2021 | <u>\$3,300.00</u> |

Effective the date of ratification, a severance account will be established, into which, for each year, (pro-rated for a partial year), of a regular employee's service in which he is a member of the defined contribution pension plan, \$1600 will be deposited. This severance will be subject to the employee retiring at Westshore and is subject to the provisions of 19.02(e).

- (d) It is understood and agreed that an Employee shall not be entitled to a benefit, in accordance with (a) above, for any year of B.C.M.E.A. service for which the Employee has received such a benefit under the B.C.M.E.A. collective agreement.
- (e) Notwithstanding any provisions contained in subsections (a), (b) and (c) above, Employees who retire or terminate their employment with the Company for any

reason whatsoever prior to their fifty-fifth (55th) birthday shall not be entitled to any severance benefits.

#### **19.03 Pensioners Medical Benefits**

The Company shall provide the following Medical Benefits <u>for those regular employees who retire</u> <u>at Westshore who have a minimum of five years of continuous service with the Company</u> <u>immediately prior to retirement</u>:

- (a) Basic Medical Coverage under Medical Services Plan of British Columbia as provided for regular Employees.
- (b) Extended Health Care as provided for regular Employees.
- (c) Details of the benefits provided by the above coverage are contained in brochures provided by the insurance carrier. Retirees in receipt of insurance benefits from other sources may "top-up" those benefits with the coverage available under this Article, subject to the other benefits provider being first payer on a claim.

## **19.04 Funeral Allowance**

The Company shall provide a Funeral Allowance of <u>\$2,500.00</u> dollars, payable to the designated beneficiary, to those pensioners who are not eligible for the benefit under the B.C.M.E.A. Welfare and Pension programs.

## **19.05 Credited Pension Service**

On or before January 31st of each calendar year, the Company and the Union shall meet to update the seniority list indicating Westshore service and B.C.M.E.A. credited service for pension purposes.

## **19.06 Vesting Provision**

Vesting provision to be changed to thirty-five (35) years of age and eight (8) years of service effective February 1, 1984.

## **19.07 Pension Plan Conditions**

The following conditions shall apply to the Pension Plan but may not necessarily be contained in the official plan document:

- (a) In future the Company will discuss with the Union any changes in the level of benefits to the B.C.M.E.A. Industry Pension Plan to determine how these changes may affect Westshore Employees.
- (b) This will also confirm that should the pension benefit rates for active members of the Westshore Pension Plan increase during the life of this Agreement, then such increase will also apply to each retired member as of the effective date of such increase. Such increase will only apply to those retired members who had retired under the normal or the normal early retirement provisions of the Plan.
- (c) For the purposes of calculating pension benefit and severance benefits only, Westshore's credited service, effective February 1, 1995, shall be based on the following:
  - (i) in the year of hire a regular Employees service shall commence January 1st, and
  - (ii) completed years of service following the year of hire, and
  - (iii) in the year of retirement or termination, service shall be based on completed months of service in such year. For example: service of more than one-half (1/2) month shall constitute one (1) month or one-twelfth (1/12) of one (1) year's service.
  - (iv) broken service as referred to in the Pension Plan, shall also be calculated on the basis of years and months of broken service.

- (d) It is understood and agreed that application of subsection (c) (i) above will not result in any duplication of benefits or service with respect to the calculation of B.C.M.E.A. credited service.
- (e) Effective February 1, 1987, the Pension Plan will be amended to provide that a permanent Employee, who has a minimum of twenty-five (25) years of credited service and has reached the age of (<u>60</u>) may retire and will receive a pension without a reduction in pension benefits as a result of exercising such option.
- (f) Effective February 1, 2021 the Pension benefit payable to retired Westshore Employees for credited B.C.M.E.A. service shall be increased to \$65.00 per month for each year of such credited service.

## 19.08 Death-In-Service Survivor's Benefits

Death-In-Service survivor's benefits will be sixty percent (60%) of the value of the pension at the date of death, and will be calculated in accordance with the appropriate pension legislation.

## **19.09 Final Week Before Retirement**

Employees who retire in accordance with the terms of the Pension Plan will receive the greater of either 36 hours pay or actual hours scheduled in their final week before retirement, subject to them being entitled to be paid for that week.

## **19.10 Defined Contribution Plan**

Effective the date of ratification, all new regular employees will become members of the defined contribution pension plan. Effective January 1, 2017 all regular employees will become members of the defined contribution plan. Those regular employees hired prior to the ratification date, may opt out of the defined contribution plan and return to the defined benefit plan by expressing their desire to do so by providing the Director of Human Resources and the President of Local 502 of their intention to do so in writing prior to July 1, 2017, or other such date as may be mutually agreed upon. Should an employee opt out, he will forfeit any DC contributions. The Company contribution to the plan will be equivalent to 15% of the base pay as per Article 25.01 (a). The

initial fund manager will be Sunlife Financial. Notwithstanding any other provisions of the Collective Agreement, there will be no relationship between the BCMEA pension plan and the pension of any employees hired after the ratification date of the agreement.

## **19.11 Pension Service Lists**

The Parties shall cooperatively review and update the list every year. In the event that any changes are made to the list it will be dated and posted for a period of ninety (90) days, in an effort to ensure the accuracy of the list by allowing Employees an opportunity to confirm or question the stated service credits. The updated list will then be the official document confirming Westshore Terminals service and B.C.M.E.A. credited service for pension purposes.

Any questions arising concerning the accuracy of the service credits detailed in the list shall be referred to the Director of Human Resources and the President of Local 502.

# **ARTICLE 20 VACANCIES**

## 20.01 Temporary Vacancies

Persons required to fill vacancies temporarily will be called in the following order:

- (a) Shift Vacancy (three (3) days or less)
  - (i) A qualified Employee who is on his scheduled day off.
  - (ii) A temporary relief Employee who will be called out from the Union Hall.
  - (iii) A qualified Employee from the previous shift who will have his shift extended four (4) hours and an Employee that is scheduled on the next shift who will be called out four (4) consecutive hours early.
- (b) Shift Vacancy (more than three (3) consecutive days)
  - (i) A temporary relief Employee who will be called out from the Union Hall.
  - (ii) A qualified Employee who is on his scheduled day off.
  - (iii) A qualified Employee from the previous shift who will have his shift extended four (4) hours and an Employee that is scheduled on the next shift who will be called out four (4) consecutive hours early.
- (c) Shift Maintenance Vacancy (three (3) days or less)
  - (i) A qualified Employee who is on his scheduled day off.
  - (ii) For Day and Afternoon Shifts, a qualified Employee shall be transferred from the corresponding crew.

- (iii) For Night Shift only, a qualified Employee scheduled on the corresponding day shift crew shall be short shifted and if there is no qualified Employee on the corresponding day shift crew then a qualified Employee on the other day shift crew will be short shifted.
- (iv) A qualified daily dispatch Employee who will be called out from the Union Hall.
- (v) A qualified Employee from the previous shift who will have his shift extended four (4) hours and an Employee that is scheduled on the next shift who will be called out four (4) hours early.
- (d) Shift Maintenance Vacancy (more than three (3) consecutive days)
  - A qualified Employee shall be transferred from the corresponding crew and the resultant vacancy shall be filled by a qualified individual who will be called out from the Union Hiring Hall.
  - (ii) A qualified individual who will be called out from the Union Hiring Hall.
  - (iii) A qualified Employee who is on his scheduled day off
- (e) Notwithstanding (a), (b), (c) and (d) above, the Company shall have the option to call a temporary relief Employee from the Union Hall for vacation relief and long term absences.

# 20.02 Shift Replacement - Short Shifting

When the Company requires that an Employee change shifts during his work schedule to work as a shift replacement, the Employee will be allowed 12 hours off work with no loss in pay prior to returning to his next scheduled shift.

Example:

| Employee "A" | schedule 8 - 4  |  |
|--------------|-----------------|--|
| Employee "B" | schedule 12 - 8 |  |

If Employee "B" advises the Company of his intended absence prior to 12 noon on the previous day and Employee "A" is required to replace Employee "B" by changing shifts from 8-4 to 12-8 shift the following day, then in this event Employee "A" will be permitted to go home at 12 noon with no loss in pay to provide for the twelve (12) hours off before reporting to work at 12 midnight. If Employee "B" does not notify the Company of his intended absence by 12 noon then the twelve (12) hour rule will be reduced by the amount of time from 12 noon to the time that the Employee "B" notifies the Company.

There will be no change to Employee "A's" regular monthly salary as a result of this change in shifts. Employee "A" may be required to return to his original shift schedule at any time, however, the twelve (12) hour rule will apply.

## 20.03 Replacements - Spareboard List

- (a) In the event the Company requests an Employee from the Union Hall to fill a vacancy temporarily, such Employee will be from a spareboard list which is satisfactory to the Union and the Company. The Company and Union will meet to review this list on a monthly basis.
- (b) Daily dispatch Employees hired from the I.L.W.U. dispatch centre will be paid a sum of two (2) hours at the straight time rate as travel time for each occasion on which they are required to report to Westshore. Such payment will not be considered for the purpose of administering any of the provisions of this Collective Agreement. Temporary relief Employees will not be eligible for travel time allowances.

(c) The provisions of Article 15, subsection 15.03 (a) shall not apply to temporary relief Employees or daily dispatch Employees.

#### 20.04 Permanent Vacancies

- (a) When the Company requires new regular Employees for any reason, such vacancies will be posted in the Union Hall. The Company may, during the Employee's probationary period, terminate the Employee, who has filled the vacancy for at least seven (7) days, by giving him seven (7) days notice of termination. If the Employee has filled the vacancy for less than seven (7) days, no notice is required.
- (b) If qualified Employees are not available through the Union Hall and should the Company find a qualified applicant, then such applicant shall be hired through the Union Hall.
- (c) The Company agrees when filling a temporary trades classification for vacation relief that a qualified trades Employee sent from the Union Hall shall be given a minimum trial period of six (6) working days.

# **ARTICLE 21 FIRST AID**

## 21.01 First Aid Man

The Company agrees to employ a First Aid Man on all shifts. The First Aid Man shall perform other duties as assigned by the Company in accordance with present practice.

# 21.02 First Aid Courses and Certificates

- (a) Regular First Aid Men shall be paid for regular time lost to attend company approved first aid courses to maintain existing certifications. The courses are to be scheduled on consecutive days and the Company for this period will call a qualified replacement from the Union Hall.
- (b) The regular First Aid Man attending the approved course will be deemed to be on day shift for the period of the course with Saturday and Sunday as scheduled days off. There will be no offset or compensation for any additional time off he may have received had he followed his normal shift schedule. If the regular Employee does not successfully complete the required course a repeat session for the purposes of rewriting the exam will be granted without special shift arrangements and without pay. A qualified replacement in such cases will be dispatched from the Union Hall.
- (c) The courses are to be scheduled on consecutive days, however, the Company shall consider other options on request of the Employee, but courses must be approved by the Company prior to the commencement of same.

# **ARTICLE 22 EMPLOYEE DEFINITION**

#### 22.01 Employee Categories

For the purposes of this Collective Agreement, there shall be three types of Employees.

- (a) Daily Dispatch Employees: An Employee who is employed for a definite period of time, specified upon hiring by the Company, to be a period of seven (7) days or less.
- (b) Temporary Relief Employee: An Employee who is employed for a definite period of time, specified upon hiring by the Company, to be a period in excess of seven (7) days.
- (c) Regular Employee: All Employees other than daily dispatch Employees and temporary relief Employees who have successfully completed their probationary period and have met the requirements of Article 13. The probationary periods for regular Employees are as follows:
  - (i) If the Employee performs Maintenance work: ninety (90) calendar days from his date of hire;
  - (ii) If the Employee performs Production work: sixty (60) calendar days from his date of hire.

# **ARTICLE 23 PROTECTIVE CLOTHING**

#### 23.01 Clothing and Equipment

The Company shall, where deemed necessary as per past practice, provide Employees with protective clothing or equipment required in the performance of their duties. Such protective clothing or equipment shall be provided at no cost to the Employee and shall remain the property of the Company.

#### 23.02 Safety Work Boots

(a) All regular Employees shall be provided with an annual allowance of \$300 to purchase or repair C.S.A. approved safety work boots. Such allowance will be paid on or before February 15 of each year.

Temporary Employees who have worked 1,200 hours in consecutive years, (commencing in the 2006 calendar year), will be entitled to an annual allowance of \$200 towards the purchase or repair of C.S.A. approved safety work boots. This payment will be made on or before February 15th in the year following the achievement of consecutive 1,200 or more hours worked years and following each subsequent year the Employee works 1,200 or more hours.

All Employees are responsible to provide their own winter outerwear and it is understood that the outerwear must have visible reflective markings or the employee must wear a Company supplied reflective vest over the outerwear.

It is agreed that employees must wear such safety work boots on the work site and must wear hard hats in areas designated by the Safety Committee.

(b) In addition to the foregoing, upon provision of a receipt confirming the purchase of either or both of C.S.A. approved safety work boots or a winter coat for work purposes, an eligible employee may be entitled to up to an additional \$100 annually, subject to the amount being in excess of the Employee's basic entitlement as outlined above.

#### 23.03 Raingear

The Company shall provide all regular Employees with raingear and such raingear shall be replaced on an as needed basis on return of the damaged or worn out article. The Company shall also provide temporary Employees with raingear on an as needed basis, as per present practice, and such raingear shall be returned at the end of the temporary Employees dispatch. It is understood and agreed that all raingear provided by the Company remains the property of the Company.

# **ARTICLE 24 MANNING**

The following provisions shall apply to the number of Employees to be maintained on each shift when the port is operating:

#### 24.01 (a) Operations

- (i) The number of Employees on each production shift (schedule A-E) shall be maintained at not less than two (2) below the crew size as presently established.
- (ii) The number of Shift Maintenance Employees attached to each production shift (schedules A-E) shall be maintained at two (2) Employees per shift on all scheduled operating shifts.
- (iii) The minimum number of Employees on each production shift shall be governed by the following conditions:

| 0 Ships Loading | - | Where no ship loading is to occur during a shift the        |
|-----------------|---|---|
|                 |   | provisions of subsection 24.01 (a) (i) only shall apply.    |
| 1 Ship Loading  | - | Where only one (1) ship is being loaded at any one (1)      |
|                 |   | time during a shift the provisions of subsections 24.01 (a) |
|                 |   | (i) and 24.02 (c) shall apply.                              |
| 2 Ships Loading | - | Where two (2) ships are being loaded concurrently the       |
|                 |   | full operating crew shall be maintained.                    |

In addition to the above the Company agrees that where there are <u>seven</u> pieces of major equipment operating concurrently in a production mode during any shift the minimum manning requirements of this section shall be the established crew size. <u>Major equipment includes the following nine pieces of equipment: Shiploader 1, Shiploader 3, Shiploader 4, Stacker 44, Stacker 45, Stacker 46, Stacker 47, Dumper 31 and Dumper 32.</u>

## 24.01 (b) Maintenance

- (i) The numbers of Employees on each maintenance shift (schedules MA-ME) shall be maintained at not less than one
   (1) below the crew size as presently established.
- (ii) With respect to Warehouse and Tool Crib Employees, it is agreed that there shall be one (1) steady day shift warehouseman and five (5) tool crib/warehousemen who shall rotate on the same schedule as maintenance Employees.
- (iii) The Company and the Union agree that the maintenance shift schedules shall not result in any change to present or past practice as it applies to Shift Maintenance Employees attached to production crews.

## 24.01 (c) Day Shift Operations

 The number of day shift Employees to be maintained at all times, in accordance with their schedules, in the following categories shall be:

Day Operators, Cleanup Operators and Lubemen -

One (1) less than established crew size of each crew.

 Where replacements are required, for Day Operators, Cleanup Operators and Lubemen absent in excess of two (2) consecutive days, such replacements shall be in accordance with Article 20 - Vacancies.

#### 24.02 Replacements

- (a) It is agreed there shall be no requirement to replace any Employee on any crew who does not notify the Company of his intended absence a minimum of two (2) hours prior to the start of his scheduled shift.
- (b) When an Employee has requested to leave his job for any reason and the Employee is granted such leave with pay, there shall be no requirement on the part of the Company to replace that Employee on that shift. This provision shall not apply prior to the Employee completing the first three (3) hours of his shift.
- (c) When a replacement(s) is required to maintain the minimum manning levels, in accordance with section 24.01 above, or for an operations Employee absent in excess of two (2) consecutive days or for a maintenance Employee absent in excess of one (1) day, such replacements shall be made in accordance with Article 20 Vacancies. It is, however, agreed that Employees on shift will continue to work while awaiting the arrival of any replacement(s).
- (d) Where a shift maintenance Employee is absent for any reason, such Employee shall be replaced in accordance with Article 20 - Shift Maintenance Vacancy.
- (e) In the event the present crew sizes referred to in Sections 24.01 (a) and 24.01 (b) above are increased or decreased the Company and the Union shall meet to mutually agree on the minimum manning provisions.
- (f) Where a "regular Employee" is required to replace another "regular Employee", such "Employee" shall be replaced in accordance with the provisions of Article 20.
- (g) The only requirement to replace a regular electrician assigned to prints shall be for absences of more than two (2) weeks and all vacations. Effective February 1, 2013, Printmen will be subject to the same replacement rules as maintenance electricians, subject to Printmen being available on the crew or from the Hall.
- (h) The regular employee holding the position of automotive mechanic and any and all permanent replacements must hold a valid automotive mechanic trades certification.

The only requirement to replace the automotive mechanic shall be either for absences in excess of two weeks or for all vacations taken by the employee in the position.

If the Company determines that it will fill a vacancy of two weeks or less caused by the absence of the employee for any reason, or when the employee is on vacation, such replacement will come from the Union Hall.

Absences of two weeks or more will be filled as per the normal posting rules.

All replacements must hold a trades certificate as an automotive/mobile mechanic or as a Heavy Duty equipment mechanic.

A mutually agreed upon list of certified automotive and heavy-duty mechanics that will be eligible to replace the automotive mechanic will be developed.

In either event, should there not be a qualified replacement available from the Hall, the Company will not be obligated to fill the vacancy.

The position will initially be assigned to MDC.

#### 24.03 Vacation Relief Dispatch

- (a) The Union agrees that a vacation relief dispatch from the Union Hall may be utilized as a replacement for all of a regular Employee's scheduled consecutive weeks of vacation. In addition, the Union agrees that a vacation relief dispatch from the Union Hall may be utilized as a replacement for consecutive weeks of scheduled vacation for different Employees, to a maximum replacement period of two (2) weeks of scheduled vacation, provided the positions replaced are in the same category.
- (b) It is understood and agreed with respect to vacations that regular operations Employees and regular maintenance Employees will have the first option of replacing other regular operations and regular maintenance Employees, provided that such replacement does not reduce any crew below the minimums as set out in Article 9, subsection 9.06 (e).

(c) It is further agreed that regular Employees on approved vacation will be replaced at all times, including a regular Employee replacing another regular Employee who is on vacation.

#### 24.04 General

- (a) Where production work is being performed, a production Employee assigned to operations will not be required to perform cleanup work except under emergency conditions. In addition, excluding emergency conditions, a production Employee shall not be required to perform cleanup work during the first hour following any stoppage of production work. Where an Employee is required to perform cleanup work, he shall be provided a one-half (1/2) hour break prior to returning to production work, excluding the lunch break.
- (b) During inclement weather, Employees required to tie-up or let-go ships shall be allowed time to change clothes prior to returning to other production work. Such time shall be provided on a rotating basis in order not to interfere with production.
- Two (2) people will be used to berth or sail a barge which normally loads 2,000 to
   4,000 L/T.
- (d) Six (6) people will be used to berth ships and four (4) people will be used to sail ships, and three (3) people will be used to shift lines on the dock. If there is a requirement for more than the normal available crew due to the safety of the ship or people, because of lines or weather conditions, the shift maintenance people will be called on to assist in berthing or sailing of ships.
- (e) Employees in the maintenance department may use the forklift to assist in the performance of regular maintenance work on regularly scheduled maintenance shifts in and around the shop area.

Typical duties that may be performed are moving components from bay-to bay in the cat shop, between the cat-shop and the pressure washer, and between the catshop and the welding/machine/electrical shops.

Other duties must be agreed upon by the parties.

This article will not be in effect when the shift operator "down rules" are being used.

# 24.05 Absence Replacements

#### A. Regular Employees

- <u>1.</u> <u>The definition of a continuous absence will apply as follows:</u>
  - (a) All absences where Salary Continuance benefits are payable regardless of scheduled SDO's. Spareboard personnel ordered to replace Salary Continuance absences shall be ordered on the basis of known duration of the absence.
  - (b) All consecutive days of absence of any type following a week(s) of vacation.
  - (c) All consecutive days of scheduled absence following a full work week of non-vacation absence. Scheduled to be defined as absence requests submitted prior to the conclusion of the first full week(s) of absence.
- 2. Other applications of absence replacements:
  - (a) O/T DIL, Stat DIL, and/or Stat O/T DIL comprising a full work week block will be considered a continuous absence for replacement purposes.
  - (b) Days preceding a requested week(s) of vacation, including a week(s) of other absences, will be treated as separate absences for the purposes of application of Article 20 of the Collective Agreement.
  - (c) All replacements, where required by the Company, related to Employee training on-site will be dispatched from the Spareboard List through the Union Hall. Where such training is conducted off-site Employees shall be replaced from the Spareboard List through the Union Hall.
  - 3. The Company and the Union pledge to administer, the replacement requirements of the Collective Agreement and the Black Book Letters regarding the replacement of Employees consistent with the intent of such documents in a timely and accurate manner.

4. Salary Continuance Applications:

The Parties agree that abuse of the Salary Continuance application process (i.e. repeated failure to accurately notify the duration of an expected absence) will be referred to the Health Committee in accordance with the provisions of Article 11 of the Collective Agreement.

#### B. <u>Temporary Employees</u>

All temporary Employees, absent for any reason, shall be replaced from the Spareboard List through the Union Hall, except where they are temporarily filling a permanent job vacancy, in which case the absence will be treated the same as a regular Employee's absence.

#### 24.06 Manpower Shortages

Notwithstanding any other clause in the Agreement, in the event there are manpower shortages in operations, the Company may deploy manpower as outlined below in order to operate the site efficiently.

A shortage may exist if Employees are absent from work, and replacements are not available from the hall or from overtime, and the Company has determined there is a requirement to replace the absent employee(s). The normal procedure for sourcing replacement Employees will be followed (Replacements – Appendix B), and subject to operational requirements, transfers may be made from within the Operations Department, and if necessary, then from the Maintenance Department. In the case of the Graveyard shift, the Company will be required to attempt to pull off-shift operations Employees prior to a transfer. The Company may, at any point in the process prior to finding a replacement for the original vacancy, stop the process of procuring a replacement.

If Employees are transferred, the Company commits to attempt to replace them, sourcing the replacement in the same sequence as the original vacancy was sourced. If there are two or more vacancies, then the replacements will be sourced in accordance with each original vacancy, and if there is a requirement to source from both the hall and overtime, then the sourcing will alternate if necessary, subject to the sourcing of each original vacancy, with overtime taking the first turn.

The sequence to initiate an on-site pull will be as per Appendix C. The transfer will be done on a volunteer basis by seniority and qualifications. Should there not be any, or insufficient volunteers, then the work will be assigned to the junior qualified Employee(s) within the Operations Department. If the ability to effect such a transfer is exhausted and there remain vacancies, then there may be a forced transfer of the junior qualified Employee(s) in the Maintenance Department. However, the ability to source a replacement for one classification, but not another, may influence the selection of the classification to transfer to the vacancy.

In addition to the normal banking practices, overtime may not be banked when overtime is worked as a result of a transfer.

# **ARTICLE 25 SALARY**

# 25.01 Rates of Pay

(a) The following salary rates shall apply during the term of this Agreement:

| Effective February 1, 2012 | -        | \$ 8,582.00 per month         |
|----------------------------|----------|-------------------------------|
| Effective February 1, 2013 | -        | \$ 8,839.00 per month         |
| Effective February 1, 2014 | -        | \$ 9,104.00 per month         |
| Effective February 1, 2015 | -        | \$ 9,469.00 per month         |
| Effective February 1, 2016 | -        | \$ 9,658.00 per month         |
| Effective February 1, 2017 | -        | \$ 10,044.00 per month        |
| Effective February 1, 2018 | -        | \$ 10,345.00 per month        |
| Effective February 1, 2019 | -        | \$ 10,655.00 per month        |
| Effective February 1, 2020 | <u>-</u> | <u>\$ 10,921.00 per month</u> |
| Effective February 1, 2021 | Ξ        | <u>\$ 11,140.00 per month</u> |
|                            |          |                               |

- (b) Employees receive one rate for all work performed. All Employees will be required to be skilled in trade or craft and, depending on the conditions prevalent for work outside their specific trade or craft, assignments will be made within their ability to perform the work.
- (c) The above monthly salaries include payment in recognition of, and in lieu of, certain shift differentials or premium time paid by the British Columbia Waterfront Industry.

# 25.02 (a) Temporary Employees

A Temporary Employee shall be paid the equivalent hourly or weekly rate of pay based on the monthly rates contained in section 25.01 of this Article, in accordance with the Temporary Pay Matrix which is set out in a separate document.

# 25.02 (b) <u>New Hire Pensions</u>

Employees, including apprentices, who become regulars and had worked regularly at Westshore since 2006 will be entitled to the following payments for years in which they worked 1200 hours and did not receive a full pensionable year at the B.C.M.E.A. for any of these years will be entitled to the following:

- Employees, including apprentices, who met the threshold of hours worked during the period of 2006 to 2010 will entitled to \$4,600 for each such year in which they worked the qualifying hours.
- Employees, including apprentices, who met the threshold of hours worked from 2011 to 2016 will be entitled to a payment of \$1,600 for each such year in which they worked the qualifying hours.
- Partial B.C.M.E.A. years will result in a top up payment to the designated Westshore amount for each such qualifying year.

Eligible employees must furnish proof from the B.C.M.E.A. that no pension was earned for potential qualifying years. All payments will be made to each employee's Sunlife account, and the employee must furnish proof of RRSP room for the contribution. The payment may be deferred until there is the requisite RRSP room. Employees who do not meet these criteria may be reviewed on a case by case basis.

# **ARTICLE 26 MISCELLANEOUS**

## 26.01 Elevations

When the Company has determined a need to elevate a Regular Employee who is a member of the Union to the rank of Foreman, and have agreed with the terms of the elevation with Local 514, they will meet with Local 502 and agree on the details associated with the elevation. The Company will review its selection of Employees for such opportunities with the Union, but will retain the right to make the final selection, providing the process is reasonable and fair.

## 26.02 Discipline

Where the Company is contemplating issuing discipline to an employee, they will make every reasonable effort to inform the Union in a timely fashion.

# ARTICLE 27 - TERM

# 27.01 Term of Agreement

This Agreement shall remain in effect until midnight January 31, 2022 and thereafter from year to year, until either party shall give notice in writing to the other party of a desire to terminate, change or modify same. Such notice may be given at any time within the four (4) month period immediately preceding the expiry date.

Unless specifically addressed and modified in the attachment hereto and/or this Memorandum of Agreement, all Black Book letters attached to the Collective Agreement, and all other terms and conditions contained in the Collective Agreement between the parties effective February 1, 2016, are to form part of this revised agreement and remain in full force and effect.

IN WITNESS WHEREOF the Parties hereto have affixed their signatures hereto this <u>21st day of May</u>, <u>2021.</u> at Delta, B.C.

## WESTSHORE TERMINALS

## LIMITED PARTNERSHIP

Nick Guppy,

Manager, Operations

Peter Prince-Wright,

Manager, Maintenance-Electrical

WAREHOUSE UNION, LOCAL 502

INTERNATIONAL LONGSHORE AND

Rick Hurtubise,

President, Local 502

Randy Chartier, Labour Relations

Joost Van Woerden,

Manager, Maintenance Mechanical

David Honsberger,

Director, Human Resources

W. Bridgs,

Representative

man and

Leif Tuura,

Representative

Ol-

Rod Butler,

Representative

S. Buckle,

Representative

Jeff Sloggett,

Representative

# **APPENDIX "A" ASSIGNED WORK SCHEDULES**

## **Assigned Work Schedules**

- (a) Employees who work according to Schedule "I" and Schedule "II" have a normal work day of eight (8) hours and thirty-five (35) minutes, per twenty-four (24) hour period.
- (b) Employees who work according to Schedule "III" have a normal work day of eight(8) hours per twenty-four (24) hour period.
- (c) Employees on Schedule "I" and Schedule "II" shall not leave their operating equipment or actual work area until their replacements arrive at the operating equipment or actual work area and are able to commence work.

|     |         | SMTWTFS S             | MTWTFS      | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|-----|---------|-----------------------|-------------|------------------|------------------|------------------|------------------|
| OAO |         | 1 N X X X A A A 2 A   | A A X X X D | 3 D D D D X >    | 4 X X X N N N N  | 5 N N X X X A A  | 6 A A A A X X X  |
|     | A       | 7 D D D D D X 8 X     | XXXNNN      | 9 N N N X X A    | 10 A A A A A X X | 11 X D D D D D D | 12 X X X X X N N |
|     | A       | 13 N N N N X X X 14 A | AAAAX       | 15 X X D D D D D | 16 D X X X X X N | 17 N N N N N X X | 18 X A A A A A A |
|     |         | 19 X X X D D D D 20 D | DXXXXX      | 21 X N N N N N X | 22 X X X A A A A | 23 A X X X X D D | 24 D D D X X X X |
|     |         | 25 X X N N N N N      |             |                  |                  |                  |                  |
|     |         | SMTWTFS S             | MTWTFS      | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
| OBO |         | 1 A A A A X X X 2 D   | DDDDX       | 3 X X X X N N N  | 4 N N N X X X A  | 5                | 6 X D D D D D D  |
|     | В       | 7 X X X X X N N 8 N   | N N N X X X | 9 А А А А А А У  | 10 X X D D D D D | 11 D X X X X X N | 12 N N N N N X X |
|     |         | 13 X A A A A A A 14 X | XXDDDD      | 15 D D X X X X X | 16 X N N N N N X | 17 X X X A A A A | 18 A X X X X D D |
|     |         | 19 D D D X X X X 20 X | XNNNNN      | 21 N X X X A A A | 22 A A A X X X D | 23 D D D D X X   | 24 X X X N N N N |
|     |         | 25 N N X X A A        |             |                  |                  |                  |                  |
|     |         | SMTWTFS S             | MTWTFS      | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
| 000 |         | 1 X N N N N N X 2 X   | ХХАААА      | 3 A X X X X D D  | 4 D D D X X X X  | 5 X X N N N N N  | 6 N X X X A A A  |
|     | С       | 7 A A A X X X D 8 D   | DDDDXX      | 9 X X X N N N N  | 10 N N X X X A A | 11 A A A A X X X | 12 D D D D D X   |
|     | C       | 13 X X X X N N N 14 N | N N X X X A | 15 А А А А А Х У | 16 X D D D D D D | 17 X X X X X N N | 18 N N N N X X X |
|     |         | 19 A A A A A A X 20 X | XDDDDD      | 21 D X X X X X N | 22 N N N N N X X | 23 ХАААААА       | 24 X X X D D D D |
|     |         | 25 D D X X X X X      |             |                  |                  |                  |                  |
|     |         | SMTWTFS S             | MTWTFS      | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
| ODO |         | 1 X D D D D D 2 X     | XXXXNN      | 3 N N N N X X >  | 4 A A A A A A X  | 5 X X D D D D D  | 6 D X X X X X N  |
|     | D       | 7 N N N N N X X 8 X   | AAAAAA      | 9 X X X D D D D  | 10 D D X X X X X | 11 X N N N N N X | 12 X X X A A A A |
|     |         | 13 A X X X X D D 14 D | D D X X X X | 15 X X N N N N N | 16 N X X X A A A | 17 A A A X X X D | 18 D D D D X X   |
|     |         | 19 X X X N N N N 20 N | NXXXAA      | 21 A A A A X X > | 22 D D D D D X   | 23 X X X X N N N | 24 N N N X X X A |
|     |         | 25 A A A A A X X      |             |                  |                  |                  |                  |
|     |         |                       | MTWTFS      |                  |                  |                  | SMTWTFS          |
| OEO |         | 1 D X X X X X N 2 N   | NNNXX       | 3 X A A A A A A  | 4 X X X D D D D  | 5 D D X X X X X  | 6 X N N N N N X  |
|     | E       | 7 X X X A A A A 8 A   | XXXXDD      | 9 D D D X X X >  | 10 X X N N N N N | 11 N X X X A A A | 12 A A A X X X D |
|     | <b></b> | 13 D D D D D X X 14 X | XXNNNN      | 15 N N X X X A A | 16 A A A A X X X | 17 D D D D D X   | 18 X X X X N N N |
|     |         | 19 N N N X X X A 20 A | AAAAXX      | 21 X D D D D D D | 22 X X X X X N N | 23 N N N N X X X | 24 A A A A A A X |
|     |         | 25 X X D D D D D      |             |                  |                  |                  |                  |

# Schedule I – Eight (8) Hour and Thirty-Five Minute Shift

|       | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|-------|------------------|------------------|------------------|------------------|------------------|------------------|
|       | 1 D X X X A A A  | 2 A A A X X X D  | 3 D D D D X X    | 4 X X X D D D D  | 5 D D X X X A A  | 6 A A A A X X X  |
| N / A | 7                | 8 X X X X D D D  | 9 D D D X X X A  | 10 А А А А А Х Х | 11 X D D D D D D | 12 X X X X X D D |
| MA    | 13 D D D D X X X | 14 ААААААХ       | 15 X X D D D D D | 16 D X X X X X D | 17 D D D D X X   | 18 X A A A A A A |
|       | 19 X X X D D D D | 20 D D X X X X X | 21 X D D D D X   | 22 X X X A A A A | 23 A X X X X D D | 24 D D D X X X X |
|       | 25 X X D D D D D |                  |                  |                  |                  |                  |
|       | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|       | 1 A A A A X X X  | 2 D D D D D X    | 3 X X X X D D D  | 4 D D D X X X A  | 5                | 6 X D D D D D D  |
| MB    | 7 X X X X X D D  | 8 0 0 0 0 X X X  | 9                | 10 X X D D D D D | 11 D X X X X X D | 12 D D D D X X   |
|       | 13 X A A A A A A | 14 X X X D D D D | 15 D D X X X X X | 16 X D D D D X   | 17 X X X A A A A | 18 A X X X X D D |
|       | 19 D D D X X X X | 20 X X D D D D   | 21 D X X X A A A | 22 A A A X X X D | 23 D D D D D X X | 24 X X X D D D D |
|       | 25 D D X X X A A |                  |                  |                  |                  |                  |
|       | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|       | 1 X D D D D X    | 2 X X X A A A A  | 3 A X X X X D D  | 4 D D D X X X X  | 5 X X D D D D D  | 6 D X X X A A A  |
| MC    | 7 A A A X X X D  | 8 D D D D D X X  | 9 X X X D D D D  |                  | 11 A A A A X X X |                  |
|       | 13 X X X X D D D | 14 D D D X X X A | 15 A A A A A X X | 16 X D D D D D D | 17 X X X X X D D | 18 D D D D X X X |
|       | 19 ААААААХ       | 20 X X D D D D   | 21 D X X X X X D | 22 D D D D D X X | 23 X A A A A A A | 24 X X X D D D D |
|       | 25 D D X X X X X |                  |                  |                  |                  |                  |
|       | SMTWTFS          |                  |                  | SMTWTFS          |                  |                  |
|       | 1 X D D D D D D  |                  |                  |                  | 5 X X D D D D D  |                  |
| MD    | 7 D D D D X X    | -                |                  | -                | 11 X D D D D X   |                  |
|       | 13 A X X X X D D |                  |                  |                  | 17 A A A X X X D |                  |
|       | 19 X X X D D D D | 20 D D X X X A A | 21 A A A A X X X | 22 D D D D D D X | 23 X X X X D D D | 24 D D D X X X A |
|       | 25 A A A A A X X |                  |                  |                  |                  |                  |
|       | SMTWTFS          |                  |                  | S M T W T F S    |                  |                  |
|       |                  | 2 D D D D D X X  |                  |                  | 5 D D X X X X X  |                  |
| ME    | 7 X X X A A A A  | -                | -                |                  | 11 D X X X A A A |                  |
|       |                  |                  |                  |                  | 17 D D D D D X   |                  |
|       |                  |                  | 21 X D D D D D D | 22 X X X X X D D | 23 D D D D X X X | 24               |
|       | 25 X X D D D D D |                  |                  |                  |                  |                  |

Schedule II – Eight (8) Hour and Thirty-Five Minute Shift

Schedule III - Eight (8) Hour Shift



|    | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|----|------------------|------------------|------------------|------------------|------------------|------------------|
|    | 1 N X X X A A A  |                  |                  |                  | 5 N N X X X A A  |                  |
|    | 7 0 0 0 0 0 0 0  | 8 X X X X N N N  |                  |                  |                  | 12 X X X X X N N |
| MA | 13 N N N N X X X |                  |                  |                  | 17 N N N N N X X |                  |
|    | 19 ХХХОООО       | 20 0 0 X X X X X | 21 X N N N N N X | 22 X X X A A A A | 23 A X X X X D D | 24 D D D X X X X |
|    | 25 X X N N N N N |                  |                  |                  |                  |                  |
|    | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|    | 1 A A A A X X X  | 2 D D D D D X    | 3 X X X X N N N  | 4 N N N X X X A  | 5                | 6 X D D D D D D  |
|    | 7 X X X X X N N  | 8 N N N N X X X  | 9 А А А А А А Х  | 10 X X D D D D D | 11 D X X X X X N | 12 N N N N N X X |
| MB | 13 ХАААААА       |                  | 15 D D X X X X X | 16 X N N N N N X | 17 X X X A A A A | 18 A X X X X D D |
|    | 19 D D D X X X X | 20 X X N N N N N | 21 N X X X A A A | 22 A A A X X X D | 23 D D D D X X   | 24 X X X N N N N |
|    | 25 N N X X X A A |                  |                  |                  |                  |                  |
|    | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|    | 1 X N N N N N X  | 2 X X X A A A A  | 3 A X X X X D D  | 4 D D D X X X X  | 5 X X N N N N N  | 6 N X X X A A A  |
| МС | 7 A A A X X X D  | 8 0 0 0 0 0 X X  | 9 X X X N N N N  | 10 N N X X X A A | 11 A A A A X X X | 12 D D D D D X   |
|    | 13 X X X X N N N | 14 N N N X X A   | 15 A A A A A X X | 16 X D D D D D D | 17 X X X X X N N | 18 N N N N X X X |
|    | 19 ААААААХ       | 20 X X D D D D D | 21 D X X X X X N | 22 N N N N N X X | 23 ХАААААА       | 24 X X X D D D D |
|    | 25 D D X X X X X |                  |                  |                  |                  |                  |
|    | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          | SMTWTFS          |
|    | 1 X D D D D D D  | 2 X X X X X N N  | 3 N N N N X X X  | 4 A A A A A A X  | 5 X X D D D D D  | 6 D X X X X X N  |
| MD | 7                | 8 X A A A A A A  | 9 X X X D D D D  | 10 D D X X X X X | 11 X N N N N N X | 12 X X X A A A A |
|    | 13 A X X X X D D | 14 D D D X X X X | 15 X X N N N N N | 16 N X X X A A A | 17 A A A X X X D | 18 D D D D X X   |
|    | 19 X X X N N N N | 20 N N X X X A A | 21 A A A A X X X | 22 D D D D D X   | 23 X X X X N N N | 24 N N N X X X A |
|    | 25 A A A A A X X |                  |                  |                  |                  |                  |
|    | SMTWTFS          |                  |                  |                  |                  |                  |
|    | 1 D X X X X X N  |                  |                  |                  |                  | 6 X N N N N N X  |
| ME | 7 X X X A A A A  |                  |                  |                  | 11 N X X X A A A |                  |
|    | 13 D D D D X X   |                  | 15 N N X X X A A |                  |                  |                  |
|    |                  |                  | 21 X D D D D D D | 22 X X X X X N N | 23 N N N N X X X | 24 A A A A A A X |
|    | 25 X X D D D D D |                  |                  |                  |                  |                  |

Schedule IV – Eight (8) Hour and Thirty-Five Minute Shift

| Schedule V – Eight (8) Hour and Th | irty-Five Minute Shift |
|------------------------------------|------------------------|
|------------------------------------|------------------------|

|     |          | S | Μ | Т | W | Т        | F | S |    | S | Μ | Т | W | Т | F : | S |    | S | Μ | Т | W | Т | F | S |    | S | Μ | Т | W  | Т | F S        |    | S      | N   | 1 T        | W | Т | F | S |    | S | М | Т | W | Т | FS         |
|-----|----------|---|---|---|---|----------|---|---|----|---|---|---|---|---|-----|---|----|---|---|---|---|---|---|---|----|---|---|---|----|---|------------|----|--------|-----|------------|---|---|---|---|----|---|---|---|---|---|------------|
|     | 1        | D | Х | Х | Х | D        | D | D | 2  | D | D | D | Х | Х | ΧI  | D | 3  | D | D | D | D | D | Х | x | 4  | Х | Х | Х | D  | D | DD         | 5  | 5 C    | D   | x          | Х | Х | D | D | 6  | D | D | D | D | Х | хх         |
|     | 7        | D | D | D | D | D        | D | Х | 8  | Х | Х | Х | Х | D | DI  | D | 9  | D | D | D | Х | Х | X | D | 10 | D | D | D | D  | D | хх         | 11 | . х    | D   | D          | D | D | D | D | 12 | Х | Х | Х | Х | Х | DC         |
| MDA | 13       | D | D | D | D | Х        | Х | Х | 14 | D | D | D | D | D | D   | х | 15 | Х | Х | D | D | D | D | D | 16 | D | Х | Х | х  | Х | ХD         | 17 | ' C    | D   | D          | D | D | Х | х | 18 | х | D | D | D | D | DC         |
|     | 19       | Х | х | х | D | D        | D | D | 20 | D | D | х | х | х | X   | x | 21 | Х | D | D | D | D | D | x | 22 | Х | Х | х | D  | D | DD         | 23 | 5 C    | х   | x          | х | х | D | D | 24 | D | D | D | Х | х | хх         |
|     | 25       | Х | Х | D | D | D        | D | D |    |   |   |   |   |   |     |   |    |   |   |   |   |   |   |   |    |   |   |   |    |   |            |    |        |     |            |   |   |   |   |    |   |   |   |   |   |            |
|     |          | S | Μ | Т | W | Т        | F | S |    | S | Μ | Т | W | Т | F : | s |    | S | Μ | Т | W | Т | F | s |    | S | М | Т | W  | Т | F S        |    | S      | N   | 1 Т        | W | Т | F | S |    | S | Μ | Т | W | Т | F S        |
|     | 1        | D | D | D | D | Х        | Х | х | 2  | D | D | D | D | D | D   | х | 3  | Х | Х | Х | Х | D | D | D | 4  | D | D | D | Х  | Х | ХD         | 5  | 5 C    | ) D | D          | D | D | Х | Х | 6  | Х | D | D | D | D | DD         |
|     | 7        | х | х | х | х | х        | D | D | 8  | D | D | D | D | х | x   | x | 9  | D | D | D | D | D | D | x | 10 | х | х | D | D  | D | DD         | 11 | . C    | х   | x          | х | х | х | D | 12 | D | D | D | D | D | хх         |
| MDB | 13       | х | D | D | D | D        | D | D | 14 | х | х | х | D | D | DI  |   | 15 | D | D | х | х | х | X | x | 16 | х | D | D | D  | D | DХ         | 17 | , x    | x   | x          | D | D | D | D | 18 | D | х | х | х | х | DD         |
|     | 19       |   | D | D | x | x        | x | x | 20 | x | x | D | D | D | DI  |   | 21 | D | x | x | X |   | D |   | -  |   |   |   |    |   | X D        |    | с<br>Г | ) D |            | D | D | x | x | 24 | x | x | x | D | D |            |
|     | 25       | D | D | - |   |          |   |   | 20 | ~ | ~ | 0 | 0 | 2 |     |   |    | 5 | ~ | ~ | ~ | 0 |   |   |    | 0 | 5 | 5 | ~  | ~ | Λ Β        |    |        |     | 2          | 0 | 5 | ~ | ~ |    | ~ | ~ | ~ | 0 | 0 |            |
|     | 23       |   |   |   |   |          | F | - |    | S | М | т | w | т | F S | 5 |    | S | М | т | w | т | F | s |    | s | М | т | w/ | т | F S        |    | 5      | M   | 1 Т        | W | т | F | S |    | s | М | т | w | т | FS         |
|     | 1        | - |   |   |   |          | D | - | 2  |   |   |   |   |   | D 1 | _ | 3  | - |   |   |   |   | D | - |    | - |   |   |    |   | <u>x x</u> | -  | -      |     |            | D |   |   | - |    | - |   |   |   |   | . <u> </u> |
|     |          |   |   |   |   |          | x |   | 2  |   |   | D | _ | - | XX  |   | -  | x |   |   |   |   | D | - |    |   |   |   |    |   | DD         |    | -      |     |            | _ | _ | x | _ | 12 | D |   |   |   |   | рх         |
| MDC | ,<br>13  | - | - | - |   |          | D | - | 14 | D | Б | р | v | x |     |   | 15 | _ | _ | D | - | - | x | v |    | 0 | - |   |    |   |            |    |        | , D | v v        | v |   | D |   | 18 | D | - | D | - | - | хх         |
|     | 13<br>19 |   |   |   |   | -        | D | - | 20 | v | v |   |   |   |     | - | 21 | Р | v | v | - | - | X |   | -  |   |   |   |    |   | хх         |    |        |     | . ^<br>. D | D | D | - | _ | 24 | v | v | v | - |   |            |
|     |          |   |   | v | - | -        | - | ~ | 20 | ^ | ^ | U | U | υ | וט  |   | 21 | U | ^ | ^ | ^ | ^ | ~ |   | 22 | U | U | U | U  | U | ~ ^        | 23 |        |     | , D        | U | U | υ | U | 24 | ^ | ^ | ^ | U | U |            |
|     | 25       | υ | υ | ^ | ~ | <u> </u> | Ň | ^ |    |   |   |   |   |   |     |   |    |   |   |   |   |   |   |   |    |   |   |   |    |   |            |    |        |     |            |   |   |   |   |    |   |   |   |   |   |            |

# **APPENDIX "B" REPLACEMENTS**

| Absence  |   |                        | Method of Replacement   | Agreement Provisions   |
|--|---|------------------------|---|--|
| 1. Banked Overtime                               |   | Union Hall             | - Subject to down rules   | Article 7.07 – Union Hall and Article 24                             |
| 2. Book- off (No Pay)                            | - Regular Employees                     | Union Hall             | - Subject to down rules   | Article 7.08(b) and Article 24                                       |
| 3. Book- off (No Pay)                            | - Temporaries/Daily Dispatch            | Union Hall             | 1   | Article 7.08(b)  |
| 4. Call-Outs                                     | I                                       | Overtime               |   | Article 7.05 and BB Letter R1-81                                     |
|  |   |                        | -out, minimum pay of 9 hours at S.T.,<br>period and no requirement to work beyond   | Overtime call-outs   |
| 5. Leave of Absence                              | - Bereavement,<br>Jury Duty, & Personal | Overtime<br>Union Hall | <ul> <li>3 days or less- subject to down rules</li> <li>more than 3 days</li> </ul> | Article 20.01, Article 24 & BB Letter 3-94                           |
|  | - Union Business                        | Union Hall             | - if required by Company  | Article 12.02  |
| 6. Short Shifting                                |   | Overtime               |   | Article 20.02  |
| 7. Sickness/Injury/<br>Disability <sup>*1/</sup> | - Regular Employees                     | Overtime<br>Union Hall | <ul><li> 3 days or less- subject to down rules</li><li> more than 3 days</li></ul>  | Article 20.01, and Article24   |
| 8. Sickness/Injury/                              | - Temporary Emp.                        | Union Hall             | - no T & T  | Mandatory Replacement  |
| Disability                                       | - Daily Dispatch                        | Union Hall             | - with T & T  | Mandatory Replacement  |
| 9. Statutory Holiday *1                          | - D.I.L., Banked Day &                  | Overtime               | - 3 days or less- subject to down rules   | Article 20.01 and Article 24   |
|  | Birthday Stat                           | Union Hall             | - more than 3 days  |  |
| 10. Temporary Employee *2                        | - 6 mth. replacement                    | Overtime               | - 3 days or less- subject to down rules   | Article 20.01 and Article 24 and BB Letter 3-94 Absence Replacements |
| (in lieu of Regular)                             |   | Union Hall             | - more than 3 days  |  |
| 11. Training * <sup>3</sup>                      | 1                                       | Union Hall             | - If required by Company  | Article 20.01 and Article 24 and BB Letter 3-94 Absence Replacements |

Notes: \*1See also BB Letter 3-94 – Absence Replacements, for definition of continuous absences following a vacation period.

<sup>\*2</sup>Requirement to replace are subject to notice provisions – Article 24, section 24.02.

<sup>\*3</sup>All absence replacements from Union Hall will receive Time and Travel.

# **APPENDIX "C" TRANSFER SEQUENCE**

Offered in Seniority Order – Junior qualified may be assigned

## Step One

Transfer from the following positions, listed in no particular order:

- Clean up, indoor and outdoor
- Warehouse
- Control Room Operator
- Lube
- Shift Operator
- First Aid
- Cat Operator
- Gearman

## **Step Two**

Transfer from the following positions, listed in no particular order:

- Mechanics
- Welders
- Electricians, including prints
- Shift Mechanic
- Shift Electrician

# CALENDARS <u>2021 TO 2025</u>

# 2021

|    | January |    |    |    |    |    |  |  |  |  |  |  |  |  |  |
|----|---------|----|----|----|----|----|--|--|--|--|--|--|--|--|--|
| Su | Мо      | Tu | We | Th | Fr | Sa |  |  |  |  |  |  |  |  |  |
|    |         |    |    |    | 1  | 2  |  |  |  |  |  |  |  |  |  |
| 3  | 4       | 5  | 6  | 7  | 8  | 9  |  |  |  |  |  |  |  |  |  |
| 10 | 11      | 12 | 13 | 14 | 15 | 16 |  |  |  |  |  |  |  |  |  |
| 17 | 18      | 19 | 20 | 21 | 22 | 23 |  |  |  |  |  |  |  |  |  |
| 24 | 25      | 26 | 27 | 28 | 29 | 30 |  |  |  |  |  |  |  |  |  |
| 31 |         |    |    |    |    |    |  |  |  |  |  |  |  |  |  |

| February             |    |    |    |    |    |    |  |  |  |  |  |  |  |
|----------------------|----|----|----|----|----|----|--|--|--|--|--|--|--|
| Su Mo Tu We Th Fr Sa |    |    |    |    |    |    |  |  |  |  |  |  |  |
|                      | 1  | 2  | 3  | 4  | 5  | 6  |  |  |  |  |  |  |  |
| 7                    | 8  | 9  | 10 | 11 | 12 | 13 |  |  |  |  |  |  |  |
| 14                   | 15 | 16 | 17 | 18 | 19 | 20 |  |  |  |  |  |  |  |
| 21                   | 22 | 23 | 24 | 25 | 26 | 27 |  |  |  |  |  |  |  |
| 28                   |    |    |    |    |    |    |  |  |  |  |  |  |  |
|                      |    |    |    |    |    |    |  |  |  |  |  |  |  |

|    | March |    |    |    |    |    |  |  |  |  |  |  |  |  |  |
|----|-------|----|----|----|----|----|--|--|--|--|--|--|--|--|--|
| Su | Мо    | Tu | We | Th | Fr | Sa |  |  |  |  |  |  |  |  |  |
|    | 1     | 2  | 3  | 4  | 5  | 6  |  |  |  |  |  |  |  |  |  |
| 7  | 8     | 9  | 10 | 11 | 12 | 13 |  |  |  |  |  |  |  |  |  |
| 14 | 15    | 16 | 17 | 18 | 19 | 20 |  |  |  |  |  |  |  |  |  |
| 21 | 22    | 23 | 24 | 25 | 26 | 27 |  |  |  |  |  |  |  |  |  |
| 28 | 29    | 30 | 31 |    |    |    |  |  |  |  |  |  |  |  |  |
|    |       |    |    |    |    |    |  |  |  |  |  |  |  |  |  |

|    | April |    |    |    |    |    |  |  |  |  |  |  |  |  |  |
|----|-------|----|----|----|----|----|--|--|--|--|--|--|--|--|--|
| Su | Мо    | Tu | We | Th | Fr | Sa |  |  |  |  |  |  |  |  |  |
|    |       |    |    | 1  | 2  | 3  |  |  |  |  |  |  |  |  |  |
| 4  | 5     | 6  | 7  | 8  | 9  | 10 |  |  |  |  |  |  |  |  |  |
| 11 | 12    | 13 | 14 | 15 | 16 | 17 |  |  |  |  |  |  |  |  |  |
| 18 | 19    | 20 | 21 | 22 | 23 | 24 |  |  |  |  |  |  |  |  |  |
| 25 | 26    | 27 | 28 | 29 | 30 |    |  |  |  |  |  |  |  |  |  |
|    |       |    |    |    |    |    |  |  |  |  |  |  |  |  |  |

|    |    |    | May | ,  |    |    |
|----|----|----|-----|----|----|----|
| Su | Мо | Tu | We  | Th | Fr | Sa |
|    |    |    |     |    |    | 1  |
| 2  | 3  | 4  | 5   | 6  | 7  | 8  |
| 9  | 10 | 11 | 12  | 13 | 14 | 15 |
| 16 | 17 | 18 | 19  | 20 | 21 | 22 |
| 23 | 24 | 25 | 26  | 27 | 28 | 29 |
| 30 | 31 |    |     |    |    |    |

|    | June |    |    |    |    |    |  |  |  |
|----|------|----|----|----|----|----|--|--|--|
| Su | Мо   | Tu | We | Th | Fr | Sa |  |  |  |
|    |      | 1  | 2  | 3  | 4  | 5  |  |  |  |
| 6  | 7    | 8  | 9  | 10 | 11 | 12 |  |  |  |
| 13 | 14   | 15 | 16 | 17 | 18 | 19 |  |  |  |
| 20 | 21   | 22 | 23 | 24 | 25 | 26 |  |  |  |
| 27 | 28   | 29 | 30 |    |    |    |  |  |  |
|    |      |    |    |    |    |    |  |  |  |

| July |    |    |    |    |    |    |  |  |
|------|----|----|----|----|----|----|--|--|
| Su   | Мо | Tu | We | Th | Fr | Sa |  |  |
|      |    |    |    | 1  | 2  | 3  |  |  |
| 4    | 5  | 6  | 7  | 8  | 9  | 10 |  |  |
| 11   | 12 | 13 | 14 | 15 | 16 | 17 |  |  |
| 18   | 19 | 20 | 21 | 22 | 23 | 24 |  |  |
| 25   | 26 | 27 | 28 | 29 | 30 | 31 |  |  |
|      |    |    |    |    |    |    |  |  |

| August |    |    |    |    |    |    |  |  |  |
|--------|----|----|----|----|----|----|--|--|--|
| Su     | Мо | Tu | We | Th | Fr | Sa |  |  |  |
| 1      | 2  | 3  | 4  | 5  | 6  | 7  |  |  |  |
| 8      | 9  | 10 | 11 | 12 | 13 | 14 |  |  |  |
| 15     | 16 | 17 | 18 | 19 | 20 | 21 |  |  |  |
| 22     | 23 | 24 | 25 | 26 | 27 | 28 |  |  |  |
| 29     | 30 | 31 |    |    |    |    |  |  |  |
|        |    |    |    |    |    |    |  |  |  |

| September |                                   |    |    |    |    |    |  |  |  |
|-----------|-----------------------------------|----|----|----|----|----|--|--|--|
| Su        | <b>u</b> Mo Tu We Th Fr <b>Sa</b> |    |    |    |    |    |  |  |  |
|           |                                   |    | 1  | 2  | 3  | 4  |  |  |  |
| 5         | 6                                 | 7  | 8  | 9  | 10 | 11 |  |  |  |
| 12        | 13                                | 14 | 15 | 16 | 17 | 18 |  |  |  |
| 19        | 20                                | 21 | 22 | 23 | 24 | 25 |  |  |  |
| 26        | 27                                | 28 | 29 | 30 |    |    |  |  |  |
|           |                                   |    |    |    |    |    |  |  |  |

| October |    |    |    |    |    |    |  |  |
|---------|----|----|----|----|----|----|--|--|
| Su      | Мо | Tu | We | Th | Fr | Sa |  |  |
|         |    |    |    |    | 1  | 2  |  |  |
| 3       | 4  | 5  | 6  | 7  | 8  | 9  |  |  |
| 10      | 11 | 12 | 13 | 14 | 15 | 16 |  |  |
| 17      | 18 | 19 | 20 | 21 | 22 | 23 |  |  |
| 24      | 25 | 26 | 27 | 28 | 29 | 30 |  |  |
| 31      |    |    |    |    |    |    |  |  |

| November |    |                   |    |    |    |    |  |  |  |
|----------|----|-------------------|----|----|----|----|--|--|--|
| Su       | Мо | Mo Tu We Th Fr Sa |    |    |    |    |  |  |  |
|          | 1  | 2                 | 3  | 4  | 5  | 6  |  |  |  |
| 7        | 8  | 9                 | 10 | 11 | 12 | 13 |  |  |  |
| 14       | 15 | 16                | 17 | 18 | 19 | 20 |  |  |  |
| 21       | 22 | 23                | 24 | 25 | 26 | 27 |  |  |  |
| 28       | 29 | 30                |    |    |    |    |  |  |  |

| December |    |    |    |    |    |    |  |  |  |
|----------|----|----|----|----|----|----|--|--|--|
| Su       | Мо | Tu | We | Th | Fr | Sa |  |  |  |
|          |    |    | 1  | 2  | 3  | 4  |  |  |  |
| 5        | 6  | 7  | 8  | 9  | 10 | 11 |  |  |  |
| 12       | 13 | 14 | 15 | 16 | 17 | 18 |  |  |  |
| 19       | 20 | 21 | 22 | 23 | 24 | 25 |  |  |  |
| 26       | 27 | 28 | 29 | 30 | 31 |    |  |  |  |
|          |    |    |    |    |    |    |  |  |  |

| January |                          |    |    |    |    |    |  |  |
|---------|--------------------------|----|----|----|----|----|--|--|
| Su      | Mo Tu We Th Fr <b>Sa</b> |    |    |    |    |    |  |  |
|         |                          |    |    |    |    |    |  |  |
| 2       | 3                        | 4  | 5  | 6  | 7  | 8  |  |  |
| 9       | 10                       | 11 | 12 | 13 | 14 | 15 |  |  |
| 16      | 17                       | 18 | 19 | 20 | 21 | 22 |  |  |
| 23      | 24                       | 25 | 26 | 27 | 28 | 29 |  |  |
| 30      | 31                       |    |    |    |    |    |  |  |

|    | February |    |    |    |    |           |  |  |  |  |
|----|----------|----|----|----|----|-----------|--|--|--|--|
| Su | Мо       | Tu | We | Th | Fr | Sa        |  |  |  |  |
|    |          | 1  | 2  | 3  | 4  | 5         |  |  |  |  |
| 6  | 7        | 8  | 9  | 10 | 11 | 12        |  |  |  |  |
| 13 | 14       | 15 | 16 | 17 | 18 | 19        |  |  |  |  |
| 20 | 21       | 22 | 23 | 24 | 25 | <b>26</b> |  |  |  |  |
| 27 | 28       |    |    |    |    |           |  |  |  |  |
|    |          |    |    |    |    |           |  |  |  |  |

| March |    |    |    |    |    |           |  |  |
|-------|----|----|----|----|----|-----------|--|--|
| Su    | Мо | Tu | We | Th | Fr | Sa        |  |  |
|       |    | 1  | 2  | 3  | 4  | 5         |  |  |
| 6     | 7  | 8  | 9  | 10 | 11 | 12        |  |  |
| 13    | 14 | 15 | 16 | 17 | 18 | 19        |  |  |
| 20    | 21 | 22 | 23 | 24 | 25 | <b>26</b> |  |  |
| 27    | 28 | 29 | 30 | 31 |    |           |  |  |
|       |    |    |    |    |    |           |  |  |

| April |                            |    |    |    |    |    |  |  |  |
|-------|----------------------------|----|----|----|----|----|--|--|--|
| Su    | J Mo Tu We Th Fr <b>Sa</b> |    |    |    |    |    |  |  |  |
|       |                            |    |    |    | 1  | 2  |  |  |  |
| 3     | 4                          | 5  | 6  | 7  | 8  | 9  |  |  |  |
| 10    | 11                         | 12 | 13 | 14 | 15 | 16 |  |  |  |
| 17    | 18                         | 19 | 20 | 21 | 22 | 23 |  |  |  |
| 24    | 25                         | 26 | 27 | 28 | 29 | 30 |  |  |  |
|       |                            |    |    |    |    |    |  |  |  |

| Мау |    |    |    |    |    |    |  |  |  |
|-----|----|----|----|----|----|----|--|--|--|
| Su  | Мо | Tu | We | Th | Fr | Sa |  |  |  |
| 1   | 2  | 3  | 4  | 5  | 6  | 7  |  |  |  |
| 8   | 9  | 10 | 11 | 12 | 13 | 14 |  |  |  |
| 15  | 16 | 17 | 18 | 19 | 20 | 21 |  |  |  |
| 22  | 23 | 24 | 25 | 26 | 27 | 28 |  |  |  |
| 29  | 30 | 31 |    |    |    |    |  |  |  |
|     |    |    |    |    |    |    |  |  |  |

| June |    |    |    |    |    |    |  |  |
|------|----|----|----|----|----|----|--|--|
| Su   | Мо | Tu | We | Th | Fr | Sa |  |  |
|      |    |    | 1  | 2  | 3  | 4  |  |  |
| 5    | 6  | 7  | 8  | 9  | 10 | 11 |  |  |
| 12   | 13 | 14 | 15 | 16 | 17 | 18 |  |  |
| 19   | 20 | 21 | 22 | 23 | 24 | 25 |  |  |
| 26   | 27 | 28 | 29 | 30 |    |    |  |  |
|      |    |    |    |    |    |    |  |  |

| July |    |    |    |    |    |    |  |  |
|------|----|----|----|----|----|----|--|--|
| Su   | Мо | Tu | We | Th | Fr | Sa |  |  |
|      |    |    |    |    | 1  | 2  |  |  |
| 3    | 4  | 5  | 6  | 7  | 8  | 9  |  |  |
| 10   | 11 | 12 | 13 | 14 | 15 | 16 |  |  |
| 17   | 18 | 19 | 20 | 21 | 22 | 23 |  |  |
| 24   | 25 | 26 | 27 | 28 | 29 | 30 |  |  |
| 31   |    |    |    |    |    |    |  |  |

| August |    |    |    |    |    |    |  |  |
|--------|----|----|----|----|----|----|--|--|
| Su     | Мо | Tu | We | Th | Fr | Sa |  |  |
|        | 1  | 2  | 3  | 4  | 5  | 6  |  |  |
| 7      | 8  | 9  | 10 | 11 | 12 | 13 |  |  |
| 14     | 15 | 16 | 17 | 18 | 19 | 20 |  |  |
| 21     | 22 | 23 | 24 | 25 | 26 | 27 |  |  |
| 28     | 29 | 30 | 31 |    |    |    |  |  |
|        |    |    |    |    |    |    |  |  |

| September |    |    |    |    |    |    |  |  |  |
|-----------|----|----|----|----|----|----|--|--|--|
| Su        | Мо | Tu | We | Th | Fr | Sa |  |  |  |
|           |    |    |    | 1  | 2  | 3  |  |  |  |
| 4         | 5  | 6  | 7  | 8  | 9  | 10 |  |  |  |
| 11        | 12 | 13 | 14 | 15 | 16 | 17 |  |  |  |
| 18        | 19 | 20 | 21 | 22 | 23 | 24 |  |  |  |
| 25        | 26 | 27 | 28 | 29 | 30 |    |  |  |  |
|           |    |    |    |    |    |    |  |  |  |

| October |    |    |    |    |    |    |  |  |
|---------|----|----|----|----|----|----|--|--|
| Su      | Мо | Tu | We | Th | Fr | Sa |  |  |
|         |    |    |    |    |    | 1  |  |  |
| 2       | 3  | 4  | 5  | 6  | 7  | 8  |  |  |
| 9       | 10 | 11 | 12 | 13 | 14 | 15 |  |  |
| 16      | 17 | 18 | 19 | 20 | 21 | 22 |  |  |
| 23      | 24 | 25 | 26 | 27 | 28 | 29 |  |  |
| 30      | 31 |    |    |    |    |    |  |  |

| November |                   |    |    |    |    |    |  |  |  |
|----------|-------------------|----|----|----|----|----|--|--|--|
| Su       | Mo Tu We Th Fr Sa |    |    |    |    |    |  |  |  |
|          |                   | 1  | 2  | 3  | 4  | 5  |  |  |  |
| 6        | 7                 | 8  | 9  | 10 | 11 | 12 |  |  |  |
| 13       | 14                | 15 | 16 | 17 | 18 | 19 |  |  |  |
| 20       | 21                | 22 | 23 | 24 | 25 | 26 |  |  |  |
| 27       | 28                | 29 | 30 |    |    |    |  |  |  |

| December |    |    |    |    |    |    |  |  |
|----------|----|----|----|----|----|----|--|--|
| Su       | Мо | Tu | We | Th | Fr | Sa |  |  |
|          |    |    |    | 1  | 2  | 3  |  |  |
| 4        | 5  | 6  | 7  | 8  | 9  | 10 |  |  |
| 11       | 12 | 13 | 14 | 15 | 16 | 17 |  |  |
| 18       | 19 | 20 | 21 | 22 | 23 | 24 |  |  |
| 25       | 26 | 27 | 28 | 29 | 30 | 31 |  |  |
|          |    |    |    |    |    |    |  |  |

# 2023

| January |    |    |    |    |    |    |  |  |
|---------|----|----|----|----|----|----|--|--|
| Su      | Мо | Tu | We | Th | Fr | Sa |  |  |
| 1       | 2  | 3  | 4  | 5  | 6  | 7  |  |  |
| 8       | 9  | 10 | 11 | 12 | 13 | 14 |  |  |
| 15      | 16 | 17 | 18 | 19 | 20 | 21 |  |  |
| 22      | 23 | 24 | 25 | 26 | 27 | 28 |  |  |
| 29      | 30 | 31 |    |    |    |    |  |  |
|         |    |    |    |    |    |    |  |  |

| February |    |    |    |    |    |    |  |  |
|----------|----|----|----|----|----|----|--|--|
| Su       | Мо | Tu | We | Th | Fr | Sa |  |  |
|          |    |    | 1  | 2  | 3  | 4  |  |  |
| 5        | 6  | 7  | 8  | 9  | 10 | 11 |  |  |
| 12       | 13 | 14 | 15 | 16 | 17 | 18 |  |  |
| 19       | 20 | 21 | 22 | 23 | 24 | 25 |  |  |
| 26       | 27 | 28 |    |    |    |    |  |  |

| March |    |    |    |    |    |    |  |  |
|-------|----|----|----|----|----|----|--|--|
| Su    | Мо | Tu | We | Th | Fr | Sa |  |  |
|       |    |    | 1  | 2  | 3  | 4  |  |  |
| 5     | 6  | 7  | 8  | 9  | 10 | 11 |  |  |
| 12    | 13 | 14 | 15 | 16 | 17 | 18 |  |  |
| 19    | 20 | 21 | 22 | 23 | 24 | 25 |  |  |
| 26    | 27 | 28 | 29 | 30 | 31 |    |  |  |
|       |    |    |    |    |    |    |  |  |

| April |    |                         |    |    |    |           |  |  |  |
|-------|----|-------------------------|----|----|----|-----------|--|--|--|
| Su    | Мо | Mo Tu We Th Fr <b>S</b> |    |    |    |           |  |  |  |
|       |    |                         |    |    |    | 1         |  |  |  |
| 2     | 3  | 4                       | 5  | 6  | 7  | 8         |  |  |  |
| 9     | 10 | 11                      | 12 | 13 | 14 | 15        |  |  |  |
| 16    | 17 | 18                      | 19 | 20 | 21 | 22        |  |  |  |
| 23    | 24 | 25                      | 26 | 27 | 28 | <b>29</b> |  |  |  |
| 30    |    |                         |    |    |    |           |  |  |  |

|    |    |    | Мау | /  |    |    |
|----|----|----|-----|----|----|----|
| Su | Мо | Tu | We  | Th | Fr | Sa |
|    | 1  | 2  | 3   | 4  | 5  | 6  |
| 7  | 8  | 9  | 10  | 11 | 12 | 13 |
| 14 | 15 | 16 | 17  | 18 | 19 | 20 |
| 21 | 22 | 23 | 24  | 25 | 26 | 27 |
| 28 | 29 | 30 | 31  |    |    |    |
|    |    |    |     |    |    |    |

| June |    |    |    |    |    |    |  |  |
|------|----|----|----|----|----|----|--|--|
| Su   | Мо | Tu | We | Th | Fr | Sa |  |  |
|      |    |    |    | 1  | 2  | 3  |  |  |
| 4    | 5  | 6  | 7  | 8  | 9  | 10 |  |  |
| 11   | 12 | 13 | 14 | 15 | 16 | 17 |  |  |
| 18   | 19 | 20 | 21 | 22 | 23 | 24 |  |  |
| 25   | 26 | 27 | 28 | 29 | 30 |    |  |  |
|      |    |    |    |    |    |    |  |  |

| July |    |    |    |    |    |    |  |  |
|------|----|----|----|----|----|----|--|--|
| Su   | Мо | Tu | We | Th | Fr | Sa |  |  |
|      |    |    |    |    |    | 1  |  |  |
| 2    | 3  | 4  | 5  | 6  | 7  | 8  |  |  |
| 9    | 10 | 11 | 12 | 13 | 14 | 15 |  |  |
| 16   | 17 | 18 | 19 | 20 | 21 | 22 |  |  |
| 23   | 24 | 25 | 26 | 27 | 28 | 29 |  |  |
| 30   | 31 |    |    |    |    |    |  |  |

| August |    |    |    |    |    |           |  |  |  |
|--------|----|----|----|----|----|-----------|--|--|--|
| Su     | Мо | Tu | We | Th | Fr | Sa        |  |  |  |
|        |    | 1  | 2  | 3  | 4  | 5         |  |  |  |
| 6      | 7  | 8  | 9  | 10 | 11 | 12        |  |  |  |
| 13     | 14 | 15 | 16 | 17 | 18 | 19        |  |  |  |
| 20     | 21 | 22 | 23 | 24 | 25 | <b>26</b> |  |  |  |
| 27     | 28 | 29 | 30 | 31 |    |           |  |  |  |
|        |    |    |    |    |    |           |  |  |  |

| September |    |    |    |    |    |    |  |  |
|-----------|----|----|----|----|----|----|--|--|
| Su        | Мо | Tu | We | Th | Fr | Sa |  |  |
|           |    |    |    |    | 1  | 2  |  |  |
| 3         | 4  | 5  | 6  | 7  | 8  | 9  |  |  |
| 10        | 11 | 12 | 13 | 14 | 15 | 16 |  |  |
| 17        | 18 | 19 | 20 | 21 | 22 | 23 |  |  |
| 24        | 25 | 26 | 27 | 28 | 29 | 30 |  |  |
|           |    |    |    |    |    |    |  |  |

| October |    |    |    |    |    |    |  |  |
|---------|----|----|----|----|----|----|--|--|
| Su      | Мо | Tu | We | Th | Fr | Sa |  |  |
| 1       | 2  | 3  | 4  | 5  | 6  | 7  |  |  |
| 8       | 9  | 10 | 11 | 12 | 13 | 14 |  |  |
| 15      | 16 | 17 | 18 | 19 | 20 | 21 |  |  |
| 22      | 23 | 24 | 25 | 26 | 27 | 28 |  |  |
| 29      | 30 | 31 |    |    |    |    |  |  |
|         |    |    |    |    |    |    |  |  |

| November |    |    |    |    |    |    |  |  |  |
|----------|----|----|----|----|----|----|--|--|--|
| Su       | Мо | Tu | We | Th | Fr | Sa |  |  |  |
|          |    |    | 1  | 2  | 3  | 4  |  |  |  |
| 5        | 6  | 7  | 8  | 9  | 10 | 11 |  |  |  |
| 12       | 13 | 14 | 15 | 16 | 17 | 18 |  |  |  |
| 19       | 20 | 21 | 22 | 23 | 24 | 25 |  |  |  |
| 26       | 27 | 28 | 29 | 30 |    |    |  |  |  |
|          |    |    |    |    |    |    |  |  |  |

| December |    |           |    |    |    |    |  |  |  |
|----------|----|-----------|----|----|----|----|--|--|--|
| Su       | Мо | Tu        | We | Th | Fr | Sa |  |  |  |
|          |    |           |    |    | 1  | 2  |  |  |  |
| 3        | 4  | 5         | 6  | 7  | 8  | 9  |  |  |  |
| 10       | 11 | 12        | 13 | 14 | 15 | 16 |  |  |  |
| 17       | 18 | 19        | 20 | 21 | 22 | 23 |  |  |  |
| 24       | 25 | <b>26</b> | 27 | 28 | 29 | 30 |  |  |  |
| 31       |    |           | -  |    |    |    |  |  |  |

| January |    |    |    |    |    |    |  |  |
|---------|----|----|----|----|----|----|--|--|
| Su      | Мо | Tu | We | Th | Fr | Sa |  |  |
|         | 1  | 2  | 3  | 4  | 5  | 6  |  |  |
| 7       | 8  | 9  | 10 | 11 | 12 | 13 |  |  |
| 14      | 15 | 16 | 17 | 18 | 19 | 20 |  |  |
| 21      | 22 | 23 | 24 | 25 | 26 | 27 |  |  |
| 28      | 29 | 30 | 31 |    |    |    |  |  |
|         |    |    |    |    |    |    |  |  |

| February |    |    |    |    |    |    |  |  |  |
|----------|----|----|----|----|----|----|--|--|--|
| Su       | Мо | Tu | We | Th | Fr | Sa |  |  |  |
|          |    |    |    | 1  | 2  | 3  |  |  |  |
| 4        | 5  | 6  | 7  | 8  | 9  | 10 |  |  |  |
| 11       | 12 | 13 | 14 | 15 | 16 | 17 |  |  |  |
| 18       | 19 | 20 | 21 | 22 | 23 | 24 |  |  |  |
| 25       | 26 | 27 | 28 | 29 |    |    |  |  |  |
|          |    |    |    |    |    |    |  |  |  |

| March |    |    |    |    |    |    |  |  |
|-------|----|----|----|----|----|----|--|--|
| Su    | Мо | Tu | We | Th | Fr | Sa |  |  |
|       |    |    |    |    | 1  | 2  |  |  |
| 3     | 4  | 5  | 6  | 7  | 8  | 9  |  |  |
| 10    | 11 | 12 | 13 | 14 | 15 | 16 |  |  |
| 17    | 18 | 19 | 20 | 21 | 22 | 23 |  |  |
| 24    | 25 | 26 | 27 | 28 | 29 | 30 |  |  |
| 31    |    |    |    |    |    |    |  |  |

|    | April |    |    |    |    |    |  |  |  |  |
|----|-------|----|----|----|----|----|--|--|--|--|
| Su | Мо    | Tu | We | Th | Fr | Sa |  |  |  |  |
|    | 1     | 2  | 3  | 4  | 5  | 6  |  |  |  |  |
| 7  | 8     | 9  | 10 | 11 | 12 | 13 |  |  |  |  |
| 14 | 15    | 16 | 17 | 18 | 19 | 20 |  |  |  |  |
| 21 | 22    | 23 | 24 | 25 | 26 | 27 |  |  |  |  |
| 28 | 29    | 30 |    |    |    |    |  |  |  |  |
|    |       |    |    |    |    |    |  |  |  |  |

| Мау |    |    |    |    |    |    |  |
|-----|----|----|----|----|----|----|--|
| Su  | Мо | Tu | We | Th | Fr | Sa |  |
|     |    |    | 1  | 2  | 3  | 4  |  |
| 5   | 6  | 7  | 8  | 9  | 10 | 11 |  |
| 12  | 13 | 14 | 15 | 16 | 17 | 18 |  |
| 19  | 20 | 21 | 22 | 23 | 24 | 25 |  |
| 26  | 27 | 28 | 29 | 30 | 31 |    |  |
|     |    |    |    |    |    |    |  |

| June |    |    |    |    |    |    |  |
|------|----|----|----|----|----|----|--|
| Su   | Мо | Tu | We | Th | Fr | Sa |  |
|      |    |    |    |    |    | 1  |  |
| 2    | 3  | 4  | 5  | 6  | 7  | 8  |  |
| 9    | 10 | 11 | 12 | 13 | 14 | 15 |  |
| 16   | 17 | 18 | 19 | 20 | 21 | 22 |  |
| 23   | 24 | 25 | 26 | 27 | 28 | 29 |  |
| 30   |    |    |    |    |    |    |  |

| July |                           |   |  |                                    |   |  |  |  |  |
|------|---------------------------|---|--|------------------------------------|---|--|--|--|--|
| Мо   | Tu                        | We  | Th   | Fr                                 | Sa  |  |  |  |  |
| 1    | 2                         | 3   | 4  | 5                                  | 6   |  |  |  |  |
| 8    | 9                         | 10  | 11   | 12                                 | 13  |  |  |  |  |
| 15   | 16                        | 17  | 18   | 19                                 | 20  |  |  |  |  |
| 22   | 23                        | 24  | 25   | 26                                 | 27  |  |  |  |  |
| 29   | 30                        | 31  |  |                                    |   |  |  |  |  |
|      | <b>1</b><br>8<br>15<br>22 | Mo         Tu           1         2           8         9           15         16           22         23 | Mo         Tu         We           1         2         3           8         9         10           15         16         17 | MoTuWeTh12348910111516171822232425 | MoTuWeThFr123458910111215161718192223242526 |  |  |  |  |

| August |    |            |    |    |    |    |  |  |  |
|--------|----|------------|----|----|----|----|--|--|--|
| Su     | Мо | Mo Tu We T |    | Th | Fr | Sa |  |  |  |
|        |    |            |    | 1  | 2  | 3  |  |  |  |
| 4      | 5  | 6          | 7  | 8  | 9  | 10 |  |  |  |
| 11     | 12 | 13         | 14 | 15 | 16 | 17 |  |  |  |
| 18     | 19 | 20         | 21 | 22 | 23 | 24 |  |  |  |
| 25     | 26 | 27         | 28 | 29 | 30 | 31 |  |  |  |
|        |    |            |    |    |    |    |  |  |  |

| September |    |                  |    |    |    |           |  |  |  |
|-----------|----|------------------|----|----|----|-----------|--|--|--|
| Su        | Мо | Mo Tu We Th Fr S |    |    |    |           |  |  |  |
| 1         | 2  | 3                | 4  | 5  | 6  | 7         |  |  |  |
| 8         | 9  | 10               | 11 | 12 | 13 | 14        |  |  |  |
| 15        | 16 | 17               | 18 | 19 | 20 | 21        |  |  |  |
| 22        | 23 | 24               | 25 | 26 | 27 | <b>28</b> |  |  |  |
| 29        | 30 |                  |    |    |    |           |  |  |  |
|           |    |                  |    |    |    |           |  |  |  |

| October |    |    |    |    |    |           |  |  |  |
|---------|----|----|----|----|----|-----------|--|--|--|
| Su      | Мо | Tu | Fr | Sa |    |           |  |  |  |
|         |    | 1  | 2  | 3  | 4  | 5         |  |  |  |
| 6       | 7  | 8  | 9  | 10 | 11 | 12        |  |  |  |
| 13      | 14 | 15 | 16 | 17 | 18 | 19        |  |  |  |
| 20      | 21 | 22 | 23 | 24 | 25 | <b>26</b> |  |  |  |
| 27      | 28 | 29 | 30 | 31 |    |           |  |  |  |
|         |    |    |    |    |    |           |  |  |  |

| November |    |    |    |    |    |    |  |  |  |
|----------|----|----|----|----|----|----|--|--|--|
| Su       | Мо | Fr | Sa |    |    |    |  |  |  |
|          |    |    |    |    | 1  | 2  |  |  |  |
| 3        | 4  | 5  | 6  | 7  | 8  | 9  |  |  |  |
| 10       | 11 | 12 | 13 | 14 | 15 | 16 |  |  |  |
| 17       | 18 | 19 | 20 | 21 | 22 | 23 |  |  |  |
| 24       | 25 | 26 | 27 | 28 | 29 | 30 |  |  |  |
|          |    |    |    |    |    |    |  |  |  |

| December |    |    |    |    |    |           |  |  |  |  |
|----------|----|----|----|----|----|-----------|--|--|--|--|
| Su       | Мо | Tu | We | Th | Fr | Sa        |  |  |  |  |
| 1        | 2  | 3  | 4  | 5  | 6  | 7         |  |  |  |  |
| 8        | 9  | 10 | 11 | 12 | 13 | 14        |  |  |  |  |
| 15       | 16 | 17 | 18 | 19 | 20 | 21        |  |  |  |  |
| 22       | 23 | 24 | 25 | 26 | 27 | <b>28</b> |  |  |  |  |
| 29       | 30 | 31 |    |    |    |           |  |  |  |  |
|          |    |    |    |    |    |           |  |  |  |  |

# 2025

| January |    |    |    |    |    |    |  |  |
|---------|----|----|----|----|----|----|--|--|
| Su      | Мо | Tu | We | Th | Fr | Sa |  |  |
|         |    |    | 1  | 2  | 3  | 4  |  |  |
| 5       | 6  | 7  | 8  | 9  | 10 | 11 |  |  |
| 12      | 13 | 14 | 15 | 16 | 17 | 18 |  |  |
| 19      | 20 | 21 | 22 | 23 | 24 | 25 |  |  |
| 26      | 27 | 28 | 29 | 30 | 31 |    |  |  |
|         |    |    |    |    |    |    |  |  |

| February |                   |    |    |    |    |    |  |  |  |
|----------|-------------------|----|----|----|----|----|--|--|--|
| Su       | Mo Tu We Th Fr Sa |    |    |    |    |    |  |  |  |
|          |                   |    |    |    |    | 1  |  |  |  |
| 2        | 3                 | 4  | 5  | 6  | 7  | 8  |  |  |  |
| 9        | 10                | 11 | 12 | 13 | 14 | 15 |  |  |  |
| 16       | 17                | 18 | 19 | 20 | 21 | 22 |  |  |  |
| 23       | 24                | 25 | 26 | 27 | 28 | 29 |  |  |  |
|          |                   |    |    |    |    |    |  |  |  |

| March |    |    |    |    |    |    |  |  |  |
|-------|----|----|----|----|----|----|--|--|--|
| Su    | Мо | Tu | We | Th | Fr | Sa |  |  |  |
| 1     | 2  | 3  | 4  | 5  | 6  | 7  |  |  |  |
| 8     | 9  | 10 | 11 | 12 | 13 | 14 |  |  |  |
| 15    | 16 | 17 | 18 | 19 | 20 | 21 |  |  |  |
| 22    | 23 | 24 | 25 | 26 | 27 | 28 |  |  |  |
| 29    | 30 | 31 |    |    |    |    |  |  |  |
|       |    |    |    |    |    |    |  |  |  |

| April                |    |    |    |    |    |    |  |  |  |
|----------------------|----|----|----|----|----|----|--|--|--|
| Su Mo Tu We Th Fr Sa |    |    |    |    |    |    |  |  |  |
|                      |    |    | 1  | 2  | 3  | 4  |  |  |  |
| 5                    | 6  | 7  | 8  | 9  | 10 | 11 |  |  |  |
| 12                   | 13 | 14 | 15 | 16 | 17 | 18 |  |  |  |
| 19                   | 20 | 21 | 22 | 23 | 24 | 25 |  |  |  |
| 26                   | 27 | 28 | 29 | 30 |    |    |  |  |  |
|                      |    |    |    |    |    |    |  |  |  |

| Мау |    |    |    |    |    |    |  |  |
|-----|----|----|----|----|----|----|--|--|
| Su  | Мо | Fr | Sa |    |    |    |  |  |
|     |    |    |    |    | 1  | 2  |  |  |
| 3   | 4  | 5  | 6  | 7  | 8  | 9  |  |  |
| 10  | 11 | 12 | 13 | 14 | 15 | 16 |  |  |
| 17  | 18 | 19 | 20 | 21 | 22 | 23 |  |  |
| 24  | 25 | 26 | 27 | 28 | 29 | 30 |  |  |
| 31  |    |    |    |    |    |    |  |  |

| June |    |    |    |    |    |    |
|------|----|----|----|----|----|----|
| Su   | Мо | Tu | We | Th | Fr | Sa |
|      | 1  | 2  | 3  | 4  | 5  | 6  |
| 7    | 8  | 9  | 10 | 11 | 12 | 13 |
| 14   | 15 | 16 | 17 | 18 | 19 | 20 |
| 21   | 22 | 23 | 24 | 25 | 26 | 27 |
| 28   | 29 | 30 |    |    |    |    |

| July |    |    |    |    |    |    |
|------|----|----|----|----|----|----|
| Su   | Мо | Tu | We | Th | Fr | Sa |
|      |    |    | 1  | 2  | 3  | 4  |
| 5    | 6  | 7  | 8  | 9  | 10 | 11 |
| 12   | 13 | 14 | 15 | 16 | 17 | 18 |
| 19   | 20 | 21 | 22 | 23 | 24 | 25 |
| 26   | 27 | 28 | 29 | 30 | 31 |    |
|      |    |    |    |    |    |    |

| August |    |    |    |    |    |    |
|--------|----|----|----|----|----|----|
| Su     | Мо | Tu | We | Th | Fr | Sa |
|        |    |    |    |    |    | 1  |
| 2      | 3  | 4  | 5  | 6  | 7  | 8  |
| 9      | 10 | 11 | 12 | 13 | 14 | 15 |
| 16     | 17 | 18 | 19 | 20 | 21 | 22 |
| 23     | 24 | 25 | 26 | 27 | 28 | 29 |
| 30     | 31 |    |    |    |    |    |

| September |    |    |    |    |    |    |
|-----------|----|----|----|----|----|----|
| Su        | Мо | Tu | We | Th | Fr | Sa |
|           |    | 1  | 2  | 3  | 4  | 5  |
| 6         | 7  | 8  | 9  | 10 | 11 | 12 |
| 13        | 14 | 15 | 16 | 17 | 18 | 19 |
| 20        | 21 | 22 | 23 | 24 | 25 | 26 |
|           | 28 |    |    |    |    |    |
|           |    |    |    |    |    |    |

| October |    |    |    |    |    |    |
|---------|----|----|----|----|----|----|
| Su      | Мо | Tu | We | Th | Fr | Sa |
|         |    |    |    | 1  | 2  | 3  |
| 4       | 5  | 6  | 7  | 8  | 9  | 10 |
| 11      | 12 | 13 | 14 | 15 | 16 | 17 |
| 18      | 19 | 20 | 21 | 22 | 23 | 24 |
| 25      | 26 | 27 | 28 | 29 | 30 | 31 |
|         |    |    |    |    |    |    |

| November |    |    |    |    |    |    |
|----------|----|----|----|----|----|----|
| Su       | Мо | Tu | We | Th | Fr | Sa |
| 1        | 2  | 3  | 4  | 5  | 6  | 7  |
| 8        | 9  | 10 | 11 | 12 | 13 | 14 |
| 15       | 16 | 17 | 18 | 19 | 20 | 21 |
| 22       | 23 | 24 | 25 | 26 | 27 | 28 |
| 29       | 30 |    |    |    |    |    |

| December |    |    |    |    |    |    |
|----------|----|----|----|----|----|----|
| Su       | Мо | Tu | We | Th | Fr | Sa |
|          |    | 1  | 2  | 3  | 4  | 5  |
| 6        | 7  | 8  | 9  | 10 | 11 | 12 |
| 13       | 14 | 15 | 16 | 17 | 18 | 19 |
| 20       | 21 | 22 | 23 | 24 | 25 | 26 |
| 27       | 28 | 29 | 30 | 31 |    |    |
|          |    |    |    |    |    |    |

# **BLACK BOOK** Letters of Agreement

between

WESTSHORE TERMINALS LIMITED PARTNERSHIP

and

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

July 31, 2017

#### **BLACK BOOK LETTERS**

This letter will confirm the understanding and agreement between the Parties hereto with respect to all matters related to this document, which is commonly referred to as the "Black Book".

This book contains Letters of Understanding or Agreement between the Parties hereto with respect to the interpretation, application and administration of the Collective Agreement between the Parties, and other matters relating to the on-going relationship between the Company, the Union and the Employees.

With respect to all letters contained herein, or to be agreed to at future dates, the Parties mutually agree to the following:

- 1. All letters contained herein shall remain in effect for the term of the Collective Agreement, as defined in the Canada Labour Code, or for such other period as specifically stated in any individual Letter of Understanding or Agreement contained herein or agreed to at some future date.
- 2. No Letter of Understanding or Agreement shall be appended to this document unless both Parties mutually agree, in writing, to include such letter or letters.

In addition to the above, the Parties hereto agree that this document contains all letters between the Parties that are in effect as of the signing of this Letter of Agreement and that all letters previously agreed to by the Parties and not contained herein are null and void.

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

**Rick Hurtubise** President, Local 502

# **Table of Contents**

| R1- 76 | Bulldozing of Coal Piles                       | 4  |
|--------|--|----|
| R1- 77 | Employee Transfers                             | 5  |
| R2- 81 | Employee Files                                 | 7  |
| R3- 82 | Birthday & Holiday for Non-Permanent Employees | 8  |
| R4- 82 | Refusal to Work for Health & Safety Reasons    | 9  |
| R1- 84 | W.S.T Gangplanks                               | 10 |
| R2- 84 | Berth #1 & Berth #2 – Safety Boat              | 11 |
| R3- 84 | Dispatch Fees                                  | 12 |
| R1- 85 | Pollution Control/Emergency Crew               | 13 |
| R1- 87 | Severance Pay - Policy Statement               | 15 |
| R2- 87 | Belt Splicing                                  | 16 |
| R4- 88 | Electricians                                   | 18 |
| R5- 88 | Manning  | 19 |
| R1-89  | Orientation/Evaluation &Training Procedures    | 20 |
| R1- 90 | Shift Operators                                | 24 |
| R2- 90 | Job Postings                                   | 26 |
| R2- 92 | Direct Deposit Payroll System                  | 29 |
| R1- 93 | Rehired Employees                              | 31 |
| R1- 94 | Payment for Local 502 Bargaining Committee     | 32 |
| R2- 94 | Utilization of Extras (HD) T.E.                | 33 |
| R4- 94 | Electrical Printmen                            | 35 |
| R6- 94 | Technologist Systems                           | 36 |
| R2- 95 | Utilization of Extra Temporary Operator        | 43 |
| R1- 97 | Maintenance Night Shift                        | 44 |
| R2- 97 | Cleanup Personnel                              | 46 |

# **Table of Contents**

| R3- 97  | Hydraulics and Electrical Motors                 |    |
|---------|--|----|
| R5- 97  | 15.04 Temporary Employee - Seniority             | 50 |
| R1- 03  | MDA/MDB Schedules                                | 51 |
| R2- 03  | M5 Schedule Language in the Collective Agreement | 52 |
| I- 07   | No One Available & Banking                       | 53 |
| 2- 07   | Severance  | 54 |
| R1 3-11 | Automation Agreement                             | 55 |
| 4- 11   | Automation Protocol                              | 57 |
| 5- 11   | Control Room Operator Protocol                   | 58 |
| 6- 11   | Subrogation Claims                               | 59 |
| 8- 11   | Apprentices Hours                                | 61 |
| 9- 11   | Electrical Department - Printmen                 | 62 |
| 2- 16   | Dedicated Spare List                             | 63 |
| 3- 16   | Dispatch Upgrade                                 | 65 |
| 4- 16   | Revenue Generating Product                       | 66 |
| 5- 16   | Scissor Lift & Third Party Work                  | 67 |
| 6- 16   | Safety   | 68 |
| 7- 16   | Apprentice Program                               | 69 |
| 8- 16   | Article 24.04 (a) Interpretation Guide           | 73 |
| 9- 16   | Amended Automation Agreement                     | 75 |
| 10-16   | Shift Operators                                  | 76 |
| 11-16   | Troubleshooters                                  | 77 |
| 12-16   | Pollution Control System                         | 78 |
| 1-21    | Mobile Equipment                                 | 81 |
| 2-21    | Dry Facilities                                   | 82 |
| 3-21    | Safe Work Procedure                              | 83 |

# **Table of Contents**

| R1-4-21 | Gearman                           | .84 |
|---------|-----------------------------------|-----|
| R1-5-21 | Building Cleaning Duties Division | .85 |
| 6-21    | Maintenance Orientation           | .87 |
| 7-21    | Electrical Mentors                | .88 |

# LETTER OF AGREEMENT between WESTSHORE TERMINALS LIMITED PARTNERSHIP and **INTERNATIONAL LONGSHORE AND** WAREHOUSE UNION, LOCAL 502

| То:        | Bruce Ferguson | R. Prince   |
|------------|----------------|-------------|
| Copies To: | G. Scott       | C. Goodall  |
|            | Shift Foremen  | H. Tardt    |
|            | Main. Foremen  | G. Taverner |
|            | Bulletin Board | H. Volkmann |
|            | R. Humphries   | B. Roberts  |

#### SUBJECT: BULLDOZING OF COAL PILES

Dozers will continue to be used on all shifts as required with the following limitations:

- a) All dozers will be provided with a closed-in cab to protect the operator from the weather.
- b) Dozing for stockpiling or reclaiming of high piles will only be done during daylight hours.
- c) Cleanup work by dozers will be performed during non-daylight hours only when required to work at ground level to cleanup spill conditions or where necessary to protect and safeguard equipment from the sluffing or slipping of coal piles caused by excessive watering of piles, heavy rain conditions, or improper stockpiling of coal.
- d) The dozing of coal during non-daylight hours may be performed when a minimum amount of coal (not to exceed 2,500 tonnes) is required to be dozed to complete a cargo or vessel, provided the operator is satisfied that a safety hazard does not exist.
- Further to the provisions of (d) above, dozing of coal for reclaiming production only during non-daylight hours e) may be performed for low piles, provided the operating procedures and rules covering such work have been reviewed and approved by the Plant Committee.
- (Revised April 18, 1984 signed by R. Elke and D. Skoyen) •
- (Amended and signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
- (Reprinted and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Reprinted and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

From: G. Scott

WAREHOUSE UNION, LOCAL 502

**Rick Hurtubise** President, Local 502

#### SUBJECT: EMPLOYEE TRANSFERS

This is to confirm and acknowledge the understanding between Westshore Terminals Ltd. and the I.L.W.U. Local 502 concerning:

The Transfer of Employees as replacements for regular Employees who are absent as a result of Sickness, Accidents, Book-offs, Leaves of absence, or Vacation but excluding transfers as a result of job postings. The terms and conditions herein will only apply after the provisions of Article 20 have been exhausted.

1.

- a) If an Employee, including an Employee who is on vacation or on scheduled days off, is notified by the Company that he will be temporarily transferred to another shift schedule when he returns to work, then the Employee will be paid a 9 X 2 hour premium for the first shift worked on the temporary schedule subject to (b) and (c) below.
- b) If an Employee is given less than 24 hours notice of a temporary transfer to another crew or is given less than 16 hours off before commencing work on a temporary schedule then the Employee will be paid a 9 X 2 hour premium for the first shift worked on the temporary schedule.
- c) If an Employee, excluding an Employee who is on vacation or on scheduled days off, is given twenty-four (24) hours or more notice of a temporary transfer to another shift schedule or if an Employee is given sixteen (16) hours or more off work before commencing work on the temporary schedule then (a) and (b) above shall not apply.
- 2. Short Shifting (i.e. less than sixteen (16) hours off prior to commencing the temporary schedule or less than twenty-four (24) hours notice of the transfer to the temporary schedule) will not apply when changing from afternoon shift to day shift or day shift to graveyard shift, or changing to a temporary schedule or returning to the Employees base schedule as a result of a transfer required by a posting or a transfer requested by an Employee.
- 3. An Employee who transfers from one schedule to another schedule will be paid at the shift rate applicable to the shift worked.
- 4. If an Employee is transferred to a temporary schedule for more than two (2) weeks the Employee will be compensated as follows:
  - a) The Employee will follow his base schedule up to and including the day immediately before the first day of the temporary schedule then change crews and begin to follow the temporary schedule. If, as a result of the crew change, the Employee works, more days, during the payroll week of the change than he was scheduled to work on his base schedule then the Employee will receive overtime for those additional days.
  - b) If the end of the transfer coincides with the last day of work on the temporary schedule the Employee will continue on the temporary schedule until the last scheduled day off following the transfer period. The Employee will then revert to his base schedule.
  - c) If the end of the transfer period does not coincide with the last day of work on the temporary schedule then the Employee will revert to his base schedule the day after the last day of the transfer. If, as a result, the Employee works more days during the payroll week of change than he was scheduled to work on the temporary schedule then the Employee will receive overtime for those additional days.

- a) If an Employee is transferred to a temporary schedule for two (2) payroll weeks or less then such Employee will be paid overtime for days worked in excess of his base schedule in the week of change.
- b) If an Employee is transferred to a temporary schedule for two (2) payroll weeks or less, and if as a result of the transfer the Employee is required to work more than six (6) consecutive days, then the Employee shall be paid one hour at double time for each day in excess of six (6) consecutive days provided that the Employee is not already entitled to receive premium or overtime pay for such day or days.
- 6. Corresponding Crew

The replacements for preferred categories, Warehouse, Day Warehouse and Shift Operator, for absences in excess of three (3) days, where the requirements of the Black Book Letters 1-90 and R2-90 have been met, then the provisions of Article 20, subsection 20.01 (b) (i) shall apply in the following sequence:

- a) replace from the corresponding crew and the Employee pulled from such crew shall be replaced from the first day from the Union Hall.
- b) replace from the adjacent crew (the crew whose schedule is closest to matching the period of absence) and the Employee transferred shall be replaced from the Hall and work all of the days of the temporary schedule, excluding days in excess of such Employee's base schedule. The excess days shall be replaced from the Overtime Board. Where the Company replaces from the adjacent crew the down rules shall not apply.
- c) If no trained Employee is available through the application of (a) or (b) above, the Employee will be replaced from the Overtime Board, or in the case of a Shift Operator, by the methods stated in Black Book letter R 1-90.

#### 7. Warehouse Categories

Absences of less than three (3) days for Warehouse categories shall be replaced as follows:

- a) Overtime
- b) Corresponding Crew
- c) Warehouseman on the overtime board who has not made himself available
- d) Adjacent Crew
- e) Rated Employee on site by seniority

Employees transferred, in accordance with the above, shall be replaced by Article 20.

- (Revised July 13, 1990 signed by P. Lanosky and J. Hoskins)
- (Revised November 29, 1994 signed by A. Kazuta and B. Ringrose)
- (Amended and signed October 19, 2000 by Alan Kazuta and Brian Ringrose)
- (Reprinted and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Reprinted and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### **SUBJECT: EMPLOYEE FILES**

This Letter confirms the agreement between the parties with regard to the matter of materials contained in Employee files:

- 1. The Company policy is that documents which relate to matters of discipline, that are placed in an Employee's Human Resources file (HR file), will be copied to the Employee and a copy will be sent to the Union President.
- 2. The Company will review each Employee's HR file's once per year and any document relating to discipline that was given more than two (2) years prior, shall be removed.
- 3. Employees are entitled to review their HR files by giving a minimum of one week notice to the <u>Director</u>, Human Resources. The review must take place in the HR office and the Employee cannot remove or alter any file materials. On request, the Employee will be provided with a copy of any document in the file.
- 4. In the event that the Employee disagrees with anything contained in his Employee's HR file, he can add a separate written and dated objection identifying the document and specifying his reasons for his objection. The Employee must also send a copy of his objection to the Union President.
- (Original dated November 27, 1981)
- (Revised May 7, 1984 signed by J. Gravestock and D. Skoyen)
- (Reprinted and signed November 29, 1994 by A. Kazuta and B. Ringrose)
- (Reprinted and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

#### WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

R3-82

# LETTER OF AGREEMENT between WESTSHORE TERMINALS LTD. and INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

#### SUBJECT: BIRTHDAY HOLIDAY FOR NON-PERMANENT EMPLOYEES

Further to the meeting of the parties on December 2, 1982, this letter will clarify the agreement with regard to the above subject.

- A. A non-permanent Employee dispatched for work at Westshore Terminals will notify the Company prior to commencing work of a situation where the Employee's birthday will occur during the dispatched work assignment.
- B. Birthday Holiday eligibility and entitlement for non-permanent Employees who are on dispatch will be in accordance with Article 8 (Holidays) of the Collective Agreement.
- C. A non-permanent Employee on dispatch who, during the dispatch, qualifies for benefits as set out in Article 8 of the Collective Agreement, and who:
  - (a) at the commencement of the work assignment failed to notify the Company of a birthday holiday occurring during the work assignment, and/or
  - (b) <u>failed</u> to make acceptable arrangements with the Company for the observance of the birthday holiday during the work assignment, will be paid out for any benefit owing, in accordance with the provisions of Article 8 (Holidays) as contained in the Collective Agreement on completion of the dispatch.
  - (Original signed by D. Skoyen and R. Elke December 17, 1982)
  - (Revised and signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

#### INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### SUBJECT: REFUSAL TO WORK FOR HEALTH AND SAFETY REASONS

- (a) Any on-site disagreements with regard to the interpretation of the Collective Agreement will be resolved by the President of Local 502, or his designee, and the Westshore Terminals <u>Director</u> of Human Resources, or his designee.
- (b) In no event, other than for health and safety reasons, will Employees refuse to continue to work as directed by their supervisor.
- (c) In the event of disputes concerning matters of safety, Article 16, Section 16.03 will be observed with regard to continuation of work, pending a resolution on the issue.
- (Original letter dated and signed by D. Skoyen and R. Elke, January 26, 1983)
- (Reprinted and signed November 29, 1994 by A. Kazuta and B. Ringrose)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

**INTERNATIONAL LONGSHORE AND** 

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

R4-82

August 9,1984

# LETTER OF AGREEMENT between WESTSHORE TERMINALS LTD. and INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

#### SUBJECT: W.S.T. GANGPLANKS

The following is to be observed when there is an off-hour requirement at the Berths to affix or remove Westshore Terminals ship gangplanks:

- (a) A qualified crane operator or crane rated operator on shift will perform the above without replacement and this task will be considered to be a minor job in nature.
- (b) The amount of time required to perform the above job will be kept to a minimum.
- (c) Upon completion of the work involved the Employee will return to his normal shift assignment.
  - (Original letter signed by R. Elke, August 9, 1984)
  - (Reprinted and signed November 29, 1994 by A. Kazuta and B. Ringrose)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

#### INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### RE: BERTH #1 AND BERTH #2 - SAFETY BOAT

A regular working safety boat is to be located in the Seaspan Tug Basin for the purposes of providing emergency response capability for Berth #1 and Berth #2 in the event of a Man In The Water occurrence.

This boat will be on station at all times available for immediate operation by a trained operator who is comfortable with the operation of the boat or boats on station. All regular A-E crew operators, shift operators and shift maintenance personnel will be offered training in the operation of the boat or boats that will be utilized.

The regular working safety boat will comprise of a suitable motorized work boat. During periods of regular or extraordinary maintenance when the regular safety boat is not available for operation, a suitable substitute boat will be provided at the Seaspan Tug Basin for immediate operation.

On Berth #1 and Berth #2 when men are called upon to work in areas outside of normal handrail protected areas over the water the safety boat should be in the water and available at the berth. The safety boat operator will at all times man the boat when work is in progress. In addition, this provision shall apply where the man-lift is being used at either berth in a position where the occupants may fall into the water.

- (Original signed September 7, 1984 by G. Mann, G. Scott, B. Ringrose, R. Elke, P. Killeen and E. Sunstrum)
- (Revised and signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS

David Honsberger Director, Human Resources

**INTERNATIONAL LONGSHORE AND** 

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### SUBJECT: DISPATCH FEES

This is to confirm the acknowledgement and understanding between Westshore Terminals Ltd., and the International Longshore and Warehouse Union, Local 502, concerning Dispatch Fees.

A dispatch fee will be paid to the Union on a monthly basis for services rendered by the Union for dispatching of necessary Union personnel to Westshore Terminals Ltd. The amount of the dispatch fee will be equivalent to the monthly salary for a regular Employee as set out in the Collective Agreement.

The following criteria will be observed with regard to the above payments:

- (a) Payments are due within ten (10) days of the end of the month to which the payment applies.
- (b) Payments will be made only for those months that Westshore utilizes the Union Hall dispatch services.
- (c) The practice of a mutually agreed to list of persons acceptable for dispatches be maintained.
- (Original signed by K. Birmingham and R. Elke, dated November 1, 1984)
- (Reprinted and signed November 29, 1994 by A. Kazuta and B. Ringrose)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### SUBJECT: POLLUTION CONTROL/EMERGENCY CREW

The following outlines the understanding reached February 13, 1985 between the Company and the Union concerning the implementation of the automatic spray pole system used for pollution control:

- 1. All pollution control work, including manual and/or automatic spray pole system, will be performed by members of Local 502 under the supervision of a Local 514 Foreman.
- 2. The input of instructions that are necessary to sequence and activate the spray pole system will be performed by members of Local 502 under the supervision of a Local 514 Foreman.
- 3. Once the automatic spray pole system is activated, 502 members will be required to report to the Foreman any malfunctions or problems with the system.
- 4. Prior to implementation of the system, it will be operated as required for the purposes of testing and training. During this period operational procedures will be finalized and form part of the training program.
- 5. All operations Employees will be trained to operate the system. Initially, and prior to implementation of the system, a minimum of two (2) operations Employees per crew will be trained. Thereafter, these trained Employees will be used to train the balance of the Employees on their respective crews based on a predetermined training schedule.
- 6. Maintenance personnel will be provided the necessary information and training to ensure the system is properly maintained.
- 7. Standby for pollution control shall consist of five (5) Operators and a minimum of four (4) Operators shall be called in to perform pollution control where such functions are required by the Company. It is understood and agreed that a First Aid Employee and two (2) Shift Maintenance Employees and the SES operator shall also form part of the pollution control crew. It is further agreed, if the Operations crews are increased above the present complement of Employees, that a minimum of five (5) Operators shall be called in to perform pollution control where such functions are required by the Company.

- 8. Where the Company requires an Emergency Crew to operate belt lines during inclement weather or during periods of Plant shutdown, such crew shall be comprised of 2 Operators, 2 Shift Maintenance and 1 First Aid on all shifts. Employees required for such work shall be dispatched in accordance with current practice (December 25, 1996) and such Employees shall be required to report at the start of the designated shifts. The appropriate shift rate shall apply to such crew.
- (Original letter signed by P. Lanosky and K. Birmingham February 15, 1985.)
- (Revised September 29, 1993, signed by P. Idema and B. Ringrose.)
- (Amended and signed November 29, 1994 by A. Kazuta and B. Ringrose)
- (Amended and signed February 4, 1998 by Alan Kazuta and Brian Ringrose)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

#### WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

#### INTERNATIONAL LONGSHORE AND

#### WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### R1-85

#### SUBJECT: SEVERANCE PAY POLICY STATEMENT

In the event of plant closure or workforce reduction, any amounts of severance pay due to individual workers will be allocated by mutual agreement to the first (1) week following termination. All other terms and conditions of the existing Severance Pay provisions of the Agreement will apply.

- (Original signed by P. Lanosky and K. Birmingham, April 3, 1987)
- (Reprinted and signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

#### WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

#### INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### SUBJECT: BELT SPLICING

- 1. For the purposes of this Agreement "belt splicing" is defined as vulcanized splicing of textile and steel cable conveyor belts and/or mechanical splicing of textile conveyor belts.
- 2. If belt splicing contractors are used by the Company, then for each employee of the contractor that is assigned to the belt splicing job, the Company must call out an Employee from the overtime board in accordance with section 3 below.
- 3. The procedure for overtime callouts for belt splicing is as follows:
  - (a) An Employee that has a belt splicing rating, and is available for overtime (i.e. his pin is turned), will be given preference for the callout.
  - (b) If there is no Employee with a belt splicing rating that is available for overtime, then an Employee who has a belt splicing rating, and who is working on the shift when the belt spicing contractors are on site, will be assigned to the belt splicing work. The vacancy on the regular maintenance crew created by this assignment will be filled by calling out a mechanic or welder from the overtime board.
  - (c) If no Employee with a belt splicing rating is available, then a mechanic or welder without a belt splicing rating will be called out from the overtime board.
  - (d) If no Employees are available for overtime work, then a regular mechanic or welder, working on the shift when the belt splicing contractors are on site, will be assigned to the belt splicing work. The vacancy on the regular maintenance crew created by this assignment will be filled by a qualified tradesman from the Union hiring hall.
- 4. An Employee called out on overtime or assigned to the belt splicing job, in accordance with section 3 above, will assist in the work necessary to prepare the belt for splicing.
- 5. Once the belt has been prepared for the splice, then the Employee on overtime or assigned to the belt splicing job, in accordance with section 3 above, will work with the belt splicing contractor to prepare the splice, carry out the vulcanized or mechanical splice and inspect the splice.
- 6. On the graveyard shift the Employee called out on overtime, in accordance with section 3 above, will remain on the job after the splice is completed in order to get the belt back into service. Once the belt is back in service or the shift is over, whichever first occurs, then the Employee called out on overtime will be released.
- 7. On the day shift or afternoon shift the Employee called out on overtime or assigned to the belt splicing work, in accordance with section 3 above, may leave the site once the belt splicing or the shift is completed, whichever shall first occur.
- 8. On the day shift or afternoon shift the Company may request the Employee called out on overtime or assigned to the belt splicing work, in accordance with section 3 above, to remain on the job after the belt splicing contractors have left the site to assist in getting the belt back in service. The Employee can elect to stay, however, the Employee is not required to stay once the belt splicing is completed.

## BELT SPLICING (continued)

R2-87

- 9. If a belt splicing assignment requires more than one (1) splice, then the Employee assigned to belt splicing shall remain on the job until the last splice is completed or the shift is completed, whichever shall occur first.
- 10. The spirit and intent of the Agreement is that when belt splicing on day shift or on afternoon shift is completed the Employees on overtime or assigned to the belt splicing work, in accordance with section 3 above, will be released.
  - (Original signed by P. Lanosky and K. Birmingham May 11, 1987)
  - (Amended and signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

#### R4-88

# LETTER OF AGREEMENT between WESTSHORE TERMINALS LTD. and **INTERNATIONAL LONGSHORE AND** WAREHOUSE UNION, LOCAL 502

Mr. K. Birmingham, President, I.L.W.U. Local 502, 71 - 10th Street, New Westminster, B.C.

Dear Kent:

### SUBJECT: ELECTRICIANS

This is further to and will confirm our recent discussions during Collective Bargaining with regard to specific functions of Electricians as follows:

1. Regular Electricians will continue to perform all regular maintenance, troubleshooting, repair of electrical equipment and minor installation of new components on existing bulk equipment used in the operation of the plant as per past practice.

In addition, as new electrical technology, used in the operation and/or maintenance of bulk handling equipment, is introduced, regular Electricians will troubleshoot, repair and maintain electrical and electronic equipment associated with such technology. This provision will not prevent the Company from continuing to utilize external technical assistance where such expertise is required. Where such technical assistance is required one (1) Westshore Electrician from the crew will be assigned to work with the Technical Assistant(s). However, where more than one (1) Technical Assistant is required and the work is being performed in different locations, more than one (1) Electrician shall accompany the Technical Assistants.

It is understood that a regular Electrician's functions shall not include management information systems and/or computer based data systems utilized in the administration of the Company.

- 2. The Parties agree that the above conditions are subject to all other provisions of the Collective Agreement.
- 3. The minimum manning requirement for day Electricians shall be one (1) less than the crew size scheduled to work.
  - (Original signed by P. Lanosky and K. Birmingham April 25, 1988) ٠
  - (Amended November 29, 1994 and signed by A. Kazuta and B. Ringrose)
  - (Amended and signed February 4, 1998 by Alan Kazuta and Brian Ringrose)
  - (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

### WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

**Rick Hurtubise** 

President, Local 502

### **SUBJECT: MANNING**

This letter will confirm the understanding and agreement between the Parties hereto with respect to the following items related to Article 24 (Manning) of the Collective Agreement.

#### Lubeman

Unless otherwise required by the Company, it shall not be necessary to replace this Employee for absences other than Vacation or L.T.D. However, should the position in future be expanded, whereby an additional regular Lubeman is hired by the Company, then the minimum manning for this position shall be one (1) less than the crew size in accordance with their schedule.

- (Original signed by P. Lanosky and K. Birmingham -April 25, 1988)
- (Revised and signed 23rd day of February, 1995)
- (Amended and signed October 19, 2000 by Alan Kazuta and Bring Ringrose
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

R5-88

### SUBJECT: ORIENTATION/EVALUATION AND TRAINING PROCEDURES

This document confirms the agreement and understanding between the Parties concerning orientation and evaluation procedures for individuals who are submitted by the Union for inclusion on the Spare List, and for new regular employees.

### <u>A</u> <u>Orientation</u>

- <u>The Union and the Company shall meet monthly to review composition of the Spare List and to</u> <u>ensure that the orientation/evaluation procedures are being implemented in a reasonable and</u> <u>consistent manner. The Union may, at the monthly Spare List meeting, submit to the Company</u> <u>names to be considered for inclusion on the Spare List.</u>
- Prior to performing any work on the Westshore site an individual who has been nominated by the Union for inclusion on the Spare List or put forward as a new regular employee and accepted by the Company for evaluation will participate in a two (2) day basic orientation program, that currently includes the following items:
  - Westshore New Employee Orientation E-course
  - Westshore Rules of Conduct and Discipline Policy
  - Westshore Whistle Blower Policy
  - Westshore Non-Smoking Policy
  - WTLP PPE Policy
  - Workplace Violence Training and policy review
  - <u>Hazard Awareness Training</u>
  - <u>Safe & Proper Use of Buoyancy Devices</u>
  - <u>Control of Hazardous Energy Principles</u>
  - Hot Coal Management & CO Monitor Training
  - Incipient Fire Response Training
  - APR Selection, Use, and Care Training and Fit Test
  - <u>WHMIS2015</u>
  - <u>(Operations) Confined Space Attendant & Confined Space Monitor Training</u>
  - <u>(Operations) Escape Hood (Drager PARAT) Training</u>
  - Program Review
    - <u>Westshore Work Refusal Program</u>
    - <u>Westshore Incident Reporting & Investigation Program</u>
    - <u>Westshore Emergency Contingency Program</u>

Amendments to the orientation content will be reviewed with a representative of the Union prior to implementation.

- 3. <u>The individual will receive eight (8) hours pay at the straight time hourly rate for each day of this basic orientation.</u>
- 4. Individuals must undergo a pre-employment medical. Currently in operations, individuals will attend this appointment on a day separate from other aspects of the orientation and receive four hours pay for so doing. Individuals in maintenance will attend the medical on the same day as they perform their test as referred to in 2 (b).

# <u>B</u> <u>Training /Evaluation</u>

# 1. Operations

In the Operations Department the individuals dispatched for training and initial evaluation will be considered as outside the regular crew but not considered an "extra" as defined in Black Bool < Letter 2 - 94.

Individuals being trained will not be permitted to book off during their training/evaluation period, except for emergency reasons acceptable to the Company.

- a) <u>Each person requiring training/evaluation will receive up to a maximum of four (4)</u> <u>continuous weeks training during the day and/or afternoon shifts, in addition to the twoday orientation.</u>
- b) Employees to be trained will be jointly agreed to by the Union and the Company, including the amount of training required for each individual currently on the Spare List based on the previous training received by that individual at Westshore.
- c) The training period, for all new employees shall be fora minimum four (4) weel < period and during this period training shall be provided based on the availability of each of the pieces of equipment, and will be comprised of 24 shifts training as follows:
  - I. <u>Dumper</u>
  - II. <u>Stackers</u>
  - III. <u>Shiploaders</u>
- d) <u>Following the above mentioned training, any remaining shifts during the training period</u> <u>shall be utilized where extra training is required on specific equipment.</u>
- e) <u>Trainees will be trained by Regular and experienced Temporary Employees who are capable</u> of operating the equipment. Trainees can be expected to work on more than one (1) piece of equipment during a shift based on production needs.

The Trainer position will become part of the "job pick". The Trainer can be expected to train on more than one (1) piece of equipment during a shift, based on production needs. Where there

### R1-89

is no other training to be performed Trainers will not be required to discharge their duties as they would normally during shutdowns.

- f) <u>Trainees will be evaluated by a Foreman with input from the Regular Employee who</u> <u>provided the training.</u>
- g) Upon successful completion of the four (4) week training/evaluation period, the individuals will be placed on the "Evaluation" portion of the Spare List and assessed for a period of time up to thirty (30) shifts on regular dispatches, or be placed in the probationary period for regular workforce.
- h) In the event a candidate is not accepted for employment, the Company shall advise the Union, in writing, the reason for such non-acceptance.

## 2. Maintenance

Individuals being trained will not be permitted to book off during their training/ evaluation period, except for emergency reasons acceptable to the Company.

- a) <u>All new candidates will possess a B.C. or Inter-Provincial Trades Qualification Certificate.</u> <u>The Company will, however, give consideration to existing members and non-members of</u> <u>the I.L.W.U., Local 502, as of December 5th, 1996, who have acquired the equivalent</u> <u>practical skills and experience required as a Mechanic or Welder. All Regular Employees</u> <u>who leave the Company for non-disciplinary reasons will, upon application, be</u> <u>automatically included on the Spare List for the applicable trade categories.</u>
- b) <u>All candidates for employment at Westshore will be interviewed and required to submit a</u> resume, perform a written skills test and provide copies of all relevant trade certificates. The <u>Company will forward a copy of the resume to a representative of the Union.</u>
- c) <u>All potential Regular workforce and Spare List Mechanics, Electricians, and Welders and Technicians will be dispatched for two (2) shifts for Orientation as set out in A(2) foregoing.</u> <u>In addition, the employees will receive two further days of orientation that may occur consecutively to the initial two days, and if not later, comprised of Fall Protection Certification and Confined Space Entry Certification. This will normally occur within 30 days of the initial orientation. A delay beyond this will be reviewed with a representative of the Union.</u>
- d) Subsequent to the above, a candidate for employment at Westshore will be dispatched, as required, for a six (6) consecutive shift evaluation period and during this period the candidate will be considered as an "extra" as defined in Black Book Letter 2 - 94. When working with other Employees, the other Employees will be Regular Employees or experienced Temporary Employees.
- e) In conjunction with the above, the suitability for employment of each candidate will be <u>determined.</u>

- f) In the event a candidate is not accepted for employment, the Company shall advise the Union, In writing, the reason for such non-acceptance.
- g) Upon successful completion of the training/evaluation period the individuals will be placed on the "Evaluation" portion of the Spare List and assessed for a period of time up to twentyfour (24) shifts on regular dispatches, or be placed in the probationary period for regular employees.
- (Original signed by P. Lanosky and K. Birmingham August 10, 1989)
- (Revised and signed November 29, 1994 by P. Idema & B. Ringrose)
- (Revised and signed December 5, 1996 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2012 by David Honsberger and Tom Doran)

Revised and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### **SUBJECT: SHIFT OPERATORS**

This letter confirms the Agreement reached between the Parties to improve the efficiency/productivity at the Terminal by redeploying the Day Operators as Shift Operators, increasing the number of Employees who are allocated to the Shift Operator classification, and training additional Employees to operate mobile equipment including the crane.

Unless otherwise agreed to in this document the provisions of the existing Collective Agreement and Black Book will apply.

#### Terms and Conditions

- 1. The three (3) Day Operator positions at Westshore Terminals Ltd. on the date that this agreement is signed will be reclassified as Shift Operator positions.
- 2. The Company agrees to increase the number of Shift Operator positions to five (5). The Company will fill the two (2) additional Shift Operator positions through the job posting system. Two (2) Employees will be hired from the Union Hall to fill the resulting vacancies created by regular Employees who are successful on the job posting.
- 3. The five (5) Shift Operators will follow the MA ME schedule.

Seniority will be the criteria used to allow the Shift Operators to choose the crew to which they will be attached.

- 4. The Shift Operators will perform various duties for the Operations and/or Maintenance departments. These duties are not limited to the following:
  - operation of the crane, front-end loader, manlift, grader, forklift, and other mobile equipment as required by the Operations and/or Maintenance departments,
  - bulldozing and truck loading,
  - cleanup, boatman, washup and pollution control work, and
  - other duties that were performed by the Day Operators prior to this agreement.
- 5. The Company agrees to train additional Employees to operate the crane so that each operations crew has, in addition to the Shift Operator, a qualified Crane Operator as part of the regular crew.
- 6. If a Shift Operator is absent on the afternoon shift then he shall be replaced. The number of Shift Operators working on the day shift will be maintained at not less than one (I) below the crew size (two Shift Operators) established by this agreement.
  - (a) If a Shift Operator is absent from work for 1, 2, or 3 days, and if the Company is required to replace the absent Shift Operator, then he shall be replaced by calling out a qualified Employee with a crane rating from the overtime board.
  - (b) If a Shift Operator is absent from work for more than three (3) days, subject to (c) below, then he shall be replaced by transferring a qualified Employee with a crane rating from the corresponding operations crew and replacing the resulting vacancy with a qualified individual from the Union Hiring Hall.

If there is no qualified Employee on the corresponding operations crew with a crane rating then the absent Shift Operator will be replaced by an Employee from the crew (by mutual agreement) and the resulting vacancy with a qualified individual from the Union Hiring Hall. Under these conditions if crane work is required on the shift then a qualified Crane Operator will be called out on overtime to perform the crane work.

- (c) If a Shift Operator is on vacation or absent for two (2) scheduled work weeks or more, then he shall be replaced by the posting rules.
- 7. If a Shift Operator has been directed to bulldoze coal for valid production reasons then he will be required to continue bulldozing at shift change time (Hot Change) until his replacement on the next shift arrives at the dozer and is able to commence work. The Union and Company pledge that this provision will not be abused.
- 8. The Union and the Company agree that the letter in the Black Book concerning crane use on site (reference #6 88) is deleted from the Black Book.
- 9. If a Shift Operator is required for a non-scheduled shift or is required as an additional operator on a scheduled shift, the Shift Operator will only perform work associated with the call out and the order of dispatch will be:
  - (i) Shift Operator available on the overtime board (low hours).
  - (ii) Shift Operator transferred from the corresponding crew.
  - (iii) Shift Operator on the overtime board (low hours) who has not made himself available.
  - (iv) Transfer (low hours) from another scheduled shift provided that no double shifts is created by the transfer.
  - (v) Outside crane upon notification to the Union, prior to the shift required.
- 10. This agreement will become effective on signing and shall remain in effect until changed by mutual agreement of the Parties hereto.
  - (Original signed by P. Idema and J. Hoskins May 16, 1990)
  - (Revised and signed October 19, 2000 by Alan Kazuta and Brian Ringrose)
  - (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### **SUBJECT: JOB POSTINGS**

#### A. TEMPORARY VACANCIES

#### 1. **Application/Replacements**

The requirement for Job Posting shall apply where the Company has been given a minimum of twenty-one (21) calendar days or more notice and the Employee's intended absence is for two (2) scheduled work weeks or more. If there are no qualified applicants for a job posting the provisions of section (a) and (b) below will apply.

Where the Company receives less than twenty-one (21) calendar days notice of an Employee's intended absence the vacancy will be filled as follows:

- (a) If the absence is for less than two (2) scheduled work weeks the vacancy shall be filled from the Operations or Maintenance corresponding crews and a qualified replacement for that crew shall be obtained from the Union Hall.
- (b) If the vacancy is for two (2) scheduled work weeks or more it shall be posted and filled from the Operations or Maintenance corresponding crews working until the posting is filled and a qualified replacement for that crew shall be obtained from the Union Hall. At such time as the job posting is filled a qualified individual shall be obtained from the Union Hall to replace the Employee who is accepted on the job posting.
- (c) If an Employee advises the Company that he will be taking banked days or days in lieu with his vacation, such additional days shall be included in the term of the job posting. If the Employee advises that such days are to be taken following the twenty-one (21) calendar days notice period, and such days to be taken are mutually agreed, the additional days shall be filled, if required by the manning rules, as follows:

| (i)  | Banked days  | - | Per Article 7.07 (f) |
|------|--------------|---|----------------------|
| (ii) | Days in lieu | - | Per Article 20.01    |

The only exceptions to the above shall be the categories of Shift Operator, Warehouseman/Tool Crib Attendant, or Shift Maintenance who shall be replaced, if required, by the manning rules, from the crews and the corresponding crew vacancy shall be filled per Article 7.07 (f) for banked days and Article 20.01 for days in lieu. Replacements for Employees accepting a job posting, except as otherwise provided herein, shall come from the Union Hall.

### 2. Job Posting Period

Where required and within five (5) calendar days of notice being given (excluding the day of notice) the Company shall post positions for a period of seven (7) consecutive days.

### 3. *Method of Posting*

All Job Postings shall be posted on the Notice Board in the Employees' lunch room and a copy shall be forwarded to the Union.

### 4. <u>Preferred Categories</u>

Preferred categories and functions to which job postings will apply, with regard to regular Employees only, shall be in accordance with past practice or as maybe mutually agreed to by the Department Manager and the Union President.

### 5. Job Posting Restrictions

- (a) An Employee in a preferred category shall not be eligible to apply for Job Postings.
- (b) Where an Employee's acceptance of a job posting would result in the established crew size being reduced below the maximum allowed on vacation at any one time, such Employee shall not be eligible for the job posting.
- (c) An Employee will not be considered for a job posting if not qualified to perform the functions of the posted position.

### 6. **Compensation for Temporary Job Postings and Transfers**

- (a) The Employee will follow his base schedule up to and including the day immediately before the first day of the temporary schedule then change crews and begin to follow the temporary schedule. If, as a result of the crew change, the Employee works more days during the payroll week of the change than he was scheduled to work on his base schedule then the Employee will receive overtime for those additional days.
- (b) If the end of the job posting period coincides with the last day of work on the temporary schedule the Employee will continue on the temporary schedule until the last scheduled day off following the posting period. The Employee will then revert to his base schedule.
- (c) If the end of the job posting period does not coincide with the last day of work on the temporary schedule then the Employee will revert to his base schedule the day after the last day of the posting. If, as a result, the Employee works more days during the payroll week of change than he was scheduled to work on the temporary schedule then the Employee will receive overtime for those additional days.

### B. PERMANENT VACANCIES

### 1. Job Posting Period

Where a permanent vacancy occurs the Company shall post such vacancy in accordance with A.3 above, within five (5) calendar days, for a period of fourteen (14) consecutive days.

### 2. Compensation for Permanent Job Postings

The Employee will follow his old schedule up to and including the day immediately before the first day of the new schedule then change crews and begin to follow the new schedule. If, as result of the crew change, the Employee works more days during the payroll week of the change than he was scheduled to work on his old schedule then the Employee will receive overtime for those additional days.

### C. LIMITATIONS

The procedures and conditions contained herein shall only apply to Job Postings and shall have no other application.

### D. EXCEPTIONS JOB POSTING PROCEDURE

It is understood and agreed that there shall be no exceptions to the above procedure unless approved by the Department Manager and the Union President

#### E. DEFINITIONS

The term "week" as referred to herein shall be defined as seven (7) consecutive days.

The term "payroll week of change" as referred to herein shall be defined as the payroll week during which the Employee works the first shift of his temporary schedule and/or the payroll week during which the Employee works the first shift after reverting to his base schedule at the end of the posting.

The term "scheduled work week" as referred to herein shall be defined as those four (4), five (5), or six (6) consecutive days of work bounded on both ends by scheduled days off (but excluding the scheduled days off) as set out in the work schedule.

- (Original signed by P. Lanosky and J. Hoskins July 13, 1990, modified July 16, 1992 and signed by K. Birmingham)
- (Modified and signed November 29th, 1994 by Alan Kazuta and Brian Ringrose)
- (Revised and signed October 19, 2000 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

July16, 1992

# LETTER OF AGREEMENT between WESTSHORE TERMINALS LTD. and INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

## SUBJECT: DIRECT DEPOSIT PAYROLL SYSTEM

This letter confirms the agreement between the Parties hereto with respect to the Direct Deposit Payroll System.

The Direct Deposit Payroll System is subject to the following conditions:

- 1. Pay Statements for each payroll period shall be distributed to the Employees participating in the program by the Employee's Foreman on their regular shift on payday or the Employee's next working day immediately following the payday.
- 2. Employees that participate in the Direct Deposit System will have their regular net pay deposited to their individual bank accounts on or before 1:00 p.m. Thursday of each payroll week.
- 3. All regular and temporary Employees shall be eligible to participate in the Direct Deposit Payroll System.
- 4. Eligible Employees shall be given the option to participate in the Direct Deposit Payroll System once per annum. Such Employees must advise the Company of their intent to participate or withdraw from the program prior to June 1st of each calendar year for the following twelve (12) month period (June 1 to May 31).
- 5. New Employees and temporary Employees approved for the "spareboard" shall be entitled to participate in the Direct Deposit Payroll System immediately or on the next annual election date, in accordance with paragraph 2 above.
- 6. Employees that elect to participate in the Direct Deposit Payroll System shall continue to participate in the program from year to year unless the Company is advised in writing, in accordance with paragraph 2 above, of their intent to withdraw from the program.
- 7. Employees participating in the Direct Deposit Payroll System must advise the Company, in writing, a minimum of one (1) week in advance, if they intend to utilize Banked or Lieu Days as cash payments. If the required notice is not made any adjustment for such days will be made in the deposit made for the subsequent payroll period.
- 8. It shall be the responsibility of each Employee that participates in the program to provide the Company with appropriate information necessary to implement the Direct Payroll Deposit System. The Company will provide Employees with an appropriate form to complete prior to implementation or election to participate in the program.
- 9. With regard to vacation pay, Employees that participate in the program shall continue to have their normal direct deposit made during the week of each week of vacation taken, but may request an advance payment of their vacation pay, provided such request is submitted in writing, a minimum of one (1) week in advance of the commencement of the vacation period.

- 10. It is understood and agreed that the Company may elect to discontinue the Direct Deposit Payroll System at any time that participation in the program is less than fifty (50) percent of the total number of regular Employees employed by the Company at any one (1) time.
  - (Original signed by K. Birmingham July 16, 1992)
  - (Modified and signed November 29, 1994 and signed by A. Kazuta and B. Ringrose)
  - (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

## WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

## INTERNATIONAL LONGSHORE AND

## WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### SUBJECT: REHIRED EMPLOYEES

This letter will confirm the understanding and agreement between the Parties hereto with respect to the following:

Those Employees who were laid off March 22, 1993, and subsequently rehired, shall be given unbroken service for pensions, vacations and benefit programs, including salary continuance.

- (Original signed by P. Idema and B. Ringrose September 29, 1993).
- (Revised and signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### **RE: PAYMENT FOR LOCAL 502 BARGAINING COMMITTEE MEMBERS**

This will confirm the Agreement between the Parties with respect to the payment of Bargaining Committee Members effective from the date of the first meeting as follows:

- 1. The five (5) members of the Local 502 negotiating committee that are regular Employees of Westshore Terminals Ltd. shall be paid their regular earnings during the negotiations.
- 2. The Committee Members shall be released from their regular crew assignments to attend scheduled Collective Bargaining meetings with the Company and with the Conciliation Officer. In addition, for any day on which a regular Employee attends Collective Bargaining, such Employee will not be required to report to work within the twenty-four (24) hour period beginning at 0800 hours on that day. It is further agreed that regular Employees who are Committee Members and are scheduled to work the shift from 0001 hours to 0800 hours on a day scheduled for Collective Bargaining will not be required to work that shift.

An Employee, as a member of the bargaining committee, shall not be required to attend collective bargaining and work more days in any week greater than that required by his regular shift schedule. At the conclusion of negotiations if the committee members have worked more days, as a result of their attendance at negotiations, than their regular scheduled number of days, such extra days may be paid out at straight time or taken as lieu days and a replacement, if required, shall be supplied by the Union from the Union Hall.

- 3. If it becomes necessary to replace an Employee, pursuant to the provisions of Article 24 of the Collective Agreement, who is absent to attend Collective Bargaining, such replacement shall be supplied by the Union Hall.
- 4. It is understood and agreed that the above conditions shall apply for this set of negotiations only and that similar or like conditions for future negotiations will depend on the circumstances which prevail at that time.
- (Signed January 26th, 1994 by A. Kazuta and B. Ringrose)
- (Modified and signed January 15, 1997 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed January 29, 2016 by David Honsberger and Tim Farrell)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### SUBJECT: UTILIZATION OF EXTRAS (HALL DISPATCHED) TEMPORARY EMPLOYEES

The following outlines the agreement and understanding between the Company and the Union concerning the use of extra temporary personnel in both the Operations and Maintenance Departments.

#### A. Operations

- <u>1.</u> Spareboard personnel may be employed in the following job functions in the Operations Department where the work required is in excess of three (3) consecutive days:
  - 1. Extra Dozer Operators for the stockpiling, reclaiming or grooming of stockpile areas.
  - 2. Extra Dozer and Mobile Equipment Operators required for special pile relocations, including relocation of residues requiring trucking operations.
  - 3. Extra Cleanup Operators for general yard cleanup, including ditches, boneyard, landscaping and brush/weed removal.
  - 4. Extra Cleanup Operators for general cleanup of the Dry facilities.
  - 5. Extra Cleanup Operators for wash down of roads and/or production equipment.
  - 6. Extra Operators, as referred to above, may be requested for a maximum of (6) days. Such Operators will follow the assigned crew schedule referred to in the dispatch request and cannot be transferred to another crew schedule during this period.

#### **B.** Maintenance

- 1. Spareboard personnel may be employed in the Maintenance Department, where the work required is in excess of three (3) consecutive days and shall follow the assigned crew schedule, as follows:
  - a) Extra Mechanics/Welders and/or Electricians to perform work related to Regular Maintenance duties and functions, including special projects and planned shutdowns.
  - b) The following shall apply to dispatches for Safety Work Orders and/or P.M.'s only:
    - dispatches for Safety Work Orders or P.M.'s involving four (4) to six (6) days will provide the specific area or specific piece of equipment and will not be combined or include any other maintenance work, and
    - dispatches for posted positions involving more than six (6) days may combine Safety Work Orders and P.M.'s on a site basis but will not include any other maintenance work.
- 2. The utilization of extras, in accordance with this provision (Section B. Maintenance), shall be subject to mutual agreement of the Company and the Union.
- 3. In addition, when extra Mechanics/Welders, and/or Electricians are required for assignments of over six (6) working days, the Company will first offer regular employees the work opportunities through the job posting system The job posting will state the expected duration of the job.
- 4. If a job posting is not filled, as noted above, the crew regular employee, and then the adjacent crew regular employees will be offered the position(s) respectively.

- 5. For job postings over two (2) working weeks, every third week the Company will have the option to transfer one (1) or more regular employees to an adjacent crew. This will ensure that all personnel will maintain their appropriate shifts, but one (1) or more employees will be assigned to a different supervisor.
- 6. For non posted assignments, (i.e. six (6) days or less for Mechanics/Welders and/or Electricians, the related work will first be offered to the crew regular employees prior to the commencement of the first shift and then to the adjacent crew regular employees. This provision shall not apply to preferred categories.

### C General

- 1. The Company agrees not to unreasonably request extras, in accordance with the intent of the above provisions, and the Union agrees not to unreasonably withhold extras requested for legitimate reasons within the intent of this Agreement.
- 2. It is understood and agreed that when the Company is employing extra temporary personnel, dispatched from the Union Hall in accordance with section A and/or subsection B.1 of this letter, the manning provisions for the regular crews scheduled at that time, (in the Operations or Maintenance Department where the extra personnel are assigned) related to the down rules contained in Article 24 of the Collective Agreement shall not apply (i.e. the regular scheduled crews in the Department will be manned at the full normal complement).
- 3. The parties agree that where Temporary Employees are dispatched from the Union Hall, in accordance with the provisions herein, and such employees are absent for any reason, their replacements shall be dispatched from the Union Hall.
- 4. The Company agrees that application of this Letter of Agreement shall not be used to circumvent the employment of regular Employees.

#### D Term

This Letter of Agreement shall continue in effect during the term of the Collective Agreement, however, in the event of a layoff of personnel in the Operations or Maintenance Department, in accordance with the provisions of Article 15 of the Collective Agreement, the provisions contained herein related to the Department where the layoffs occur shall be suspended for the period of such layoff or such other agreement as may be mutually agreed to by the Parties.

- (Original signed November 29th, 1994 by Alan Kazuta and Brian Ringrose)
- (Amended and signed February 4, 1998 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### SUBJECT: ELECTRICAL PRINTMEN

This will confirm the Company's advice to the Union with respect to the employment of an additional Electrical Printman.

In addition, this Letter of Understanding outlines the agreement and understanding between the Company and the Union concerning the assignment, duties and/or functions and replacement requirements of the Electrical Printmen.

- 1. The two (2) regular Electrical Printmen shall be assigned to the M1 M5 schedules, or such other schedules as mutually agreed to by the Parties.
- 2. The duties and functions of the Electrical Printmen shall include:
  - (a) to update existing plant electrical prints, including corrections, modifications and/or alterations;
  - (b) to maintain and update documentation related to plant controller and ladder logic programs in both hard copy and electronic format;
  - (c) to perform field investigation work to determine the accuracy of prints associated with the above functions;
  - (d) to maintain all computer application data bases used in the documentation and record keeping of all electrical systems as outlined in (a) and (b) above;
  - (e) to assist in the engineering, documentation and commissioning, as required, for plant modifications;
  - (f) to perform other plant electrical maintenance work where an emergency situation occurs or the workload related to prints permits such other work, and
  - (g) to continue to perform other functions related to the Electrical Department.
- 3. Any requirement to replace a regular Electrician assigned to prints shall be in accordance with the provisions of Article 24, sub-section 24.02 (g).
- (Original signed November 29, 1994 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of December 2020 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

## SUBJECT: TECHNOLOGIST SYSTEMS

As part of the evolution of the electrical department, four technologist's positions will be introduced, reducing the Printmen role to one position. Of these four new positions, one will be filled by Mr. Milan Rebrec who completed the educational prerequisites while being sponsored to do so by Westshore, two will be filled by Mr. Butler and Grewal, subject to their making themselves available for required upgrade training, and completing the same in an acceptable time frame. Westshore will pay for these requisite courses and any time lost from work for attendance at such training. The final position will be filled by an outside hire who has the appropriate designation. A representative of the Union may participate in the selection process, but the final decision on the successful candidate will rest with the Company.

The Union will ensure there are adequate qualified replacements available from the hall and will work towards having at least three available at all times.

These positions will initially be assigned to work the MDD and MDE dayshift schedules, with two employees assigned to each rotation. This schedule may be amended as the role matures and as determined by the requirements of the Company. At minimum, one of these four positions must be in attendance on their scheduled shifts at all times, and therefore these employees must co-ordinate their vacation and other time off to achieve this requirement. There will not be a requirement to replace any absence of less than two weeks.

Absences of two weeks or more will be filled by qualified replacements from the hall. Extra manpower requirements may be filled using currently trained Printmen, (i.e. those trained prior to September 2015), provided they have the skill set required for the task.

If there are no suitable Printmen, current replacement rules will be used. In any circumstance, failing the availability of suitably trained and qualified employees, either for extra manning requirements or for vacancy replacements, outside contractors may be used to perform the work.

The Technologist position will assume the duties and responsibilities of the Printman, except for those duties outlined below, which will remain the responsibility of the lone remaining printman. In addition, the technologist's duties will include being the primary programmers of site PLC's, and they may assist in the engineering and commissioning, as required, of plant modifications.

These employees will not be eligible to apply for a job posting or be available to work outside of the technologist position unless they have the required designation or by mutual agreement.

With respect to the Printman position, the parties agree to meet to review the qualifications required for the role for future openings. This letter will become effective on the Monday following the electrical department's move to the new building.

#### R6-94

### Printmen:

Duties will include the following:

- Updating existing plant electrical prints, including corrections, modifications and/or alterations;
- Maintaining and updating documentation related to plant controller and ladder logic programs in electronic format;
- Performing field investigation work to determine the accuracy of prints as required;
- Maintaining all computer application data bases used in the documentation and record keeping of electrical systems;
- Assisting in documentation and commissioning as required;
- Performing other plant electrical maintenance work where an emergency situation occurs or the work load related to prints permits such other work; and,
- Performing other functions related to the Electrical Department.

Any requirement to replace a regular Electrician assigned to prints shall be in accordance with the provisions of Article 24, sub-section 24.02.

The printmen will initially be assigned to the MDB schedule.

### R6 - 94 APPENDIX "A"

| SYSTEM    | CATEGORY  | 1            | 2            | 3            |
|-----------|---|--------------|--------------|--------------|
| 1. Real-1 | ime Display, Management and Control System  |              |              |              |
| HMI Wor   | k Stations (used to operate the site)   |              |              |              |
|           | (i) Hardware  |              | $\checkmark$ |              |
|           | (ii) Operating System   |              | $\checkmark$ |              |
|           | (iii) Application Software (Utilities & Programs)                                     |              |              | $\checkmark$ |
|           | (iv) HMI Screens  |              |              |              |
|           | Maintenance and Minor Modifications   | $\checkmark$ |              |              |
|           | New Development   | $\checkmark$ |              |              |
| B.        | HMI Servers   |              |              |              |
|           | (i) Hardware  |              |              |              |
|           | Initial Troubleshooting   | $\checkmark$ |              |              |
|           | Maintenance & Development   |              | $\checkmark$ |              |
|           | (ii) Software and System Management   |              |              |              |
|           | Operating Systems, System Management and Application Software<br>(Database, Programs) |              | $\checkmark$ |              |
| C.        | PLC System Associated Wiring, and Field Operators PC's                                |              |              |              |
|           | (i) PLC 5/250   | $\checkmark$ |              |              |
|           | (ii) Field PLC's  | $\checkmark$ |              |              |
|           | (iii) Field PLC Data Highways   | $\checkmark$ |              |              |
|           | (iv) Field Operator Stations  | $\checkmark$ |              |              |
|           | (v) Control Logix Gateway   | $\checkmark$ |              |              |
| D.        | Industrial Ethernet (Industrial Domain)   | I            | I            | I            |
|           | (i) Industrial Routers  |              | √            |              |
|           | (ii) Sub Station Ethernet Switches  |              | $\checkmark$ |              |
|           | (iii) Sub Station Workstations  | $\checkmark$ |              |              |

### F Commercial Domain

| (i) Printman/Technologist Computer   |              | $\checkmark$ |              |
|--|--------------|--------------|--------------|
| (ii) Operations Foremen Computers  |              |              | √            |
| (iii) Administrative Computers   |              |              | √            |
| (iv) Maintenance Superintendents Computers   |              |              | √            |
| (v) Maintenance Managers Computers   |              |              | √            |
| (vi) Maintenance Foremen's Computers   |              |              | √            |
| (vii) Commercial Routers   |              |              | √            |
| F. DMZ Servers (secure are commercial and industrial domains)                      |              |              |              |
| (i) Industrial Network Security (WTLP-DC)  |              |              | $\checkmark$ |
| (ii) Rockwell Asset Center Server (WTLP – ASSET 1)                                 |              | $\checkmark$ |              |
| (iii) Rockwell Software Terminal Server (WTLP-RSTS)                                |              | $\checkmark$ |              |
| (iv) HMI Terminal Server (WTLP-WWTS)   |              | $\checkmark$ |              |
| (v) HMI Object Server (WTLP – WWTS)  |              | $\checkmark$ |              |
| (vi) HMI Database  |              | $\checkmark$ |              |
| (vii) HMI Application Server   |              | $\checkmark$ |              |
| (viii) HMI re Historian  |              | $\checkmark$ |              |
| (ix) Client Management Server  |              | $\checkmark$ |              |
| (x) Industrial Virtual Server  |              | $\checkmark$ |              |
| (xi) Industrial Backup Resource  |              | $\checkmark$ |              |
| (xii) Network Attached Storage System (WTLP-NAS01)                                 |              | $\checkmark$ |              |
| G. Industrial Ethernet Ring  |              | I            |              |
| (i) Ring Manager   |              | $\checkmark$ |              |
| (ii) Software and Systems Management   | $\checkmark$ |              |              |
| H. Control Tower (Site Automation Control)   |              | I            | I            |
| (i) Hardware (Computer and Thin Clients)   | $\checkmark$ |              |              |
| (ii) Cabling   | $\checkmark$ |              |              |
| (iii) HMI Screens  |              |              |              |
| Maintenance and modifications  | $\checkmark$ |              |              |
| New Development  | $\checkmark$ |              |              |
| anagement Information and Administration Systems                                   |              |              | -            |
| Servers, Work Stations, LAN (Local Area Network, Cabling and Associated Equipment) |              |              | $\checkmark$ |
| Applications and Data Transfers between Servers                                    |              | 1            |              |

 WTL Technologist
 WTL Technologist\*/CO\_OPERATION with MIAS\*
 MIAS GROUP\*(includes outside contractors retained by the Company) \*Primary Responsibilities

## SUBJECT: COMPUTERIZED SYSTEMS

The following is an overview of the present computerized systems utilized in the Company's operations. Confirming the involvement of WTL Electricians in the maintenance of such systems,

The present computerized on-site systems include a "Real- Time Display, Management, and Control System," and "Management Information and Administration Systems,"

In brief these systems involve the following components:

### 1. Real-Time Display, Management and Control System

#### A. HMI Work Stations (Used to operate site)

- (i) Hardware
- (ii) Operating System
- (iii) Application Software (Utilities & Programs)
- (iv) Screens
  - Maintenance and modifications
  - New Development (See subnotes A (iv))

### B. HMI Servers

- (i) Hardware
  - Initial Troubleshooting
  - Maintenance & Development
- (ii) Software and System Management
  - Operating Systems, System Management, and
  - Application Software (Database, Programs)

### C. PLC System, Associated Wiring, and Field Operators PC's

- (i) PLC 5/250
- (ii) Field PLC's
- (iii) Field PLC Data Highways
- (iv) Field Operator Stations
- (v) Control Logix Gateway
- (vi) Field PLC Workstations

### D. Industrial Ethernet (Industrial Domain)

- (i) Industrial Routers
- (ii) Sub Station Ethernet Switches
- (iii) Sub Station Workstations

### E. Commercial Domain

- (i) Printman Computers
- (ii) Operations Foreman Computers
- (iii) Administrative Computers
- (iv) Maintenance Superintendents Computers
- (v) Maintenance Managers Computers
- (vi) Maintenance Foremen Computers
- (vii) Commercial Routers

#### R6-94

### F. DMZ Servers (secure area between commercial and industrial domains)

- (i) Industrial Network Security (WTLP-DC)
- (ii) Rockwell Asset Center Server (WTLP-ASSET 1)
- (iii) Rockwell Software Terminal Server (WTLP- RSTS)
- (iv) HMI Terminal Server (WTLP-WWTS)
- (v) HMI Object Server (WTLP-WWTS)
- (vi) HMI Database
- (vii) HMI Application Server
- (viii) Client Management Servers
- (ix) Industrial Virtual Servers
- (x) Industrial Backup Resource

### G. Industrial Ethernet Ring

- (i) Ring Manager
- (ii) Ethernet Switches

### H. Automation Control Room

- (i) Hardware (Computers and Thin Clients)
- (ii) Cabling

### 2. Management Information and Administration Systems

- A. Servers, Work Stations, LAN (Local Area Network, Cabling & Associated Equipment)
- B. Applications and Data Transfers between Servers

With respect to the involvement of both WTL Electricians and the MIAS (Management Information and Administration Systems Group) in the maintenance of the above systems, the attached Appendix" A" shall establish the involvement of the parties.

The Parties recognize that the equipment and software referred to herein may be superseded by future product advancements and/or product substitutions as part of the regular upgrading of WTL computerized systems. In regards to upgrades and changes in the hardware and software, WLT will provide applicable training to WST required personnel in a reasonable time frame.

### Sub-note A (iv)

The Company will inform the Technologist of all new object requirements. It is understood that the Technologist will develop and/or deploy new objects, and it is further understood that the Technologist will fully co-operate with the Company to ensure that the MIAS group will be included in this development jointly. Any such objects created will be approved by the Company prior to implementation/deployment.

### Sub-note as per Section F

The parties to this letter will establish the level of access to PLC and HMI Software for the Electricians and Printmen, and Technologist. The Technologist in conjunction with the IT department will set individual 502 access to both the HMI and PLC software.

#### R6-94

Technologists will be responsible for configuration and updates to HMI and PLC software with the technical assistance of the IT department.

Employees accessing either system in order to make changes will be required to log in with an individual username and password. No personal login will be required for read only access. Technologists will be given direct read-only access to the data-base theory containing the asset center activity logs. This access will be provided at the Technologists normal work station.

- (Original signed by David Honsberger and Tim Farrell November 6, 2015)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

### WESTSHORE TERMINALS

David Honsberger Director, Human Resources

### INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

**Rick Hurtubise** 

Rick Hurtubise President, Local 502

## SUBJECT: UTILIZATION OF EXTRA TEMPORARY OPERATOR

The following outlines the agreement and understanding between the Company and the Union, without prejudice or precedent, concerning the Utilization of an extra temporary Operator.

- 1. The Operations crew(s) will be manned at the full normal complement prior to an extra Operator being ordered (i.e. down rules as contained in the Collective Agreement shall not apply).
- 2. The extra Operator will be assigned to a production shift (Schedule A E).
- 3. On dispatches of up to three (3) shifts, the extra Operator will be ordered from the overtime board. For dispatches of four (4) shifts or more, the extra Operator will be ordered from the Union Hall.
- 4. This Agreement will be in effect during the term of the Collective Agreement, unless either Party gives to the other Party thirty (30) days notice in writing of a desire to modify or terminate this Letter of Agreement.
- (Original signed February 4, 1995 by Alan Kazuta and Brian Ringrose)
- (Amended and signed February 23, 1995 by Alan Kazuta and Brian Ringrose)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

## SUBJECT: MAINTENANCE NIGHT SHIFT

This Letter of Agreement confirms the agreement between the parties with respect to implementation of a Maintenance Night Shift in accordance with the provisions of Article 6, section 6.03 - Normal Work Pattern, as follows:

- 1. The Company shall give the required notice under Article 6, subsection 6.03 (e) of the Collective Agreement, however, the Night Shift shall not be implemented prior to January 1st, 1998.
- 2. This will also confirm that the Company will post for the following additional positions on signing of this Letter of Agreement:

1 Mechanic 3 Electricians

3. In consideration of the terms contained in this Letter of Agreement, and on employment of the additional personnel referred to above, the Employees on designated shifts shall be as follows (Schedule II - MA - ME, Days and Afternoons only), and the present down rules contained in Article 24, subsection 24.01 (b) (i) shall continue to apply:

| MDA*  | Days Only        |            | 1 Mechanic     |
|-------|------------------|------------|----------------|
|       |                  |            | 1 Welder       |
|       |                  |            | 2 Electricians |
| MDB*  | Days Only        |            | 1 Mechanic     |
|       |                  |            | 1 Welder       |
|       |                  |            | 2 Electricians |
| M5    | Days Only        |            | 1 Warehouseman |
| MA-ME | Days, Afternoons | and Nights |                |
|       |                  | 4 crews    | 2 Mechanics    |
|       |                  |            | 2 Welders      |
|       |                  |            | 2 Electricians |
|       |                  |            | 1 Warehouseman |
|       |                  | 1 crew     | 3 Mechanics    |
|       |                  |            | 1 Welder       |
|       |                  |            | 2 Electricians |
|       |                  |            | 1 Warehouseman |

Schedule V attached

#### R1-97

#### MAINTENANCE NIGHT SHIFT (continued)

#### R1 - 97

At such time as it is deemed necessary to implement the Night Shift, in accordance with Point 1 above, Employees on the MA - ME schedule will be required to work the attached Schedule IV, Days, Afternoons, and Nights.

- 4. With regard to vacation replacements, on the MDA or MDB schedule that has two (2) Welders, the Company may, where required, replace one (1) of the Welders for vacation absences with a Mechanic. In addition, on the MA ME schedule, the Company may, where required, replace one (1) Mechanic for vacation absences on the crew comprised of three (3) Mechanics and one (1) Welder, with a Welder.
- 5. On implementation of the Night Shift schedule, the following provision shall be added to Section B. Maintenance (Point 7) of R 2 -94:

Where extras are required on the Day Shift the Company will make every effort to provide the Union with a minimum of seventy-two (72) hours notice and the available work will commence at the start of the night shift cycle and first be offered to the Regular Employees working such shift. Where extras are required on the Afternoon and/or Night Shift the corresponding crew provision shall not apply and the work will first be offered to the Regular Employees working such shift.

- 6. In addition to the above, the categories working the MDA MDB and <u>MDC</u> schedules shall be considered preferred categories.
- In accordance with the existing letter (BB R I 90) Shift Operators shall continue to work the MA ME (Schedule II) as contained in the Collective Agreement.
- 8. It is further agreed that if the Night Shift is implemented, and the Company decides to revert to Schedule II (MA ME Days and Afternoons only) as referred to in Point 3 of this Letter of Agreement, the Company shall provide a minimum of thirty (30) calendar days notice prior to instituting such schedule.
- 9. Subsequent to January I, 1998, in the event the Company determines there is a business requirement to modify the number of Mechanics and/or Welders employed on each shift, as referred to herein, the Company shall discuss the matter with the Union in accordance with the provisions of Article 3, subsection 3.03 (e) of the Collective Agreement.
  - (Original signed May 20, 1997 by Alan Kazuta and Brian Ringrose)
  - (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Revised and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

### SUBJECT: CLEANUP PERSONNEL

This Letter of Agreement replaces the former letters R 2 - 83 and R 3 - 90 and confirms the understanding and agreement between the Company and the Union with respect to Inside and Outside Cleanup Personnel.

#### A. Inside Cleanup Personnel

This will confirm the understanding and agreement between the Parties hereto with respect to cleanup personnel.

- 1. The prime function of the existing two (2) Inside Cleanup Personnel will be to maintain the Employees' lunchroom and dry facilities, the two (2) Employees will be assigned to the A and B Schedules however, the afternoon and night shifts contained in such schedules shall be worked as "day shifts".
- 2. It is further understood and agreed that when not required for cleanup duties in the Employee's lunchroom and dry facilities, one (1) or both of the Employees may be assigned to work with the regular outside clean up crew to perform regular cleanup crew duties, but not as replacements and only when both Employees are on duty.
- 3. In addition to the above, the Parties have agreed that First Aid personnel will do cleanup in the lunchroom on the afternoon and night shifts.
- 4. Inside Cleanup Personnel shall be replaced for absences as follows:
  - a) Annual Vacations of one (1) week, and days in lieu of less than one (1) week combined with one (1) week of vacation shall be replaced with a qualified replacement from the Union Hall, all absence, of two (2) weeks or more shall be replaced in accordance with Article 24, section 24.03 and R 2 90.
  - b) Banked Overtime Days and Book Offs a qualified replacement from the Union Hall.
  - c) Other Absences in accordance with the provisions of Article 20.
  - d) If both Employees are scheduled to work on the same day, and if one (1) of the Employees is absent then there shall be no requirement to replace the absent Employee for the first two (2) days of such absence. If one (1) Employee is scheduled to work and the Employee is absent then such Employee shall be replaced in accordance with (b) or (c) above.

### B. Outside Cleanup Personnel

- 1. The Employees in this position will perform the following functions which will be considered as part of their duties:
  - Washup/Cleanup
  - Pollution Control
  - Operating the front end loader, bobcat, water-trucks, as required for cleanup purposes.
- 2. It is also understood and agreed that, if the Company purchases additional equipment for cleanup purposes, then the Cleanup Operator will be trained and will operate such equipment.
- 3. Effective the date of signing of this Letter of Agreement the Company shall post for two (2) additional Outside Cleanup Personnel. On employment of the additional personnel, the four (4) Outside Cleanup Personnel shall be assigned to the C and E Schedules, however, the afternoon and night shifts contained in such schedules shall be worked as "day shifts".
- 4. It is understood and agreed that Outside Cleanup Personnel will not be required to replace or perform the duties of Inside Cleanup Personnel.
- 5. Outside Cleanup Personnel shall be replaced for absences as follows:
  - Annual Vacations of one (1) week, and days in lieu of less than one (1) week combined with one (1) week of vacation, shall be replaced with a qualified replacement from the Union Hall.
     All absences of two (2) weeks or more shall be replaced in accordance with Article 24, section 24.03 and R 2 90.
  - b) Banked Overtime Days and Book Offs a qualified replacement from the Union Hall.
  - c) Other Absences in accordance with the provisions of Article 20.
  - The minimum manning provisions shall be maintained at two (2) per shift at all times. An Employee absent in excess of two (2) consecutive days shall be replaced in accordance with (b) or (c) above.
- (Original signed May 20, 1997 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS

LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND

WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

## SUBJECT: HYDRAULICS AND ELECTRICAL MOTORS

This letter confirms the understanding and agreement between the Parties, as discussed during negotiations, with regard to work to be performed by Westshore Maintenance personnel related to Hydraulic Cylinders and Electrical Motors.

In conjunction with the above the Company has agreed to undertake specific modifications to existing facilities and purchase required equipment to facilitate the performance of functions referred to in this letter.

The specific functions the Parties agreed would be performed by Westshore Maintenance personnel are as follows:

#### 1. Hydraulic Cylinders

All cylinders to a maximum of 12" bore by 110" stroke will be repaired on site by Westshore maintenance personnel, except for the following:

| Mobile Equipment       | Equipment Where an exchange program exists with a vendor.   |  |
|------------------------|---|--|
|                        | All critical lift cylinders.  |  |
| Luff Cylinders         | The following cylinders will be dismantled by Westshore maintenance personnel, repaired and rebuilt by an outside shop and on return installed intact by Westshore maintenance personnel.     |  |
|                        | -Stacker Reclaimer 41   |  |
|                        | -Stacker Reclaimer 42   |  |
|                        | Note: The parties have agreed that at such time as the employees become proficient in performing the above functions, the Luff Cylinders will be repaired and rebuilt on site.                |  |
|                        | Dismantling, repair and rebuild of Luff Cylinders for Stacker Reclaimer 43 shall be performed by an outside shop and on return will be reinstalled intact by Westshore maintenance personnel. |  |
| Machining and Chroming | g All cylinders requiring machining and/or chroming shall have such work done by outside resources.   |  |

### 2. Electrical Motors

Removal, repair and reinstallation of Electrical Motors as follows:

| Junction Boxes  | rotations up to 250 Hp Motors (except                  |
|-----------------|--|
|                 | hazardous location motors)                             |
| Bearing Changes | up to 250 Hp Motors (except hazardous location motors) |
|                 | economics may dictate if repaired or replaced          |

It is acknowledged that Explosion Proof Motors are "Hazardous Locations Motors" and that GE is the only shop in the Lower Mainland that can recertify a motor for hazardous locations. Any motor of this type that has had any repair work or modifications to it must be recertified prior to being put back into "Hazardous Location" service. The work can be done at other shops, but must be recertified at GE.

#### 3. Exclusions

It is understood and agreed that Hydraulic Cylinders and Electrical Motors exceeding the above conditions or parameters will be removed from existing equipment by Westshore Maintenance personnel, sent off site intact for necessary repair or rebuild, and on return will be reinstalled intact by Westshore Maintenance personnel.

This Letter of Agreement will take effect at such time as the modified facilities and equipment are in place. The Company agrees that the required modifications to existing facilities and purchase of required equipment shall be completed on or before July I, 1998.

- (Original signed February 4th, 1998 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

February 4, 1998

# LETTER OF AGREEMENT between WESTSHORE TERMINALS LTD. and INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

## SUBJECT: 15.04 TEMPORARY EMPLOYEE SENIORITY

This Letter of Agreement is to clarify application of the provisions of subsection 15.04 (a) (ii) of the Collective Agreement with respect to entitlement to a credited year of Westshore service for the year 1997 only.

On application of Article 15, subsection 15.04 (a) (ii) the following shall apply:

- a) If a Temporary Employee has a break of twenty one (21) days or more, but works a minimum of twelve hundred (1,200) hours, then such Temporary Employee shall be given a credited year of Westshore service for the year 1997.
- b) If the Temporary Employee worked a minimum of six hundred (600) hours during 1997, from the last break in service of twenty-one (21) days or more, then such Temporary Employee shall be given a credited year of Westshore service for the year 1997.
- c) If the Temporary Employee, during the period October I to December 31, 1997, did not have a break in service of twenty- one (21) day or more, and worked a minimum of three hundred (300) hours during this period, then such Employee shall be given a credited year of Westshore service for the year 1997.
- (Original signed February 4th, 1998 by Alan Kazuta and Brian Ringrose)
- (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
- (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### SUBJECT: MDA/MDB SCHEDULES

This Letter of Agreement confirms the agreement between the parties with respect to changes to the specific terms and conditions that apply to the MDA/MDB schedules:

- 1. Article 24 of the Collective Agreement does not apply to these schedules.
- 2. Subject to the limitations specified in this Letter of Agreement, the Company will determine when absent employees have to be replaced.
- 3. Replacements must be provided for vacation absences.
- 4. In the case of absences of less than two (2) weeks, the Company can decide to replace or not, on a day-to-day basis.
- 5. In the case of absences of two (2) weeks or more, the Company must decide at the outset to either replace or not to replace for the entire period of the absence.
- 6. When the Company makes the decision to replace an absent employee, the replacement will be in accordance with existing rules.
- 7. In all cases where the absence is known at the outset to be for four (4) or more days duration, the replacements will be provided from the Union hall.
- 8. The Company cannot use extra maintenance employees when any maintenance crew is not fully manned. When replacements must be provided to satisfy this requirement, the replacements will be handled in accordance with the rules in this Letter of Agreement.
  - (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

**Rick Hurtubise** 

Rick Hurtubise President, Local 502

#### SUBJECT: M5 SCHEDULE LANGUAGE IN THE COLLECTIVE AGREEMENT

This Letter of Agreement confirms the agreement between the parties with respect to the collective agreement language as it relates to the subject of M5 schedules:

- During the term of the 2011 2016 Collective Agreement, all 2000-2003 Collective Agreement language that relates to the terms and conditions for establishing, changing and regulating M5 schedules is suspended until 11:59 p.m., January 31, 2016. During the period of the suspension the Company will not assign any additional employees pursuant to Schedule III attached to the collective agreement.
- 2. The suspended M5 schedule language has been temporarily deleted from the printed version of the 2003-2007 collective agreement for the limited purpose of giving effect to the suspension.
- 3. Notwithstanding the suspension of the M5 schedule language, the three (3) positions currently assigned to an M5 schedule will be maintained. The terms and conditions that apply to those currently assigned positions will be continued unchanged.
- 4. At 11:59 p.m., January 31, 2007 the suspension of the M5 language will end. Thereafter, all M5 language contained in the 2000-2003 collective agreement will be deemed to be reinserted into the 2003-2007 collective agreement and that language will then be in full force and effect as it was prior to the suspension, unless and until it is terminated, changed or modified pursuant to Article 26.01.
- 5. Notwithstanding any provisions of the Collective Agreement, the parties agree that when the Union is unable to supply manpower from the Union Hall at straight time, and the position is filled in accordance with the manning rules at overtime rates, the overtime may not be banked.
  - (Revised and signed May 20, 2004 by Alan Kazuta and Kent Birmingham)
  - (Revised and signed July 18, 2007 by David Honsberger and Kent Birmingham)
  - (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
  - (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### SUBJECT: NO ONE AVAILABLE (NOA) AND BANKING

Notwithstanding any other provision in the Collective Agreement, if there is no one available from the Union Hall to fill a dispatch, and the vacancy is filled by an employee on overtime, then that overtime may not be banked, but must be paid out.

- (Signed on July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

#### **SUBJECT: SEVERANCE**

This document confirms that employees who have left the employ of the Company voluntarily and are subsequently rehired will be entitled to severance for all their years of service at Westshore subject to meeting the criteria outlined in 19.02 of the Collective Agreement and subject to the exception immediately below.

Employees who are terminated for just cause or resign as part of an arrangement between the parties as a result of a workplace infraction, will only have their Westshore service from their most recent date of hire considered for the calculation of service.

Example One: An employee with 15 years of WTLP service resigns but maintains 1200 hours/year at WTLP for five years and then is rehired as a regular and works for 15 years and then retires will be eligible for 35 years severance.

Example Two: An employee with 10 years of WTLP service resigns and does not maintain 1200 hours/year of WTLP service, but earns 15 years of waterfront service, is rehired as a regular at WTLP and works for five years and then retires is eligible for 25 years severance.

Example Three: An employee with 15 years of WTLP service resigns and commutes his pension, is rehired as a regular and works for a further 20 years will be eligible for 35 years severance.

- (Signed on July 18, 2007 by David Honsberger and Kent Birmingham)
- (Reprinted and signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### SUBJECT: AUTOMATION AGREEMENT

- 1. Reduction of one employee per crew through attrition, total of five positions eliminated.
- 2. Minimum manning reduced to seven per crew in operations.
- 3. Control room when three or fewer stacker/reclaimers operating remotely, one operator with relief.
- 4. Four stacker/reclaimers in auto mode, two operators who will provide relief for each other.
- 5. Prior to engaging a stacker that has not been in production; an employee will confirm the machine status and report any issues. In addition, and on a daily basis, while in production mode, an employee will ascend the machine to the operator's cab to assess the status of the machine. This will be done by the relief operator or the second control room operator, or if the relief operator is otherwise assigned, another available employee.
- 6. One Control Room operator per shift to be posted, with the second position per shift becoming part of the job pick. Relief is in job pick. Minimum of 5 employees on each crew to be trained in control room. Vacancies in the Control Room will be filled as per the current application of R-2 90. Corresponding crew includes the crew the vacancy is currently on. In the case of any other vacancy that is required to be replaced, the replacement will come from a qualified employee on the crew, and the subsequent replacement, if required, will be made as per Appendix B of the Collective Agreement.
- 7. The declaration of intended operating mode of stackers will be made at the job pick. Should the status change during the shift, affected employees may be assigned to other duties.
- 8. Status quo on utilization of extras.
- 9. No reduction through attrition prior to July 1, 2011, for the five positions as outlined above, and no further attrition will occur as a result of additional technological change during the term of the Collective Agreement that becomes effective on February 1st, 2011.
- 10. At least one week of specific training in control room required.
- 11. Employees in the CR, including those assigned to relief for the CR, may be assigned other tasks as per the normal practice in the operations department. The Control Room position is part of the operations department and as it is a continuous operation, a hot change for the position is required.

#### R1 3-11

- 12. This agreement is without prejudice to any future automation except as stated herein.
- 13. In order to eliminate work place shortages, a Dedicated Spare Board List is to be created for WTL. The number of employees on the board to be determined, but half of the employees on the initial posting, must be on the list for one year, and the others for six months, after which all postings are for one year. The parties to review and agree on the numbers of employees on the list and in each category. The Union will establish rules best designed to eliminate NOA's, and will meet monthly with representatives of the Company to review the effectiveness of the Dedicated Spare Board and associated rules. The parties, through mutual agreement, may make changes to the numbers of employees on the list, and/or to the service pay thresholds, should they agree that such changes will be of benefit in addressing the NOA issue.
- 14. Employees on the Westshore Dedicated Spare Board List with 600 hours worked in any calendar year will accrue 3% of pay for hours worked up to 800 hours, after which they will accrue 5% for hours worked. This stipend will be deposited to an individual locked in RRSP with Sunlife, or other similar fund manager. Each employee will have an account for years in which there fewer than 1200 hours worked (Account B), and an account for years in which there were 1200 or more hours worked (Account A). The deposit will be made in January following the year in which an employee reached 600 hours worked, and for every year the employee achieves that milestone. The money will be invested in an age based fund, and all fees associated with the fund will be paid by the account holders. Upon becoming a regular employee, should an employee qualify for a pensionable year of service under the Westshore plan and qualify for an accrual of service pay for the same year, the employee may opt for either the service pay or the pension on an all or none basis for all years in which he worked 1200 hours or more. This decision must be made within 90 days of achieving regular status at Westshore. Should he accept the service pay, he forgoes the Westshore pension plan for those years. Should he select the Westshore pension, the balance of his locked in RRSP (Account A only) will be transferred to the pension as payment or partial payment to the plan for the required pension contribution. Service pay for years in which there were 600 hours worked, but less than 1200 hours will belong to the employee (Account B), and will not affect the calculation of his Westshore pension should he achieve regular status. Effective upon the ratification date of the renewal of the agreement that expired January 31, 2016, new regular employees will no longer have the option to have their previous service recognized under the previous defined benefit pension plan, but will be eligible to participate fully in the defined contribution plan, and as such their RRSP balance as set out herein will be transferred to and locked in under the defined contribution plan.
- 15. Control room to be cleaned by the inside clean up personnel.
- 16. As a gesture of good faith, the letter of advice dated September 29th, 2010 from Mr. Denis Horgan, addressed to Mr. Chad O'Neil, will be withdrawn. However, the Company reserves the right to take appropriate measures should further manpower shortages occur.
- 17. The Letter of Understanding R-2 95 will form part of the Collective Agreement that commences after the expiry of the Collective Agreement with the commencement date of February 1, 2011.
- (Original signed by David Honsberger and Chad O'Neill June 21, 2012)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

#### SUBJECT: AUTOMATION PROTOCOL

In reference to Point 6 of the automation agreement signed on August 25, 2011, the following will apply.

There will not be a requirement to post any vacancy for which there was not 21 days notice given ,(R2-90 1(a & b)), until there are a minimum of five employees on each crew trained in the operation of the Control Room. Once this minimum is achieved, R2-90 will be implemented in its entirety as per its current application.

Once all the postings for regular operators in the Control Room have been filled, these employees, and/or their replacements, will be assigned to the control room for a 90 day period. During this period, these employees may only be reassigned to deal with emergency pollution control issues. At the conclusion of the 90 day period, the parties will evaluate the operation of the stackers in automation mode, and may extend the 90 day period.

This letter will expire upon the achievement of having five employees trained in the Control room on each crew, and the expiration of the 90 day period and any extensions thereto.

- (Original signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

#### SUBJECT: CONTROL ROOM OPERATOR PROTOCOL

Prior to the start of a shift, should the Control Room be deemed not required for the entire shift, the Control Room Operator will be part of the job pick. (The Control Room Operator will retain a selection in the job pick, which will only be exercised if the Control Room is not being operated for the entire shift).

Should the Control Room be operated for a portion of a shift, and not required for a portion of a shift, the Operator may be assigned to perform inside or outside cleanup, shop clean–up or pollution control duties when not required in the Control Room.

- (Original signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### SUBJECT: SUBROGATION CLAIMS

The parties to this letter agree that the Company has the right to recover salary continuance benefits it pays employees when such employees are absent from work as a result of a motor vehicle accident.

The employee in receipt of such benefits from the Company will co-operate with the Company in the Company's filing of a subrogation claim with a third party, and will endorse an agreed upon subrogation agreement. The Company will cover any and all costs associated with its filing of a subrogation claim. Should an employee be in receipt of such benefits from the third party and from the Company, the employee must advise the Company immediately of such circumstances and repay the Company the monies.

The Company agrees that it will only make claim of the amount it has paid the employee in salary continuance, and specifically will not claim monies received by the employee for pain or suffering, general damages, loss of premium pay or overtime opportunities for the period the employee was in receipt of salary continuance.

The employee may also be required to co-operate with our LTD carrier for any subrogation claim it may have.

The foregoing is without prejudice to either party's position on the matter of subrogation.

- (Original signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

## SUBROGATION AGREEMENT

I, \_\_\_\_\_, agree and acknowledge as follows:

- Westshore Terminals Limited Partnership ("WTLP") has agreed to pay me pursuant to the salary continuance program (the "Policy") wages in the amount of \$------ representing my wages for the period of time \_\_\_\_\_\_, to \_\_\_\_\_ (the "Benefits"), during which I am absent from work as a result of my motor vehicle accident injuries of \_\_\_\_\_\_\_, (the "Accident"), and as a result is fully subrogated to any claim I may have arising out of the Accident but only to the extent of all salary continuance benefits paid to me;
- 2. WTLP is entitled to recover from me the Benefits from any compensation I may in the future receive for wage loss as a result of a claim or cause of action against any person, organization or legal entity (the "Third Party") arising out of the Accident;
- 3. WTLP may, at its sole option and at its cost, commence an action in my name against any Third Party arising out of the Accident for the salary continuance paid to me (a "Subrogated Claim").
- <u>4.</u> If WTLP commences a Subrogated Claim, I will fully cooperate with WTLP, and without limitation will provide full details of my injury including names of any possible Third Party and possible witnesses, and provide copies of all documents in my possession, power or control which WTLP may request and which are relevant to the Action, the Beneficiary's Claim or WTLP's consideration of the issues which arise in the Action.
- 5. If I commence an action or make claim against any Third Party arising out of the Accident, I will advance a claim for the amount of the salary continuance paid to me by WTLP, and will repay WTLP the amount of any monies recovered from the Third Party attributable to wage loss up to the amount of the salary continuance paid to me. The amount to be repaid to WTLP will be reduced by the legal fees, disbursements, and PST/GST/HST attributed to the WTLP share based on the same ratio as the WTLP's interest in the amount recovered to the total recovery.
- <u>6.</u> I agree to make such repayment on the condition that WTLP will issue a letter confirming the amount repaid by me to WTLP for tax purposes.
- <u>7.</u> The existence of an action commenced by or on behalf of an employee does not preclude WTLP from joining the employee's action or commencing an action on its own behalf respecting the salary continuance paid.
- 8. Where WTLP or the employee intends to commence or join such an action, they shall advise the other in writing of that intention.

Date:

Signature:

Witness:

### SUBJECT: APPRENTICES HOURS

The Company will review the records of all active apprentices who have successfully completed their apprenticeship as at the date of ratification of this agreement, at Westshore and who have become regular members of the work force to ensure that they have been credited with a Westshore pensionable year of service for each of the prescribed years of the apprenticeship program they served at Westshore.

Employees who did not receive a credited year of service in any of the years they served as an apprentice, will receive such credit within 60 days of the ratification of this agreement.

Notwithstanding the foregoing, apprentices indentured as of October 28th, 2011, and all future apprentices, will have to work a total of 1200 hours in a calendar year at Westshore in order to receive a year of Westshore credited service.

- (Original signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

It is understood that the composition of the Electrical Department will be evolving over the term of the Collective Agreement.

The current vision of the department includes the creation of positions within the electrical group that would be trained to the level of a technician. The first two employees receiving such training will be selected as set out below, with employees with adequate training ultimately being supplied by the hall.

The parties will meet and review candidates for this training and mutually agree on the successful candidates. Employees selected for such training must meet all eligibility requirements for the training that will likely occur at BCIT, or a similar institution, and successfully complete all components of the training, including all examinations. The employees must sign an agreement with the Company that will allow the company access to all scholastic records including attendance. A failure to maintain a passing grade, or acceptable attendance, will result in the removal of the employee from training. The costs of tuition and books will be paid by the Company. The employee will receive his regular monthly salary while in attendance at school. Replacements must be qualified as printmen, and the subsequent vacancy will be filled from the hall.

The parties will also review and agree upon the amendments to the roles of printmen and "technicians", and create a training curriculum for the roles in the electrical group.

- (Original signed June 21, 2012 by David Honsberger and Chad O'Neill)
- (Reprinted and signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### SUBJECT: DEDICATED SPARE LIST

Rules for List:

- 1. There will be a dedicated list to be reviewed and posted every 6 months.
- 2. The list will provide a number of employees that are suitable to both Parties.
- 3. The Parties agree to meet once a month to review the list.
- 4. If after the 6 month period an employee in the list has not maintained a minimum 250 hours, then they will not be eligible to be placed on the dedicated spare list for the next 6 months.

| Category          | Number | Minimum Hours |
|-------------------|--------|---------------|
| Operator/clean up | 80     | 250           |
| Electricians      | 40     | 250           |
| Mechanics         | 40     | 250           |
| Welders           | 20     | 250           |
| Cat               | 10     | Exempt        |
| Lubeman           | 10     | Exempt        |
| First Aid         | 10     | Exempt        |

- Black Book Letter 3-11 point 14 will be paid annually.
- Lubeman and Clean-up will be included in the operators category and will be exempt from the qualifying hours.
- The Parties recognize there may be an exemption from the qualifying hours due to the inconsistent work opportunities.
- If an employee is in the Cat, Lube or F/Aid category, and is included in any of the above said categories, then they will not be exempt.
- When on their scheduled day's off, employees on non time and travel postings from the dedicated spare list who have made themselves available to work in accordance with the Union's rules of dispatch and are called into work, will be paid at the overtime rate of pay. Should this be abused, the parties will meet to review the situation with the view of correcting the behavior.

- The Union agrees that there will be no book offs allowed for employees from the hall that are non Union members, except under extenuating circumstances.
- It is further agreed that if any employees who is a Union member on the dedicated list and on time and travel books off, they will be down for any job dispatch out of the hall for the remainder if the posting.
- (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21<sup>st</sup> day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

## INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

#### 2-16

### SUBJECT: DISPATCH UPGRADE

The parties agree to meet within 90 days of the ratification of the Collective Agreement, or at another such mutually agreed time, to review the current method of communications between the Union Hall and the Company as it relates to dispatch of employees. These discussions shall include a review of more efficient options and associated costs.

• (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

**Rick Hurtubise** 

Rick Hurtubise President, Local 502

### SUBJECT: REVENUE GENERATING PRODUCT

The Company recognizes the Union's concern with respect to revenue generating product arriving on site other than by rail and understands the Union's concern with respect to jurisdiction. All such occurrences will be reviewed with the Union prior to movement of product to site.

Therefore, effective upon ratification, and retroactive to any incident or occurrence after March 3, 2016, the following shall apply:

- For product that is spilled prior to delivery on site, and is subsequently brought to site by a mode of transportation other than by rail, a product movement surcharge will be paid for each such shift that product is moved onto site and placed in an area(s) designated by the Company to a maximum of 20 shifts per incident.
- If a car has been dumped on site or an attempt has been made to do so, and there remains coal in the car that is returned to site, no product surcharge is required for such secondary delivery.
- For product that is delivered to site originating at source by other than rail, the parties will meet to review the situation and to arrive at a mutually acceptable product movement surcharge.

A product movement surcharge will equate to eight hour's pay at the regular 502 hourly rate, and will be added to the plant Committee's pay claim tally.

If requested by either party, a meeting will be held within a reasonable time period for the purpose of reviewing, amending, or cancelling this letter.

• (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

**Rick Hurtubise** 

Rick Hurtubise President, Local 502

## SUBJECT: SCISSOR LIFT AND THIRD PARTY WORK

When work is being performed with the Scissor Lift, the Shift Operator will operate the Scissor Lift and work along with the employee assigned to that job. There will not be a requirement to assign any other employee to that task.

Further, it is understood that if a third party is performing contract work for the Company and using the Company's mobile equipment, there will be a shift operator assigned. If they are supplying their own equipment, there will not be a requirement to assign a shift operator to the work being performed.

Notwithstanding the foregoing, prior to a third party performing crane work on site, it will be reviewed with a representative(s) of the contracting out committee, and if the work is being performed separately and distinctly from our employees, then there will neither be an observer nor a shift operator attached to the work. If the third party supplies a crane and is working in conjunction with our employees, then the parties will review whether an observer will be paid or a shift operator assigned.

• (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

### SUBJECT: SAFETY

Where there is a conflict between Article 16.02 of the Collective Agreement and the Terms of Reference developed and signed by the JOSHC, the Terms of Reference shall govern. It is further agreed that the Terms of Reference must be reviewed and approved annually by the Plant Chair or his designate, and the Director of Human Resources or his designate prior to implementation. The parties further agree that if there is no agreement on the Terms of Reference, then the Collective Agreement shall govern.

• (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

This letter confirms that the parties have agreed to establish and implement the Westshore - I.L.U.W. Local 502 Apprenticeship Program (hereinafter called the "Program") in accordance with the provisions of Article 18 – Training of the Collective Agreement.

It is the intent of the Program to train individuals to become competent, reliable, safe and productive tradespersons for employment as temporary relief, daily dispatch or regular employees of Westshore Terminals Ltd.

The parties agree that the following terms and conditions will apply to the Program. Unless otherwise agreed to below the term and conditions of the existing Collective Agreement and the Black Book will apply.

#### **TERMS AND CONDITIONS**

- The Company and the Union will each appoint three (3) representatives to an Apprenticeship Council that is created by this Agreement. The President of Local 502, one electrician, and one mechanic will comprise the Union representation on the Council. The Manager – Maintenance Mechanical, the Manager – Maintenance Electrical, and a Maintenance Superintendent will comprise the representation of Management on the Council. The appointments will be for a four (4) year term.
- 2. The Council will be responsible to administer the terms and conditions of the Program, develop policies and procedures that will increase the effectiveness of the Program, select individuals for indenture, and monitor the progress of all apprentices.
- 3. The Company and the Union recognize that the Program must comply with the provisions of the Apprenticeship Act and any other legislation and/or statutory regulations governing apprentices.
- 4. The Company will, on instructions from the Apprenticeship Council, post for an apprentice in either the Millwright or Electrician trade following posting procedures that are established by the Council.
- 5. The Apprenticeship Council will accept applications for an apprenticeship from regular employees, temporary relief employees, daily dispatch employees, and/or other individuals it deems appropriate
- 6. The Council will select individuals for an apprenticeship based on the following criteria: education, related training/experience in the trade, mechanical or electrical aptitude and seniority.
- 7. Applicants for apprenticeships must have successfully completed the Entry Level Trades Training program in the trade for which he is making application or possess experience in a vocation that is deemed by the Apprenticeship Council as being equivalent to successful completion of the Entry Level Trades Training program.
- 8. If an individual is selected by the Apprenticeship Council to be indentured and if the individual has worked as an uncertified tradesperson for a period of time or if the individual has related experience to the trade in which he will be apprenticing, then the individual will be required to write a slotting examination that will be administered by the Job Training Division of the Ministry of Advanced Education and Job Training. The individual will be then assigned to the appropriate level in the apprenticeship program based on the test results and subject to the approvals of the Apprenticeship Council and the Job Training Division.
- 9. Any person who is selected for an apprenticeship must successfully complete a ninety (90) day probationary period before being indentured. It is understood that the person selected for the apprenticeship must pass the Company physical that is outlined in Article 13 of the Collective Agreement as part of the probationary period requirements.
- 10. Once an apprentice is indentured his performance will be monitored by the Company and periodic interview will be held with the apprentice to review his performance.

#### Apprenticeship Program (continued)

- 11. The Company will ensure that the apprentice will be given the necessary on-the-job practical training to allow the apprentice to become competent in the trade so far as the facilities and the scope of the business will permit. Practical training may include rotation through the crews (MA-ME, M1-M4) to facilitate job training on all equipment.
- 12. The apprentice will be required to attend vocational school for training as set out by the Job Training Division. While the apprentice attends vocational school his work schedule will be considered to be the school schedule for the weeks that he is required to attend vocational training.
- 13. Progression through the various levels in the apprenticeship program will be based on working the prescribed number of hours and successfully completing the examinations at the end of the period of required vocational training.
- 14. While an apprentice is attending vocational school for training under the Program, the company will maintain the apprentice's straight time wages less any supplemental allowance or subsidy that he receives for the vocation school period of training provided the apprentice attends the scheduled training classes outlined by the vocational school.
- 15. If an apprentice fails the required examination at the end of the period of technical school training then the apprentice will be required to withdraw from the Program.
- 16. When an apprentice graduates to certified tradesperson status he will be included on the Spareboard List of acceptable individuals referred to in Article 20, Section 20.03. of the Collective Agreement. Once an apprentice graduates from the Program he will be required to work, subject to his seniority in the Union, on temporary relief and/or daily dispatches as a tradesman at Westshore for a period of time (measured in calendar months) that is equal to the period of time of his apprenticeship.
- 17. The Company will not be required to replace the apprentice at any time. The apprentice (s) will not be considered part of a maintenance crew for the purpose of administering Article 9 Section E.2 paragraph 3, and/or Article 24 Section B, D, and E of the Collective Agreement.
- 18. The Council may make decisions concerning the requirement of an apprentice to work overtime or extensions for job training or work experience purposes. The apprentice will not be eligible to work overtime from the designated trades overtime board unless the Council so agrees.
- 19. The company will provide the apprentice with necessary tools commensurate with his progress in the Program. The tools, provided to the apprentice, remain in the property of the Company and are not to be removed from the Company property.
- 20. The rate of pay that an apprentice receives will be graduated on the basis of the apprentice's progress in the apprenticeship program and will be a percentage of the negotiated wage rate set out in Article 25 of the Collective Agreement. The rates are appended to this agreement as Appendix I and form part of this Agreement.
- 21. The apprentice shall, subject to the eligibility requirements, be entitled to the benefits outlined in Article 10 and 11 of the Collective Agreement except that life insurance, accidental death and dismemberment insurance, salary continuance, long term disability, and other salary related benefits will be based on the salary that the apprentice earns or earned at the time the apprentice is entitled to the benefit requested.
- 22. Any increase in the apprentice's rate of pay will occur once the Company has been notified in writing by the Job Training Division that the apprentice has successfully completed the exams at the end of the vocational period of training, and the apprentice has completed the number of hours required by the Apprenticeship Agreement to advance to the next level in the apprenticeship. In addition, the apprentice must have completed the previous period of vocational training, pass a skill testing job examination which is mutually agreed upon on or about the date the increment is due and be actively at work on the date the increment is due.
- 23. The apprentice will normally be assigned to one of the MA-ME or M1-M4 shift schedules unless otherwise agreed to by the Council for training/job experience purposes.

7-16

#### Apprenticeship Program (continued)

7-16

24. The apprentice shall, subject to the provisions outlined in Article 9 of the Collective Agreement, become eligible for entitlement to vacation during the period of his apprenticeship with Westshore Terminals Ltd. For the purposes of calculating entitlement to vacations the first shift worked as an apprentice shall be considered as the date of hire. Once an employee has completed his apprenticeship his entitlement to vacation will cease. If, at the time the apprenticeship is completed, the apprentice has not taken all of the vacation to which he is entitled then he shall be paid out his vacation based on the appropriate percentage of earnings year to the date the apprenticeship is complete less any vacation pay he had received at the time vacation was taken. The calculation of the payout will be made on the same basis as is the calculation referred to in Article 9, Section A.2 of the Collective Agreement.

The parties further agree that the above conditions do not set precedent for the granting of vacation to temporary workers nor do they prejudice the position of the Company and the Union in respect of the interpretation/application of the Collective Agreement or the Black Book.

- 25. The Apprentice Council, depending upon the performance of an apprentice and during the last year of the apprenticeship, may make application to the appropriate Job Training Division to grant the apprentice a credit of hours towards the completion of the Apprenticeship Program. If a credit is approved then the Apprenticeship Agreement will be amended accordingly.
- 26. The Union and the Company agree that this Agreement will terminate thirty (30) calendar days after either party has received written notice by double registered mail of the other party's decision to cancel the Agreement. If the Agreement is terminated then the terms and conditions will remain in full force and effect until such time as the apprentices complete their apprenticeship or withdraw from the program, whichever shall first occur. The parties further agree that neither party may terminate the Agreement for the purpose of negotiating changes to this Agreement.
- 27. The Union and the Company agree that this Agreement constitutes the entire and sole Agreements between the parties in respect of apprenticeships at Westshore Terminals Ltd. This Agreement takes precedence over any other undertakings or understandings between the Union and the Company. No other agreements are either expressed or implied.
- (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

# APPENDIX I APRENTICE SALARY SCHEDULE

| Apprenticeship Level | Percentage of Salary |
|----------------------|----------------------|
|                      |                      |
| 0 – 6 months         | 55%                  |
| 7 – 12 months        | 60%                  |
| 13 -18 months        | 65%                  |
| 19 – 24 months       | 70%                  |
| 25 – 30 months       | 75%                  |
| 31 – 36 months       | 80%                  |
| 37 – 42 months       | 85%                  |
| 43 – 48 months       | 90%                  |

### SUBJECT: ARTICLE 24.04 (A) INTERPRETATION GUIDE

- 1. Production employees is understood to mean the crew operators in the job pick dispatch who perform work on the shiploaders, stacker/reclaimers, dumpers or the relief for these positions.
- 2. A stoppage of production work includes a rerouting of product, a reduction to a single source loading, or the inability to dump, stockpile, reclaim or load a vessel, due to a plug up or an emergent coal spillage. All such scenarios that involve a production employee on days or afternoons being assigned to clean up will be reviewed quarterly. Should it be determined that there was an improper assignment, a payclaim(s) shall be paid within thirty (30) calendar days of the review.
- 3. When a stoppage of production work as defined by the foregoing requires cleanup, normally the cleanup operators will be dispatched to perform the required cleanup. When no cleanup operators are available, the Company may assign other production employees to perform this cleanup work. The production employees assigned to perform this cleanup will be given an hour break prior to performing this work and one half hour break prior to returning to production work.
- 4. If the company requires a production employee to perform general clean up during the shift, then it will be declared at the commencement of the job prior to the start of the shift. If cleanup is required at the start of the shift, then there will be no requirement for the one hour break prior to starting cleanup, but they will receive a one hour break before commencing production. Otherwise, they will receive a one hour break before commencing cleanup work and a one half hour break prior to returning to production work, excluding the lunch break.
- 5. If a stoppage of work occurs prior to the start of the shift and the Company requires this work to be done at the commencement of their shift, then there will be no requirement to provide such employees with an hour break before performing the clean up work. Such employees shall be provided a one half hour break before returning to production work, excluding the lunch break.

#### **Graveyard Shift Exceptions**

- 6. Because there is no permeant clean up operators on the graveyard shift, the Company may decide to utilize the SRO's as general cleanup operators as stated in point 4.
- 7. If the SRO is required to perform clean up after the job pick where a stoppage of work occurs, they shall receive one half hour off before doing the work and shall be provided a one half hour break prior to returning to production work (1 hour if moved and 2 or more times), excluding lunch break.
- 8. If the SRO's have been assigned to general clean up in the job pick and are required to reposition a stacker, they will perform the work and after will receive a one hour break prior to returning to their duties.

#### **General Interpretations**

9. Training:

If training is ongoing in the crew, the trainer and trainee will have first pick of the available jobs to ensure they have adequate opportunity to cover all training requirements.

10. Extensions:

If an employee is extending 4 hours into the next shift, these employees shall receive a break of up to 30 minutes at the beginning of the shift. This break will be provided within the first hour, but it is understood that the machines will not be idled due to a failure to provide this break.

11. Cats:

If the Company decides to add cat operators in the job pick on dayshift, they may assign them to other operator duties one time. It must be declared in the job pick at the commencement of the shift and assigned no later than the end of their lunch break.

If the Company decides to add cat operators in the job pick on afternoon or graveyard shift, it is understood that there will be no other duties attached to such employees.

• (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

**INTERNATIONAL LONGSHORE AND** WAREHOUSE UNION, LOCAL 502

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

## LETTER OF INTENT

June 21, 2012

Mr. Tim Farrell President I.L.W.U. – Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Tim,

In consideration of the presentation made at our most recent Plant Committee meeting and discussions held thereafter, the Company is willing to amend the Automation Agreement as set out below.

These amendments are made with the understanding that there will be no additional costs to the Company, and that the parties recognize that the position is not a preferred category as set out in the Collective Agreement. In addition, either part may serve 30 days notice of the cancellation of this letter and therefore revert to the initial Letter of Understanding.

The parties agree to amend the Automation Agreement as follows:

Point Six – Replacements - If there exists a vacancy in the Control Room, and the vacancy is one that will initially be sourced from overtime after transferring a qualified employee from the crew, then the vacancy in the Control room will be filled through offering overtime to a qualified control room operator through the normal overtime procedures. For clarity, the Company may use the minimum manning requirements and choose to transfer an employee to the Control Room and not fill the vacancy. Vacancies for which a replacement is initially sourced from the hall will continue to be filled as per point six in the initial automation agreement Letter of Understanding.

Point Six – Job Pick – The Control Room Operator will only be an active participant in the job pick if the Control Room is not going to be manned for the shift and this intention is made known prior to the job pick.

Should you agree with the terms as set out, please endorse where indicated below.

Yours truly,

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger

Director, Human Resources

Rick Hurtubise President – I.L.W.U. Local 502

Revised and signed this 21st day of May 2021.

## LETTER OF INTENT

Mr. Tim Farrell President I.L.W.U. – Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Tim,

As a follow up to our meeting of July 12, 2012, the Company puts forth the following position with respect to the addition of one shift Operator to each crew.

- 1. A total of five Shift Operator positions will be posted effective August 1, 2012 on each crew.
- 2. One each of these five positions will be assigned to A-E crew, and will ultimately work a rotating schedule consisting of Days, Afternoons and Nights.
- 3. Until the Maintenance night shift commences, the schedule will rotate to include days and afternoons only. The Night shift rotation will be re-scheduled to Day shift on an interim basis.
- 4. For clarification purposes only, the current Shift Operators will continue on a day and afternoon rotating schedule.
- 5. As per current practice, Shift Operators may be assigned duties in either department at any time during their shift.
- 6. There will be no requirement to replace the first two Shift Operator vacancies on Day shift or the first Shift Operator vacancy on Afternoon shift. When required, replacements will be sourced in the normal fashion.

It is understood that should the Night shift in Maintenance be terminated, or the tonnage reduced significantly, then some or all of these new positions may be reduced through attrition. Should this occur, the parties will meet to review the impact this will have on the operation, including scheduling and replacement issues.

Please contact me should clarification be required.

Yours truly,

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

## Letter of Intent

March 2, 2015

Mr. Tim Farrell President - I.L.W.U. - Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Tim,

The following outlines our position with respect to the addition of two troubleshooters per crew in Operations.

- The positions will be posted as regular work force positions and will be for one Electrician and one Mechanic / Welder on each of A, B, C, D, and E crews.
- These are hot change positions and will be dispatched by Operations.
- The positions will be on a trial period of one year, and should the Company determine it does not wish to maintain these positions after the one year period, the employees will be assigned to work in the Maintenance Department and reduced by attrition. The foregoing is subject to a general layoff occurring under Article 15. Should the positions be maintained, then the terms of this document and the requirement to replace absences other than those caused by vacation or in excess of two weeks will be reviewed by the parties after the one year period. Amendments will be by mutual agreement.
- Should the vacancy be as a result of a NOA, another qualified troubleshooter, and failing that a tradesman who is a regular employee, from the same category will be assigned to work as a troubleshooter. This tradesman may be replaced at the Company's option. Further, there will not be a requirement to pull an employee from another crew or extend shifts to replace one of these troubleshooters.
- When more than one set of troubleshooters is working on any given shift, their coffee and meal times will be staggered.
- Operations will declare at the commencement of dispatch if the second set of troubleshooters is not required for the shift due to the amount of equipment in operation. If they are not required, at the discretion of the Company, they will be assigned to perform operations equipment checks, or to perform site lighting checks. The parties agree these employees may be reassigned during the shift to troubleshooter duties, and if so, they must remain as troubleshooters for the duration of that shift.
- While these are hot change positions the company may not require either set of troubleshooters to await the arrival of their relief if the other set of troubleshooters is on duty.

Yours truly,

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger

Director, Human Resources

Mr. Tim Farrell President I.L.W.U. - Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Tim,

Further to our discussions of May 16, 2013, the following outlines our position with respect to the new pollution control system. Normally the operation of the rain bird system will be initiated from the automation control room by the Automation Operator for each shift, in conjunction with a Foreman and one of the five newly created positions entitled Site Environmental Serviceman (SES). These positions are being created as part of our renewed commitment to environmental stewardship, and their main purpose is to ensure that our pollution control systems are working to their maximum potential to eliminate the risk of air or water born contaminants leaving our site.

While we are confident these positions will prove to be worthwhile, as discussed on numerous occasions, each employee is responsible to ensure that he can work safely and comfortably while in the yard by either locking out or deselecting particular rain birds through a request to either a Foremen or to the Automation Operator.

#### Replacements (SES)

The SES positions do not preclude the inclusion of a pollution position within the job pick, or the use of other employees on pollution duties, whether or not the SES is at work, or is absent and is not being replaced. Normal replacement rules for non-preferred jobs apply to this position, except for vacation which shall be replaced following the provisions of R2-90.

#### Position (SES)

The SES will be a distinct hot change position and is not part of the job pick. They will be provided with a suitable light vehicle and a radio. The positions may be terminated at the discretion of the Company. Should this occur, the parties will meet to review the terms of the termination of the position. The employees would then be reduced by attrition.

There will be a minimum of two additional employees on each crew trained for this position, and all regular work force crew operators will be offered training first, then temporary employees may be trained for this position as required. This position will be included in Step One of Appendix C of the Collective Agreement.

This position is subject to the regular protocol as it relates to re-assignment.

This position will perform their regular duties on all Statutory Holidays if the site is operating on any of these days. When the site is operating on standby, the SES will be added to the standby pollution crew and be called in as part of the crew. He may be required to access the Automation Control Room to operate the new pollution system. The standby crew will be maintained as per RI-85, and if additional pollution control help is required, they will be called in, again as per RI-85.

#### 12-16

Letter of Intent - Pollution Control System (continued)

#### Duties (SES)

The following is not intended to be a comprehensive list of duties, and is subject to addition or deletion following discussion with the Union. Note the SES employee must be trained as an Operator.

- Perform normal pollution control duties, including the operation of water trucks to wet piles and/or roads.
- Inspect and maintain valve, pump and reservoir houses, and the "foam" plant upon receiving the appropriate training from maintenance.
- Complete check lists as required (e.g. daily, weekly, and/or monthly).
- Perform minor maintenance on the new pollution control system, such as replacing valve handles and lubrication of rain birds. When assistance is required in the performance of these duties, it may come from either Operations or Maintenance.
- Maintain rain bird colour scheme and numbers and the housekeep around the rain bird blocks.
- Monitor, maintain and operate all flume systems, including Berth Two Dock and Surge Bin area, Berth One Dock-face and Surge Bin Area, and the flume at the head-end of Six B conveyor.
- Operate sieve bins and ocean discharge.
- Perform winterization duties related to pollution control systems. Dependent upon the duties, he may be working with other employees, and specifically with Maintenance employees when working with compressors or welding machines etc.
- Ensure hoses are available and stored properly at transfer stations and other designated areas.
- Operate sanding truck and assist in snow removal.
- Train other employees on the operation of the new pollution system.
- Assist in troubleshooting any and all areas of the pollution control system, including assisting in clearing plugged lines, and operating valves to isolate specific lines.
- Assist other employees and third party personnel in identifying proper lockout locations in the various valve houses.
- Operate the new pollution control system, in either manual or automatic mode from the valve houses when required.
- Assist in tie up or let go duties when there are shortages.

It is understood that while these duties are primarily those of the SES, other employees may assist in the performance of them. Check lists will be developed for the position with input from the employees holding the position, the SES trainer, and Company and Union representatives. These lists will reflect the duties and are subject to change. The lists and changes are subject to mutual agreement.

12-16

Letter of Intent - Pollution Control System (continued)

12-16

#### Automation Control Room

Training in the operation of the rain bird system in automatic mode will be conducted initially by the proper designated trainer.

Please contact me if clarification is required. Otherwise please endorse where indicated below and return the endorsed document to me and post the positions as soon as possible.

Yours truly,

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger

Director, Human Resources

our

Tom Doran President – I.L.W.U. Local 502

• (Original signed July 31, 2017 by David Honsberger and Tom Doran)

Revised and signed this 21<sup>st</sup> day of May 2021 on behalf of:

## WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise President, Local 502

## Letter of Intent

Mr. Rick Hurtubise President - I.L.W.U. – Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Rick,

### **RE: Leasing Mobile Equipment**

The Company will advise the Union prior to leasing any additional pieces of mobile equipment that is to be used on site and is related to the normal day to day operation of the terminal.

Further, if the Company has determined that it will be a long-term lease, they will explore with the vendor having Westshore employees perform some or all the maintenance on the leased unit.

Yours truly, WESTSHORE TERMINAL LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

## Letter of Intent

Mr. Rick Hurtubise President - I.L.W.U. – Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Rick,

## **RE: Dry Facilities**

Further to the concerns raised during negotiations with respect to the dry facilities, the Company commits to continuing discussions with the Union and will review modifications that can be made to the dry to address such concerns. This agreement does not constitute a guarantee of renovations or modifications to the facility.

Yours truly, WESTSHORE TERMINAL LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

### Subject: Safe Work Procedure

The parties agree to develop a Safe Work Procedure through the HPP process for operating the grader during non-daylight hours. This procedure will be developed prior to operating the grader during such times.

Printed and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise

President, Local 502

3.21

## Subject: Gearman

Within the first year following ratification of the Collective Agreement, the position of a "Gearman" will be created. The duties of this position will include: the inspection and maintenance of all rigging and fall arrest harnesses; inspections of fire extinguishers and eye-wash stations throughout the site; maintaining the tool crib; clean up and housekeeping in mobile, weld, electrical and rebuild shop areas, including areas around shops and equipment; sorting, labelling, maintaining etc. of tools; and, miscellaneous maintenance work in tool crib. The parties will meet to review theses duties with a view to ensuring this position is fully utilized.

This will be an operations position but will be supervised by a maintenance foreman, and will be posted and awarded to the most senior applicant with the necessary aptitude to fill the role. The parties will review the selection process and the employee must successfully complete any and all required training.

The first day of any absence from work will be at the Company's discretion as to whether it will be filled, unless the absence is as a result of an employee' birthday holiday as per Article 8.01 (b).

Replacements will be from the operations department, with the subsequent replacement coming from either overtime or the hall, subject to the normal source of replacement, (including the procedures as outlined in Letter of Understanding R2-90), for the nature of the absence. There will be no requirement to double or extend (before or after shift).

Employees interested in being eligible to fill vacancies must indicate so by signing a posting for orientation. <u>Two employees will be trained on of the Operations A-E crews</u>. The orientation for this position will be a minimum of one shift, and it is understood that replacements will not generally be given training in all aspects of the position.

The schedule will be a steady day schedule following E crew.

• (Original signed July 4, 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

Rick Hurtubise

President, Local 502

R1-5.21

## Letter of Intent

Mr. Rick Hurtubise President - I.L.W.U. – Local 502 11828 Tannery Road Surrey, B.C. V3V 3W7

Dear Rick,

### **RE: Building Cleaning Duties Division**

The following will confirm our discussions with respect to the division of work between Local 517 janitors and Local 502 inside and outside clean-up as it relates to the Westshore offices and shops building, <u>and among the two ILWU locals and outside contractors in the employee and contractor parking lots</u>.

Local 502 employees will be responsible to clean:

- All corridors on the first floor, except for the main entrance foyer.
- The 502 men's dry and 502 lunch room.
- The auto room, warehouse and tool crib area.
- The First Aid room and garage.
- The men's and women's washrooms in the shop area and the disabled washroom on the first floor.
- All areas south of the swing doors except for the <u>following areas on the mezzanine above the cat</u> <u>shop: the training room, hallway, storage rooms and office.</u>
- The building perimeter, including parking areas and lower windows, within the fences.
- <u>Maintenance of lighting</u>, the vehicle wash-station, <u>and pump stations</u> in the <u>parking lots</u> outside the fences.
- The cleaning of the training rooms on the first floor when they are being used as additional lunchrooms <u>for 502 employees</u>.

Local 517 employees will be responsible to clean:

- All training rooms and offices on the first floor, including the fit-test room.
- The 514/Management dry.
- The 502 female dry.
- The main entrance foyer and elevator.
- The "blues" storage rooms.
- All staircases north of the swing doors.
- The front of the building outside the fences.
- The office, training room, hallway and proposed washroom(s) on the mechanical shop mezzanine.

- All areas on the second floor including windows.

In addition, Local 517 employees will clean the new guard shack. They will not be responsible for cleaning elsewhere on site.

A contractor may be used to treat the parking lots outside the fences as required, and Local 502 employees may also be required to treat the parking area. <u>One pay claim will be conceded for each employee for each eight-hour period, or portion thereof, a contractor performs work</u>. Local 502 employees will perform snow removal in the parking lot if required.

The Company will continue to advise the union of the gardening needs required outside the fences, and concede pay claims equal to the time spent by a contractor performing the gardening. <u>This method of calculating pay claims is restricted to this circumstance only unless the parties agree otherwise.</u>

Please review the division of duties contained herein and endorse where indicated below if you accept them as laid out.

Yours truly, WESTSHORE TERMINAL LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

#### Subject: Maintenance Orientation

The Company will, in consultation with a designated Union Representative from each of the mechanical and electrical areas create a departmental orientation template for new employees.

The template will include an introduction to the respective department's systems, processes, equipment, relevant procedures and policies, duties and other aspects of the department that may be identified during the template development process. The template will include an orientation checklist and sign off that will be completed identifying that the new employee has had exposure to all identified aspects of the department. This template may be amended as required in consultation with a designated Union Representative.

The duration of the department orientation will be determined following the template development and may be amended as required. The department orientation will commence following a one shift site orientation that will be conducted by a member of 502 during their regularly scheduled shift and the six day evaluation period as contained in R 1-89 2(d), or the commencement of the probationary period for new regular employees.

There will be positions posted in the respective departments for an employee and a backup to perform the orientation which will be awarded to the senior qualified applicants. Qualifications will include length of continuous service at the terminal, with a minimum of five years preferred, the ability to communicate effectively both orally and in writing, and an acceptable standard of job performance. Neither employee involved in the orientation will be permitted time off during this time frame. The employee providing the department orientation will do so on his regular shift and will not be replaced and will receive a premium of two hours pay at straight time for such shifts.

The creation of this template will begin within three (3) months of the ratification of this agreement. One year following implementation of the orientation process, the parties will meet to review its effectiveness and discuss changes either may recommend.

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502

Page | 87

### **Subject: Electrical Mentors**

The company will hire a regular work force electrician per each Apprentice to ensure the appropriate level of training for the apprentice is achieved. Each electrician so assigned, commits to mentoring/training apprentices and is to provide feedback with respect to the performance and competency of the apprentice. As this position is created to ensure the proper mentoring and success of the apprentice, the electrician's time away from work should be scheduled to minimize their absences when the apprentice is at work.

These positions will be assigned A to E days/afternoon shift schedules as set out in the Collective Agreement. With appropriate notice, and following discussions with the Union, these positions may be assigned to other schedules as per the Collective Agreement, based on the requirements of the Company.

Replacements will be as per the Collective Agreement. Generally, replacements for this position will be filled (when possible), from the crew.

When the apprentice is absent, the mentor may be assigned shop work or be partnered with an electrician sourced from the hall. There will be no requirement to replace the position when the Apprentice is absent.

If there is a reduction in the number of electrical apprentices indentured or an elimination of electrical apprentices, these electricians will be absorbed into the department and assigned a crew, with any reduction of employees as a result of the loss of an apprentice(s) being by attrition. If the reduction of apprentices is part of an overall reduction of forces, these employees will be subject to layoff as per the provisions of the Collective Agreement.

This agreement may be cancelled by either party upon the submission of 60 days written notice. The parties will meet to discuss the associated reduction in forces required as a result of the cancellation of this letter prior to the implementation of a layoff.

• (Original signed this 28<sup>th</sup> day of April 2017 by David Honsberger and Tom Doran)

Reprinted and signed this 21st day of May 2021 on behalf of:

WESTSHORE TERMINALS LIMITED PARTNERSHIP

David Honsberger Director, Human Resources

INTERNATIONAL LONGSHORE AND WAREHOUSE UNION, LOCAL 502

Rick Hurtubise President, Local 502